ANALYSIS OF LONGITUDINAL CATEGORICAL AND COUNT DATA SUBJECT TO MEASUREMENT ERROR







Analysis of Longitudinal Categorical and Count Data Subject to Measurement Error

© Yunqi Ji

A thesis submitted to the School of Graduate Studies in partial fulfilment of the requirements for the degree of Dector of Philosophy in Statistics

Department of Mathematics & Statistics Memorial University of Newfoundland

January 2011

St. John's

Newfoundland

Abstract

In biaseling, weid, behaviora, and environment areas, the data set for output enlishest from encyces, registration approximation, finisiar trials, and stellar obervariant or experimental matter, which are often contaminated with measurement ences. This may be due to the imperfect interments and procedores, limited or proteine and knowledge or mainters and examines. Registin measurement encyc in responses results in blands estimates of model parameters. Explicit module are required to describe the michaedinations on categorical mappings and court terms and any encycles. To abata now related inference, one mode to take the measurement encode lateria in the developing statistical methods to analyze misemented data.

In this thesis, we define a generalized thinning operation, based on which we propose a transition model for categorical longitudinal data. This new transition model can flexibly accommodate a variety of linear and nonlinear transition models: We also discuss a thinning-operation-based transition model and an ordinary linear transition model for dynamic count data.

More importantly, we present near severe measurement error models for entryperiod data and conta data, which first for two responses with the observed, could be measured responses by exploit expressions. A meaningful applications of the exploit minimization model is to donerher the unbiasted minimizations in tempertical data, which produces an admensive severy priority model the data and definition both minimizations and neares singly subus due to "hauring" markers. Mercore, the course are models with assessments the both to "marker" anteress. Mercore, the course are models with assessments the both the reservents and on discovered data can be used to describe some interesting count data of direase cases with different situations of the dynamic population sizes of an area. We apply these explicit measurement error models and transition models to analyze the longitudinal discrete data subject to measurement errors.

Methods have on the gravallest estimating equations (CRS), prevalual queues, detailed (CRQ), the sense of set CG (CRQ), so on reactions in Helderold (DR), set developed to shoke its charge exists estimates of the subcover parameters in large/statical models for categories of our encodes. The expectite manutement of the subcover of the subcover of the subcover parameters models and us models an indexist prediction of these sequences. These methods but su models an indexists gravational of subcover models and us models an indexist synthesis of model parameters, estimated mathematical encodes and understative statistical of model parameters, estimated models around factor development of the the future topological parameters for the future prediction of the sequences could also, so encode each of the half-half-ood equivalence but helders of spranch. The future future is the half-half-ood equivalence is the Half-inoperation of the future prediction of the structure is used for expensive cound data, even the CQL approach provide shares as good estimates as the Ma sigparads. The future future is the site is business to the Ma significant data, which complicated dependence structure is the history to bargetophila data which complicated dependence structure is the history is bounded in the Ma significant data. The complicated dependence structure is the history is bounded and data. The proposed methods are discussed by an example of database matham data from Harvel SC Clares Subs.

Acknowledgements

I would like to take this opportunity to express my heartiest appreciation to my advisor, Dr. Zhaoshi Fan, for his insightful guidance, consistent support, and impressive encouragement throughout the four years of my PhD program. I would also like to thank all my PhD advisor committee members, Dr. Hong Wang, Dr. J. Concorciolic Lacebook for their support during my research.

I sincerely thank Dr. Beajendra Sutradhar, Dr. Alwell Oyet, Dr. Yingwei Peng for their academic support during my program. I also want to acknowledge Dr. Cary Saeddon, for his kind support and help during the Graduate Program in Teaching.

I would like to address my sincere acknowledgements to the examination committee members, Dr. Yingwei Peng, Dr. Alwell Oyet, and Dr. Yanqing Yi. They gave me a lot of useful comments on my thesis.

I am grateful to the School of Graduate Studies and the Department of Mathematics and Statistics for financial support during my PhD program. Also I want to thank all faculties and staff in the department for their kindness and help.

Special thanks go to my family, especially my wife. Their endless love and support is always essential to my success.

Finally, it is my great pleasure to thank my friends and fellow student who encouraged and helped me during my Ph.D. program.

Contents

| Abstract | 15 |
|---|-----|
| Acknowledgements | iv |
| List of Tables | ix |
| List of Figures | xii |
| 1 Introduction | 1 |
| 1.1 Longitudinal Studies | 1 |
| 1.1.1 Overview | 1 |
| 1.1.2 Transition models for dynamic categorical data | 4 |
| 1.1.3 Transition models for dynamic count data | 7 |
| 1.1.4 Generalized estimating equations and generalized quasi-likelihood | |
| approaches . | 9 |
| 1.2 Measurement Errors | 12 |
| 1.3 Objective of This Thesis | 17 |
| 2 Classification Error and Count Error Models | 20 |

| | 2.1 | Overv | iew | 20 |
|---|-----|--------|--|----|
| | 2.2 | Gener | alized Thinning Operation | 23 |
| | 2.3 | Classi | fication Error Models | 31 |
| | 2.4 | Count | Error Models | 36 |
| | | 2.4.1 | Multinomial count error model | 37 |
| | | 2.4.2 | Corrected additive count error models | 42 |
| 3 | Lon | gitudi | nal Transition Models for Categorical Data and Count | t. |
| | Dat | a | | 45 |
| | 3.1 | Trans | ition Models for Categorical Data | 45 |
| | | 3.1.1 | A transition model for dynamic categorical data | 45 |
| | | 3.1.2 | The transition model for dynamic binary data | 54 |
| | 3.2 | Longi | tudinal Models for Count Data | 58 |
| | | 3.2.1 | Non-stationary AR(1) model | 59 |
| | | 3.2.2 | Linear transition model | 62 |
| | | 3.2.3 | Moments of the NS-AR(1) and LT models | 64 |
| | | 3.2.4 | Estimation of the model parameters | 67 |
| | | | 3.2.4.1 Generalized quasi-likelihood method | 68 |
| | | | 3.2.4.2 GQL2 approach | 68 |
| | | | 3.2.4.3 Maximized likelihood method | 70 |
| | | 3.2.5 | Simulation studies | 73 |
| | | | 3.2.5.1 Designs | 74 |
| | | | 3.2.5.2 Estimation of model parameters | 75 |
| | | | 3.2.5.3 Misspecified baseline observations . | 78 |
| | | | | |

| | | | 3.2.5.4 | Misspecification of models | 82 |
|---|-----|--------|------------|---|-----|
| 4 | Mo | deling | Misclase | ified Longitudinal Categorical Data | 87 |
| | 4.1 | Overv | iew | | 87 |
| | 4.2 | Miscla | ssified Lo | ngitudinal Binary Data | 92 |
| | | 4.2.1 | Model d | escription | 93 |
| | | 4.2.2 | Estimat | ion of the model effects | 98 |
| | | | 4.2.2.1 | GQL method | 98 |
| | | | 4222 | Maximum likelihood method | 99 |
| | | | 4.2.2.3 | GQL2 (OGQL) method | 105 |
| | | 4.2.3 | Simulati | on Studies | 107 |
| | | | 4.2.3.1 | Covariate designs | 108 |
| | | | 4.2.3.2 | Estimation of the model parameters | 109 |
| | | | 4.2.3.3 | Insight to robustness: a continued simulation study . | 121 |
| | 4.3 | Applie | sation to | Children Asthma Data | 126 |
| | 4.4 | Joint | Modeling | the Misclassified Data with Missing Information Due | |
| | | to "Ur | sure" Re | sponses | 141 |
| | | 4.4.1 | Model d | escription | 141 |
| | | 4.4.2 | Estimati | ion of model effects | 145 |
| | | | 4.4.2.1 | Ignoring the "unsure" responses | 148 |
| | | | 4.4.2.2 | Taking missing values into account | 154 |
| | | 4.4.3 | Simulati | | 156 |
| | | | 4.4.3.1 | Design | 156 |
| | | | 4.4.3.2 | Simulation results | 158 |

| 5 | Mo | leling Mis-measured Longitudinal Count Data | 163 |
|---|-----|--|-----|
| | 5.1 | Overview | 163 |
| | 5.2 | Miscounted Binomial Count Data with Dynamic Population | 164 |
| | | 5.2.1 Models | 164 |
| | | 5.2.2 Estimation of the model parameters | 167 |
| | | 5.2.3 Simulation studies | 170 |
| | | 5.2.3.1 Covariate design | 170 |
| | | 5.2.3.2 Data generation | 171 |
| | | 5.2.3.3 Simulation results | 172 |
| | 5.3 | Miscounted Longitudinal Data with Little Information about Popula- | |
| | | tion Size | 177 |
| | | 5.3.1 The model | 177 |
| | | 5.3.2 Estimation of the model parameters | 180 |
| | | 5.3.3 Numerical study | 182 |
| | | 5.3.3.1 Covariate design | 182 |
| | | 5.3.3.2 Data generation | 184 |
| | | 5.3.3.3 Simulation results | 185 |
| 6 | Dis | ussion and Future Studies | 191 |
| | 6.1 | Some Remarks | 191 |
| | 6.2 | Future studies | 195 |
| | | | |

List of Tables

| 1.1 | Misclassification from true category T into observed category Y with | |
|-----|--|---------|
| | equal numbers of categories. | 14 |
| 2.1 | Misclassification from true category T into observed category Y with | |
| | unequal numbers of categories. | 21 |
| 2.2 | Balanced misclassification of air quality in an environmental study. | 34 |
| 2.3 | Unbalanced misclassification with $r < s$ in an example of asthma study. | 35 |
| 2.4 | Blood types and their genotypes, an example of unbalanced misclassi- | |
| | fication with $s < r$ | 35 |
| 2.5 | Example of misclassified disease cases | 41 |
| 3.1 | Transition probabilities from $T_{i,j-1}$ to T_{ij} | 46 |
| 3.2 | Simulation results for the NS-AR(1) model with the true values of | |
| | parameters $\beta = (1, -1, 1)$ | 77 |
| 3.3 | Simulation results for the LT model with the true values of parameters | |
| | $\beta = (1, -1, 1)$ | 79 |
| 3.4 | Mis-specifying the baseline observation $t_{d0}=50$ when $t_{d0}\sim Pois(50)$ | |
| | with $\beta = (1, -1, 1)$ and $\gamma = 0.65$. | 81 |

| 3.5 | Misspecified LT model under true NS-AR(1) model, where $\beta = (1,-1,1)$ | . 85 |
|------|---|------|
| 3.6 | Misspecified NS-AR(1) model under true LT model, where . $\hfill \ldots$. | 86 |
| 4.1 | Misclassified Asthma Status | 93 |
| 4.2 | Simulation results under Design 1 with $(\pi^+,\pi^-)=(0.95,0.90)$ and the | |
| | true values of parameters $\beta = (1,1)$ | 115 |
| 4.3 | Simulation results under Design 2 with $(\pi^+,\pi^-)=(0.95,0.90)$ and the | |
| | true values of parameters $\beta = (1,1)$ | 116 |
| 4.4 | Simulation results under Design 3 with $(\pi^+,\pi^-)=(0.95,0.90)$ and the | |
| | true values of parameters $\beta = (1,1)$ | 117 |
| 4.5 | Simulation results under Design 1 with $(\pi^+,\pi^-)=(0.75,0.80)$ and the | |
| | true values of parameters $\beta = (1,1)$ | 118 |
| 4.6 | Simulation results under Design 2 with $(\pi^+,\pi^-)=(0.75,0.80)$ and the | |
| | true values of parameters $\beta = (1,1)$ | 119 |
| 4.7 | Simulation results under Design 3 with $(\pi^+,\pi^-)=(0.75,0.80)$ and the | |
| | true values of parameters $\beta = (1,1)$ | 120 |
| 4.8 | Robustness about estimated (π^+,π^-) based on 500 simulations under | |
| | Design 3 with true values $(\pi^+,\pi^-)=(0.95,0.90),\beta=(-1,1),\gamma=1,0$ | 124 |
| 4.9 | Robustness about estimated (π^+,π^-) based on 500 simulations under | |
| | Design 3 with true values $(\pi^+,\pi^-)=(0.75,0.80),\beta=(-1,1),\gamma=1,0$ | 125 |
| 4.10 | Exploratory Analysis of Asthma Data of 537 Children from Steubenville, | |
| | Ohio in H6CS | 128 |
| 4.11 | Analysis of Asthma Data of 537 Children from Steubenville, Ohio in | |
| | H6CS Taking Misdiagnosis into Consideration | 135 |

| 4.12 | Unbalanced misclassification of children asthma | 142 |
|------|--|-----|
| 4.13 | Simulation results under GQL approach for imperfect data due to miss- | |
| | ine values and misclassification with the true value: $\theta = (1, 1, 1.5)$, . | 161 |

xź

List of Figures

| 2.1 | The True and Reported Disease Cases in An Area | 41 |
|-----|--|-----|
| 4.1 | Estimates of the Intercept β_1 in Model (4.34) for Asthma Data of 537 | |
| | Children from Steubenville, Ohio in H6CS Taking Misdiagnosis into | |
| | Consideration. | 136 |
| 4.2 | Estimates of the Effect of Mother's Smoking Status β_2 in Model (4.34) | |
| | for Asthma Data of 537 Children from Steubenville, Ohio in H0CS | |
| | Taking Misdiagnosis into Consideration. | 137 |
| 4.3 | Estimates of the Dynamic Dependence Parameter γ in Model (4.34) | |
| | for Asthma Data of 537 Children from Steubenville, Ohio in H6CS | |
| | Taking Misdiagnosis into Consideration. | 138 |
| 4.4 | Estimates of the Odds Ratio about Mother's Smoking Status for Asthma | |
| | Data of 537 Children from Steubenville, Ohio in H6CS | 139 |
| 4.5 | Estimates of the Odds Ratio about Prior Asthma Status for Asthma | |
| | Data of 537 Children from Steubenville, Ohio in H6CS | 140 |

Chapter 1

Introduction

1.1 Longitudinal Studies

1.1.1 Overview

Longitudin districts data such a comparid and sourd and way and way to who range of arms good heading, abelian, communis, acatelog, and such as the communis, acatelog and activation and are colled particles. In Addition, there are other names for different types of data, for example, separated many and the mestics. Baseling, the baseptication data are a simple of analysis, such of the abelgets are repeatedly measurements, characteristic and the other strength separated measurements, strength, the Heard's 2016 of Pablic Hadio control in Lipse particular data and the strength separated measurements of strength, the Heard's 2016 of Pablic Hadio control is large balantian of a ways. The particular data and strength separate strength and the superstay hadin harmony addition that the effective of al (1904). Wore, et al. (1904). A part of the strate₂, as analysis of data ways and the particular to be superstay hadin harmony hading and the particular data the the effective of the trade term annial strength trades are are particular to the trade that the superstay hadin harmony hading and the particular data the trade the trade that the trade term and the superstay and the trade term and the trade term of the trade term of the superstay hadin harmony hading and the theter stress are trade to the trade term and superstay trade the trade term of the was determined based on the information provided by their parents through some standard questionnaires and the results of pulmonary function test by means of a portable survey upinometry.

In larginguised actuales, basiles the observations of internets reproperty of the second seco

The defining characteristic of longitudinal data is that the multiple descertaion within subjects are not independent of each other. Therefore, in data analysis, one should take into account the correlation beare modervation. These these mes object. To combine different correlation activations into the analysis, opecial statistical methods are recepted. This helps to draw more reliable statistical inference, especially for diverse data.

There are three models which are extensions of generalized linear models (GLMs) for longitudinal data: marginal models, random effects models and transition models. In summinal models, when the correlation structure of the responses is not of direct intrest to the researchers, one mainly focuses on the effects of contristes. Hence the marginal expectation, $\mu_{ij} = E(Y_{ij})$, is modeled as a function of some explanatory variables. Therefore, the regression of the responses on covariates can be modeled separately from the correlation within subjects. For example, a logistic marginal model for longitudinal binary data can be given by

$$logit(\mu_{ij}) = z'_{ij}\beta,$$

 $Var(Y_{ij}) = \mu_{ij}(1 - \mu_{ij}),$
 $Corr(Y_{ij}, Y_m) = \rho$

where Y_{ij} is the response of subject *i* at the *j*th time point, and x_{ij} is the corresponding covariates, for i = 1, 2, ..., I, and j = 1, 2, ..., J.

When we are interested in making statistical inference about the individuals but not population average, a random effects model will be helpful. Random effects models along for the natural heterogeneity cross-subjects by assuming that coefficients of some covariants follow a probability distribution. For example, a random effect model in the GLM knowedve can be given by

(1)
$$\mu_{ij}^c = g(x_{ij}^c \beta + z_{ij}^c U_i)$$
, where $\mu_{ij}^c = E(Y_{ij} = 1|U_i)$,

- (2) U_i, i = 1, 2, . . . , I, are mutually independent with a common multivariate distribution F,
- (3) g(·) is the inverse of a specific link function in GLM. For binary data, it may be the logit or probit function, that is, logit(x) = log(^x_x) or probit(x) = Φ⁻¹(x), and Y₀(U₁ → Φ(1,µ²₀). Whereas, for count data, g(x) = log(x), and Y₀(U₁ → Paisson(y¹₀). Gives 1, Y₀(U₂ → 1, 2, ..., A see independent of each other.

If one is interested in both the effects of covariates and the dynamic departitions many discretization within induces, as transition model serves on a good advantative. Under a business model, the personst response is explicitly influenced by the historici of sub-residues prior to time j., which is demodel by $\mathbb{N}_0 = \{ u_{ijk} = -1, 2, ..., -j = 1 \}$. Therefore, bulk the explanations variables and the part contenues are transfer as transfer deter variables. Let $\mu_{ijk}^{(j)} = E(Y_{ijk}) \mathbb{M}_2^{(j)}$ and $\Psi_{ij} = \mathrm{Ver}(Y_{ijk}) \mathbb{N}_2^{(j)}$ be the conditional expectation and variance of Y_{ijk} graps part extrames and the covariants. Thus, a granued transition model on the graps by

$$g_{ij}^{c} = g(x_{ij}^{i}\beta, f(H_{ij}^{i}; \gamma))$$
 (1.1)

$$t_{ij}^{e} = v(\mu_{ij}^{e}\phi),$$
 (1.2)

where $g(\cdot)$ is the inverse of a specific link function, and the transition from the previous states is represented by a series of known functions $f = (f_1, ..., f_N^*)$, to the current response. Due to the intuitive dynamics among outcomes within subjects, in this thus, we will focus on developing statistical analysis of transition models for the correlated activencial and count data.

1.1.2 Transition models for dynamic categorical data

There are different transition models proposed for dynamic categorical data.

 Tong (1990, p. 113) discussed a linear transition model for dynamic binary data. This model is given by

$$y_{ii} = b_{ii}y_{i,i-1} + (1 - b_{ii})e_{ii},$$
 (1.3)

where $b_{ij} \sim b(1, \gamma_{ij})$ is the random dynamic dependence variable, $\epsilon_{ij} \sim b(1, \xi_{ij})$,

and h_{ij} is independent of ϵ_{ij} . This can be easily generalized to accommodute the dynamic categorical data of dimension r by assuming that h_{ij} still follows $(1, \epsilon_{ij})$, hot ϵ_{ij} follows a r-dimensional multinomial distribution, that is multinomial, $(1, \epsilon_{ij})$. Similarly h_{ij} is assumed to be independent of ϵ_{ij} . It then follows that $1_{ij}^{ij} \sim (4L, \mu_{ij})$, where

$$\mu_{ij} = \gamma \mu_{i,j-1} + (1 - \gamma)\xi_{ij}$$

= $\gamma^{j-1}\mu_{i1} + (1 - \gamma)\sum_{n=2}^{j} \gamma \xi_{in}.$ (1.4)

As discussed by Satradhar and Farrell (2007), the mean μ_{ij} in equation (1.4) is a function of not only the current covariants x_{ij} but also all historical covariate $\{x_{in}, u < j\}$. Actually, this model can further be generalized to the kth order transition model for categorical datas as follows:

$$y_{ij} = \sum_{u=1}^{k} b_{ij(u)} y_{i,j-u} + (1 - \sum_{u=1}^{k} b_{ij(u)}) \epsilon_{ij},$$
 (1.5)

where $h_{ij} \sim \text{multisonial}(1, \gamma)$, and γ is a vector of probability. Sutradhar and Farrell (2007) pointed out that the correlation coefficients between binary observations under model (1.3) do not cover the full ranges from -1 to 1, which limits the use of this linear model in practice.

2. A similar linear transition model was proposed by Qapish (2003). They used a family of multivariate binary distributions through a linear dynamic conditional expectation to construct the linear dynamic model. This conditional linear family of order k can be given by

$$P(Y_{ij} = 1|\mathcal{H}_{ij}^{k}) = E(Y_{ij}|\mathcal{H}_{ij}^{k}) = \mu_{ij} + \sum_{u=1}^{k} b_{ij(u)}(y_{i,j-u} - \mu_{i,j-u}),$$
 (1.6)

where $\mathcal{H}_{ij}^{ij} = (y_{i,j-1}, y_{i,j-2}, \dots, y_{i,j-k})$ denotes the history at the previous k time points. The vector $h_{ij} = (h_{ij(1)}, h_{ij(2)}, \dots, h_{ij(k)})'$ can be computed based on the specified correlation structure using

$$h_{ij} = [Cov(\mathcal{H}_{ij}^k)]^{-1}Cov(\mathcal{H}_{ij}^k, Y_{ij}).$$

As maximally by Mällick (2003), this model can use different working correlation structures, such as Gauss type AR(1), MA(1) or exchangeable correlation, by specifying in GC, which is free Gorce(Cd) = $V_{i}^{(1)}C_{i}V_{i}^{(1)}$, where $V_{i} = diag(\sigma_{0}, ..., \sigma_{0})$ with $\sigma_{m} = Vire(T_{m})$. However, it was noticed by Mallick (2003) and Eventi and Startaflact (2006), the magnet for the correlations in G_{i} are bound to be screetined as disc. $D(D_{i}) = V_{i} = V_{i} = V_{i}$.

 A nonlinear transition model was discussed by some authors [Korn and Whittemore (1979); Zeger, Liang and Self (1985)]. They comprise a first-order Markov chain which is given by

$$logit(\rho_{ij}^{*}) = logit(P(Y_{ij} = 1|H_{ij})) = x_{ij}^{i}\beta + \gamma y_{i,j-1}$$
 (1.7)

Some econometricians [Anemiya (1985); Maniki (1987)] called it a non-linear binary dynamic model. It has been shown by Tarrell and Sotradiar (2006) that this model produces a reasonable correlation at nucture that allows the ranges of the correlations to be from -1 to 1. Diggle et al. (2002, p. 191) estended model (1.7) to order 4, and has in.

$$logit(\mu_{ij}^c) = logit(P(Y_{ij} = 1|H_{ij})) = x'_{ij}\beta_k + \sum_{u=1}^k \gamma_u y_{i,j-u},$$
 (1.8)

where β_k is the regression coefficient with the Markov chain of order k.

1.1.3 Transition models for dynamic count data

For dynamic count data, there are also some transition models proposed.

1. Wong (1986) discussed a transition model

$$\mu_{ij}^{\epsilon} = exp(x_{ij}^{\prime}\beta)\{1 + exp(-\gamma_0 - \gamma_1 y_{i,j-1})\}, \text{ where } \gamma_0, \gamma_1 > 0.$$
 (1.9)

As a consequence of the constraints on γ_0 and γ_1 , this model only allows for a negative correlation between the prior and current responses. In addition, the conditional expectation μ_0^c must vary within a limited range from $exp(x_0^i,\beta)$ to twice this value under the assumption about γ_0 and γ_1 , which makes this model imparctical in more cases.

 Beaug (1974) and Diggle, et al. (2002 p204) discussed a nonlinear dynamic model for longitudinal count data which is given by

$$\mu_{ij}^{*} = exp(x_{ij}^{\prime}\beta + \gamma y_{i,j-1}).$$
 (1.10)

This model arous to be a makey with the logistic transition model (17). Here we it is has limited spin-fields in its practice brane the conditional expectation p_{ij}^{i} increases as an exponential function of the previous observations p_{ij} , when $\gamma > 0$. In the mass that the conditional expectation is independent on courting, the manufactory $p_{ij}^{ij} = \gamma_i$ both to a solutionary present only when $\gamma < 0$. Remay the model can only characterise together association without responsible memory unity may time.

3. Zever and Oanish (1988) introduced another log-linear transition model

$$\rho_{ij}^{c} = exp\{x_{ij}^{c}\beta + \gamma\{y_{i,j-1}^{*} - x_{i,j-1}^{c}\beta\}\}.$$
 (1.11)

where $g_{k_{d-1}}^{i} = \max_{j \in \{k_{d-1}, d\}}$ and 0 < d < 1. When $\gamma = 0$, is reduces to an ordinary log-linear model. When $\gamma < 0$, there is a negative correlation between g_{ij} and $g_{ij,-1}$. When $\gamma > 0$ there is a positive correlation. This model describes a multiplicative pattern among the y_{ij} 's.

4. Blundell, Griffith and Windmeijer (2002) proposed a linear feedback model (LFM) to analyze the relationship between RL62 and patents for a panel of US firms. Let $\xi_{ij} = exp(x_{ij}^2)$, and $E(Y_{ij})\eta_i = exp(x_{ij}^2) + \eta_i) = \xi_{ij}$, where η_i is the subject-periodic markon effect, and $\eta_i = exp(\eta_i)$, the the LFM is given by

$$E(Y_{ij}|y_{i,j-1}, v_i) = \gamma y_{i,j-1} + \xi_{ij}v_i$$
 (1.12)

Since the $\xi_{ij}v_i$ is non-negative, the conditional mean of y_{ij} is bounded by $\gamma y_{i,j-1}$ from below

 McKenzie (1988) discussed a stationary AR(1) model for count time series, and Sutradhae (2003) used it to model longitudinal count data. The model in longitudinal context is given by

$$y_{ij} = \gamma * y_{i,j-1} + \epsilon_{ij} \qquad (1.13)$$

where is the binamid thiming operation. Under this model, $y_{0,1} \sim Poisson(y_{0,2} - e_{1})$, $y_{0,1}(x_{0,1})$, $y_{0,1}($

and $\epsilon_{ij} \sim Poisson(\mu_{ij} - \gamma \mu_{i,j-1})$. Similarly, μ_{ij-1} is assumed to be independent of ϵ_{ij} . Under the non-stationary model, one may then show that $E(Y_{ij}) = Var(Y_{ij}) = \mu_{ij} = exp(x'_{ij}\beta)$ for j = 1, 2, ..., J. This non-stationary model can be used for the dynamic count data with time-varying covariates.

1.1.4 Generalized estimating equations and generalized quasilikelihood approaches

To estimate the unknown parameters in longitudinal models, different approaches are proposed. Among these approaches, the maximum likelihood (ML) method is considered to be the most efficient estimation procedure. Suppose that longitudinal data y_{ij} , for i = 1, 2, ..., I and $j = 1, 2, ..., J_i$ follows a first-order transition model, the likelihood function can be written as

$$L(\theta|y) = \prod_{i=1}^{J} f(y_{i1}) \prod_{j=2}^{J} f(y_{ij}|y_{i,j-1})$$
 (1.14)

The estimates can be obtained by maximizing the likelihood functions of the loglikelood functions ($0^{-1}_{\rm est} = 0^{-1}_{\rm est} (d_{\rm est}))^{-1}_{\rm est}$ results ($0^{-1}_{\rm est} = 0^{-1}_{\rm est} (d_{\rm est})^{-1}_{\rm est}$ results), developed the ML estimates of the parameters in the logistic transition model (1.7) for dynamic blauey data. However, the ML approach investing dynamics where the simulation that the logistic distribution of a js insures and energy. In the logistic transition of a js investing end energy of the end parameters of the simulation of the simulation of the simulation of the distribution of a js invest on end energy description of the ML approach the simulation of the (1.10). McKenice (1009) presented the complexities the simulation parameters in the logistic transition of the simulation the distribution may be violated in parameters are not never showever dependent values are simulated without in parameters of the new results in the theorem parameters are been apprecised values are simulated without in parameters of the new results with the theorem parameters and the simulation of the sinterval of the simulatin sinterval or the simul As an alternative of the ML approach, the generalized estimating equations insthod is proposed by Liang and Zeger (1986) for continuous or discrete longitudinal data. Their method is based on the estimating equations

$$\sum_{i=1}^{l} \frac{\partial \mu'_i}{\partial \theta} W_i^{-1}(y_i - \mu_i) = 0, \quad (1.15)$$

where $\mu_i = E(Y_i)$, and W_i is the working covariance matrix which can be decomposed into $V_{i}^{1/2}C_{i}V_{i}^{1/2}$. In the decomposition of W_{i} , $V = \text{diag}(\sigma_{i1}, \dots, \sigma_{in})$, and $C = C(\theta, \alpha)$. is the working correlation matrix which may depend on the parameters θ in the mean structure μ_{ii} and a correlation parameter α . In practice, an estimate of α can also be obtained, either based on a moment estimator or a second set of estimating equations. There are several popular correlation structures in practice. For example, the independence structure in which observations are assumed to be independent, the exchangeable structure is which the correlation between observations within subjects time between observations, and the unstructured correlation in which there is not an assumed nattern of correlations. It can be seen that the consistency of the GEE estimates only depends on the mean structure a... Therefore, regardless of the choice of the working covariance attracture, one can always obtain a consistent estimate of θ as long as the mean structure is correctly specified. The less dependence on the model assumptions makes the GEE approach one of the most popular methods in dealing with correlated data. More discussions can be found in Zeger and Lianz (1986) [also ree Zener and Oacish (1988)- Hardin and Hilbs (2003)- Directo et al. (2002)]

In the GEE approach, one takes into account the correlation by choosing a working covariance structure W_i in the estimating equations. However, if the chosen W_i is far from the true covariance matrix Σ_{c} , it will result in loss of efficiency. To improve the efficiency of estimation, a quasi-blackhood-based method was proposed by Welderburn (1974) by using the true covariance matrix. Satzahlar (2001) and Satzahlar and Farrell (2007) further discussed this generalised quasi-blackhood (GQL) methods, of which the estimating equations are given by

$$\sum_{i=1}^{l} \frac{\partial p_{i}^{i}}{\partial \theta} \Sigma_{i}^{-1}(y_{i} - \mu_{i}) = 0, \quad (1.16)$$

where Σ_i is the true covariance matrix of Y_i . The use of Σ_i leads to higher efficiency of the GQL satimate than the GEE estimates which are derived by using a working covariance structure. The GQL approach only depends on the first and second order moments of response Y_i which are smallable for many prediction models.

To further improve the efficiency of estimation of model parameters, Startaburs and Farerfi (2007) introduced a second order GQL approach by employing the first and the second order expression in the stimulating proceedings. We first the the GQL approach in this dustion. Let $F_i = (V_i^*, S_i^*)$, where $Y_i = (Y_{11}, ..., Y_{ij})$, and $S_i = (Y_{11}, ..., Y_{ij})$, $A_{ij} = (X_{ij}, X_{ij})$, where $Y_i = (Y_{11}, ..., Y_{ij})$, and $S_i = (Y_{11}, ..., Y_{ij})$, $A_{ij} = (X_{ij}, Y_{ij})$, where $Y_i = (Y_{11}, ..., Y_{ij})$, and $S_i = (Y_{11}, ..., Y_{ij})$, $A_{ij} = (X_{ij}, Y_$

$$\Omega_i = \begin{pmatrix} Cor(Y_i) & Cor(Y_i, S_i) \\ Cor(S_i, Y_i) & Cov(S_i) \end{pmatrix}$$

be the $m(m + 3)/2 \times m(m + 3)/2$ covariance matrix of F_c . The GQL2 estimating equations are given by

$$\sum_{i=1}^{l} \frac{\partial \delta_{i}^{i}}{\partial \delta} \Omega_{i}^{-1} (f_{i} - \delta_{i}) = 0, \quad (1.17)$$

where f_i is the observation of F_i . It can be seen that the GQL2 approach utilizes the moments up to order 4. However, the consistency of the estimates of model parameters depends on the correctly specified first and second order moments. Due to the use of more information from the data, the GQL2 is demonstrated to gain higher efficiency than both GQL and GEE approaches. In some cases, the GQL2 approach performs almost as well as the ML approach [Sotradhar and Farrell (2007)]. Therefore, in this structure, the GQL2 can be the optimal GQL (OGQL) approach.

1.2 Measurement Errors

Abbing most rather are well neighed to obtain accurate information, more summarizen traves in data all lever due to many shrown and unknown future three most imported in transmission and psecolaters, hantled knowledge and experiments of examines a canasimose, and on a Maximument errors may recursite in continuous data (e_{\pm} , the weights of etablisms), contegorized adus (e_{\pm} , indextion status), and count duiks (e_{\pm} , the weights of etablisms). Contegorized adus (e_{\pm} , indextion status), and count duiks (e_{\pm} , the weights of etablisms), contegorized adus (e_{\pm} , indextion status), and the star of an also-addisculture (content) during the during star graving numericand (columbition errors) which is do not to indextion starting or accurate starts in the start form of minoreal (count error) which is in the into indexting starting or accurate starts).

As an ensure, in a large population-based study to examine the effect of passive based on the probability of the ensure of the ensure probability constrained because they are relatively single and economical to conduct when compared to the chinal examination of each child. However, it is impossible to trading a perfectly enable ensurement of the ensurement on the engoded stream (see the enderation ensurement of the end-the end-the engoded ensure (see the end-the end-the end-the later end-the end-the end-the end-the end-the ensurement, the accuracy of a diagonal hand on the pathic [Jaharkov et al. (1992), there examines the end-the end-t very poor because of the great energie of ansaurometrie between backlip dathers and these with previous whereing. Therefore, the two backh states of a ddd is of directly downchik. Instead, where we can obtain its the diagonic tatata band on sum inspective information from the questionnaires. The data may therefore interministical by discussion (SEEE) are pourse to ensure its truth hands the matural state of the state of the state of the state state of the protonges of the SEEE states (SEEE) are pourse to ensure the two fittings and incompared the SEEE states (SEEE) are pourse to ensure the two fittings (SAB) et al. (SAB) (

There many literatures about measurement error models for different types of manuscenses errors in emissions data, for example, the channel resources of manuscenses errors in emission of the first (1997), Carell et al. (2008). Banay, Takono, and Schnakist (2003), quarkin error model (2004), and the at the main-final discriptional data are concerned, the chanic way to describe the minimalization from the true (binn, harrow) requests T in the description of all (1998) errors are presented as the second second second second second energy of the second second second second second second second second energy of the second second second second second second second energy of the second second second second second second second energy of the second seco

| | true category (Y) | | | |
|----------------------|-------------------|------------|--|---------------|
| observed category(T) | 1 | 2 | | r+1 |
| 1 | π_{11} | π_{12} | | $\pi_{1,r+1}$ |
| 2 | π_{21} | π_{22} | | $\pi_{2,r+1}$ |
| | | | | |
| r + 1 | #r+1.1 | Tr+1.2 | | Rr+1,r+1 |

Table 1.1: Misclassification from true category T into observed category Y with equal

numbers of caterories.

our knowledge, there is not an explicit misclassification model proposed for categorical data which clearly describe the relationship between the true remonse T and the observed response Y that is similar to the continuous data case. In this thesis, we propose such explicit misclassification models for mis-measured categorical data.

More interestingly, this model can accommodate unbalanced misclassification which can be used to deal with a special type of missing values. Aside from the case of misclassified categorical data, there are two explicit models

for mis-measured count data. The first model is the additive measurement error for Poisson count data proposed by Cameron and Trivedi (1998). It is given by

$$Y = T + e_s$$
 (1.18)

In this model, both the true count T and the addition error e are assumed to be representing random variables for example T ~ Poisson(n) and e ~ Poisson(f) Sec the nonmerative measurement error leads to a larger mean and variance relative to T. Therefore, it is useful only for describing count inflation.

Whiteoses and Gau(1999) possible there coult error model for the mixchandled result based on an example about metridly rate of enricolations. Let n_1 also donate, mogeneticity, the excert and incorect disoned columitations, and they are assumed to be independent. Finance matching, fastice $n_1 \rightarrow (n_1 + n_2)$ where n^+ is the minimizing. Suppose that T and Y are the two count and the response count of disoner cours, and L is the population site with the manual to be known. Then course curve models for Y index in the static distribution of the finance of the finance court n_1 and $N = N_1 + n_2 + n_3$.

$$Y \sim P(\mu^{c}),$$

 $\mu^{c} = \pi^{+}\eta = \pi^{+}\lambda L$

where η is the same of T. This model allows for statistical inference on the dimenrate λ [Whitemess and Gauge (1990); Batanity et al. (2000); Batanity et al. (2000); Batanit, Yang and Bataney (2000); Beenere, the andoid is only simble for the case of operfact specificity, i.e. $\tau^{-1} = 1$, but imperfect samithity, this is, $\mu^{-1} = 1$. (Channers and Thirefu (1990) p_{-} 207-2112; A we have, there is not as exploit material which can account of the same, there is not as exploit imperfect smultity and specificity. In this thesis, more of our objectives is to develop with a model.

Classic approaches for analysis of longitudinal data are often based on the assumption that there are no measurement errors in the observations. In practice, it is often not the case. Therefore, there exist literatures studying the adverse effects of measurement errors. Most of them are about the measurement errors in covarision. Fuller (1997) combards an entropic discussion allocations measurement errors. models, and Carroll et al. (2000) investigated measurement errors in nonlinoar models. Also Stefanski and Carroll (1985) and Stefanski (1985) studied the effects of min-measured covariatos in guerralized linear models, opecially for legistic model for binary data [also see Scheft (1987); Speinghman, Romer and Lagun (2000); Hosnin and Gustafon (2000); Rubel-Belseth, Peides and Sherodal (2000).

There are also some literature focusing on the minementer represents. For eangle, Cantadia (1997), 2003, Hoy, Rangier and Maila (1998), Hoy and Baserjae (2008), Borchuk (1998), Roychuk and Thomyson (2004). Roychuk and Main (2004) and Neaham (1998), Roychuk and Thomyson (2004). Roychuk and Main (2004) and Neaham (1998), 2003 (2004) and 2004 and 2004 (2004) and 2004 (2004) and Neaham (2002) (2004) are are about effects of anisolated minimum (2004) and 2002 (2004) are are about and an ionimization lass for solving regression coefficient for minicalized humay mapping. The has have based to may norm conclusion to strong degress in half-hadron deradies. The entre that strueation is the simulation degress in half-hadron deradies. The entre hast to may pound, for example the Bayonia matched [2004], McGitzkin, Shanny and anna, (2008), Roychuk and Bahan (2006), SMLX metelo [Xiebouth, Metelli and Londre (2006)] and these expected seminating equations matched by Wanger (2009) discussed a anathelianed approach which as used to dot with the indensified Hassy response, while complexes based to an anternet weres.

In this thesis, we develop the corrected GEE, quasi-likelihood and maximum likelihood methods to handle the statistical inference on mis-measured longitudinal categorical and count data.

1.3 Objective of This Thesis

An entrational before, measurement stress which some in constraints and responses metatodis in reprimeting, medicion, secondi, and stockletti. Strikpel 'parcent measurement armone in other constraints or anyonous back to kined statistication of model parameters and han of power in doctoring interesting associations meng univliable. However, there are is doctoring interesting associations meng univdicated back and the statistical asymptotic back to the undocreted time responses on which the interesting associations or offstuff. Another parallel remembers on which the interesting associations or offstuff. Another parallel remembers the statistical experiment of the interesting associations are offstuff. Another parallel remembers the discussion of the interesting associations are staffinded. Another parallel remembers the discussion of the interesting associations are staffinded. Another parallel remembers the interesting association of the interesting associations and the interesting and the interesting association of the interesting associations are interesting associations and the interesting association of the interesting associations are interesting associations and the interesting association of the interesting associations are interesting associations are interesting associations and the interesting associations are int

There are two main objectives of this thesis. The first one is to develop explicit measurement error models for categorical data and count data which can clearly describe the relationship between the inherent responses and the observed responses. The other objective is to develop approaches to consistently estimate the unknown model summatries for longitudinal rationguidad and encound tata.

The remains of this thesis is equation in differs. In Chapter 2, we introduce guarantimed thinking operation which can be used to don'the the transition between probability of the control of the thinking operation, as explicit model is proposed to characterize measurement error due to industinguistication in exspectical data or multicational data. The explicit model can be used to don'the the biological data or multicational data. The explicit model can be used to don'the data or model hasps in the simple dondparent of institutional approximation models, the new model hasps in the simple dondparent of institution approximation approximation of uniform from the data in the institution of the simple dondparent of institutional proposed new models approximation of the simple dondparent of institution approximation approximation of the simple dondparent of institution approximation approximation of the simple dondparent of institution approximation approxi proposed in Chapter 2 to describe the error-contaminated count data. These two count error models can accommodate both the oversumerated and undersumberated count data.

An methanism before, the effects or associations of interest are solved during on the true responses. For example, dietts of covariants and association with part extension may be included in at samairies models for dynamic entrymeisten. Therefore, we interolore some flort during the transmission models of expansion. Categorical data is very fielded and it can associated are solved as the sama secondaries without here and maintener transition models. For count data, we well prepares a new statisticary A(11) model and a linear transition model which have well experiments in best transmission and which have best experiments in the structure of the same statisticary and which have best experiments in presents.

For the mischandled longitudinal categorical data, we consider a scalinar truesition model for true categories of models and the adjustic main-fassification models for the inclinational between the histart response and the adjusted of the models of the the inclinational between the histart response and the adjusted of the adjustic main strength entries of model parameters. In the first model of the OOGL appends, we use information from both the first and second order response to gain higher diffusions; on the parameters the transmitter, the OOGL appends, we use information [2013] adjustic to be estimated and adjustmetres. In practice, the data may contain the single information from our participant, "many" responses to a specific quarkine. This can be modeled by the unbialized the impaction data strength quarkine. This can be modeled by the unbialized instandiation models. The force is a of categories of the including of the impaction data strength quarkine, the simulation the models of the impaction data strength quarkine, means and a specific type of white here the simulation of the In Chapter 5, we apply the binomial count error model and the corrected additive error model to the sine-mounted harpitabilite card atch. In Sectoria 52, see we a combination of the binomial count error model, a binomial model for two counts of dimense counts, and the hore transmiss model for (down counts), and the transmission of the down of the probability of the transmission of the down of the probability of the down of the probability and the prophysical massive. The GE and GL approximation are employed notamizet the model parameters. Finally, we present the conclusions and future studies in Chapter 4.

Chapter 2

Classification Error and Count Error Models

2.1 Overview

Momenness errors in directs vanishabs soully take the level of michikukhikuka, the computs, a paire indirectly a starkma single to michikukhi take the briekly graup and an individual who is free of arthma may be mindingstomed as an arthma raw. When the vanished of interest and its observation mare built comprised, built indirections are resisted on indipendent of the laberest compression when the dimension beam indipendent of the laberest compression when the dimension beam indipendent of the laberest compression when the stars are discussed interpretentions of the relationship between T and Y can be downline the value production. The heat-baseling between T and Y can be downline the T and R = 0 much that analyzes from the T the downline downline the T shows T = 0 models where $(\tau_{T_{total}}, \cdots, \tau_{T_{total}})$, where T and T and T and R = 0 much that analyzes from the T the downline downline the T shows T and T shows T ($\tau_{T_{total}}, \cdots, \tau_{T_{total}}$).

| | true category (Y) | | | |
|----------------------|-------------------|---------------|--|--------------------|
| observed category(T) | 1 | 2 | | r + 1 |
| 1 | τ_{11} | π_{12} | | π _{1,s+1} |
| 2 | τ_{21} | 7722 | | $\pi_{2,s+1}$ |
| | | | | |
| s + 1 | $\pi_{s+1,1}$ | $\pi_{s+1,2}$ | | $\pi_{s+1,r+1}$ |

Table 2.1: Misclassification from true category T into observed category Y with unequal numbers of categories.

 π_{uv} is the probability that a number of the uth category is classified into the ath category. In general, r = a, which implies the numbers of the observed categories and the true categories are the same. But sometimes it might be the case that r > s or r < s. Some sumplex are given in Section 2.3 in this charter.

. We now define a matrix consisting of the classification probabilities in Table 2.1 as follows:

 $\tilde{\Pi} = \begin{pmatrix} \pi_{11} & \pi_{12} & \cdots & \pi_{1,r+1} \\ \pi_{21} & \pi_{22} & \cdots & \pi_{2,r+1} \\ \vdots & \vdots & \ddots & \vdots \\ \pi_{r+1,1} & \pi_{r+1,2} & \cdots & \pi_{r+1,r+1} \end{pmatrix}.$ (2.1)

In the throny of stochastic precesses, T represents the state of a process at a specific time point j and T be the state of this precess at a time point k after j. The matrix $\overline{\Pi}$ can be used to model the dynamic transition of this precess from time j to k. In this case, $r = s_{j}$, and $\overline{\Pi}$ is the so-called transition matrix.
In the classification context, we refer to the matrix $\hat{\Pi}$ as the full mixclassification matrix (FMC matrix) due to the fact that $\sum_{i=1}^{N} \tau_{im} = 1$ for any v = 1, ..., v + 1. This also implies that the mixclassification from T to Y can be completely characterized by a simplified matrix Π obtained by defining the last row from $\hat{\Pi}$, since $\tau_{s+1,c} = 1$ $-\hat{\Sigma}$, τ_{m} . The matrix Π can be given by

$$\Pi = \begin{pmatrix} x_{11} & x_{12} & \cdots & x_{1,r+1} \\ x_{21} & x_{22} & \cdots & x_{2,r+1} \\ \vdots & \vdots & \ddots & \vdots \\ x_{d1} & x_{d2} & \cdots & x_{d,r+1} \end{pmatrix}, \quad (2.2)$$

and it is named the mixelassification matrix (MC-matrix). We rewrite MC matrix as $\Pi = [\pi_1, \pi_2, ..., \pi_{r+1}]_r$ where π_i is a column vector of dimension π_i for i = 1, ..., r+1. Let $\Pi_i = [\pi_1, \pi_2, ..., \pi_r]$ be assumantic of Π with the last column π_{r+1} dicted from Π . Then Π_i describes the classification from the first r categories of T to Y, and π_{r+1} refers the classification from the r (+1) descriptor by T.

In existing Birenium, the PBZ-entropy is a set to equive the relationship betom the latter variable. The of the multiler spaces V. The chain lense (node in just a descriptor way to characteristic the inclustifying relationship between T and V, therefore, we must it the description inclustifying relationship between T and V, therefore, we must it the description and endowed by formalizing the dynamic relationship between T and V like the chain error model or Bohava error model for minimulant dimensional data. To being the queue, we propose not on explicit minimulations (EMC) model which a superstanding dimensioner and the theory inclusion (EMC) model which a superstanding dimension protection dimensioner dimensioner and contrast model be non-zero superstand dimining experiments of theory. in the next section.

2.2 Generalized Thinning Operation

In integret-solute time series, a probabilistic operation softward binomial thinning peratrix in charmo well order: the transmitt reverse variables at different time points. This operation has been proposed by Sineti and Hara (1977) and applied to model time wells by Molecuin (1988), 1989, 1989; For a direct readow weight of the dimension of the solution of the second s

As a generalization of the bisonali taining operation, a multihousid himing operation has been introduced by McKennik (1995) and 2003) to develop the vectorvalued rans series models. It is also denoted by u, if $u = (u_1, u_1, \dots, u_l)^2 + a$ -adimensional vector of probabilities with $\sum_{i=1}^{l} p_i \leq 1$, and N is a mompative integravalued random vectories of probabilities with $\sum_{i=1}^{l} p_i \leq 1$, and N is a mompative integravalued random vectories of models and u > u > u is a defined to be a random vector in in independent and detected triats where the probability of such as automous of tasks in the 1, N_{i} -metric $\frac{1}{2}$, $U_{j} = 0$ (which implies $u \neq u > 0$, where $(U_{j}) \stackrel{(N)}{\cong} M$ is momental. and 0 is the vector of parso. Available, for a measure n_{ij} were marginalized fields $m = N_{lower} = 0$, where $i = -N_{lower}$ model, where $i = -N_{lower}$ model, where $i = -N_{lower}$ model, where $i = -N_{lower}$ that there will be $N - 12^{i}$ unleged to the size distribution of the line $i = \log m p + 1$ in fields that all of those n and plotter are distribution. If the n + 1 model is the interval N = -1 is the line of the line $n + \log m p + 1$ of the n + 1 model on the line $n + \log m p + 1$. The model n = 1 model is the interval N = N - 1 model is the n + 1 model is the n + 1 model n + 1is the line n + 1 model is the n + 1 model n + 1 model n + 1 model n + 1 n + 1 model n + 1 n + 1 model n + 1 n + 1 model n + 1 n + 1 model n + 1 n + 1 model n + 1 n + 1 model n + 1n + 1 model n +

$$U_{s \times 1} = \Pi * N = \sum_{i=1}^{r} \pi_i * N_i.$$

In this thesis, we let A_{max} denote a matrix A of dimension $m \times n$.

To comprehensively use the multinomial thinning operation + to model the relationship between multiple categorical variables, we further generalize its definition, sepecially in form of matrix, as

Def. 2.1 $\pi_{s\times 1} * N_{1\times 1} \triangleq U_{s\times 1}$ where $U = \sum_{i=1}^{N} U_i$ and $U_i \stackrel{\text{ref}}{\sim} Maltnomial(1, \pi)$. The notation \triangleq means "is defined as".

Def. 2.2 $\Pi_{ssr} * N_{r\times k} \triangleq \sum_{i=1}^{r} (\pi_i * N_{i1}, \pi_i * N_{i2}, \dots, \pi_i * N_{ik})_{s\times k};$

Def. 2.3 If N_{exturn} is a three-dimension array, the matrix in its *j*th folder is

$$(N_{ij})_{resk} = \begin{pmatrix} N_{11j} & N_{12j} & \cdots & N_{1kj} \\ N_{21j} & N_{22j} & \cdots & \pi_{2kj} \\ \vdots & \vdots & \ddots & \vdots \\ N_{r1j} & N_{r2j} & \cdots & N_{rkj} \end{pmatrix}$$

where N_{urj} is a scalar. Let $U_{solvom} = \prod_{str} * N_{rolsom}$ the matrix in the *j*th folder of three-dimensional array U is

$$(U_{,j})_{sek} = \Pi_{ser} * (N_{,j})_{rek}$$

= $[\Pi_{ser} * (N_{,j})_{rel}, \Pi_{ser} * (N_{2j})_{rel}, \dots, \Pi_{ser} * (N_{kj})_{rel}]$

Some special cases are given as follows:

- 1. If k = 1, the formula in Def. 2.2 becomes $\Pi_{aver} * N_{r\times 1} \triangleq \left(\sum_{i=1}^{r} \pi_i * N_i\right)_{r\times 1}$.
- If r = 1, the formula in Def. 2.2 becomes π_{s+1} + N_{1+k} ≜ (π + N₁, π + N₁,..., π + N_k)_{s>k}, where N = (N₁, N₂,..., N_k) and N_k's are scalars.

Actually, the two-dimensional matrix N_{ret} can be viewed as a reduced form of the three-dimensional matrix N_{retax} in the case that three is only one folder, i.e. m = 1. Therefore, the Def. 2.3 can accommodates the Def. 2.2. Furthermore, the guaranised thinning operation - accommodates the coffmany multihomial thinning operation according to the definition above haves further to the biomain thinning operation.

From the definition, it can be seen that the generalized thinning operation is similar to the multiplication product for matrices in the operation rules. In addition, the multimonial thinning operations have good properties and some are given below. π *(N₁+N₂) ^d/_d π *N₁+π *N₂, where N_i, i = 1, 2 are non-negative integer-valued scalars, and the notation ^d/_d means "identical in distribution".

2.
$$\Pi_{axr} * (N_{rxk} + M_{rxk}) \stackrel{d}{=} \Pi_{axr} * N_{rxk} + \Pi_{axr} * M_{rxk}$$

3. If
$$Z_{x\times k} = \prod_{x\times r} * Y_{r\times k}$$
 and $Y_{r\times k} = \Gamma_{r\times m} * X_{m\times k}$, then $Z = \Pi * (\Gamma * X) \stackrel{d}{=} (\Pi \Gamma) * X$

It is straightforward to prove the first two properties from the dedution. Here, you for get the physicalization of the third case. We first show that the metric $\Lambda_{m,m} = 0$ $R_{m}\Gamma_{mn} = 0$, we are a MC-matrix in the operation. Let $\Lambda = \{\lambda_1, \lambda_2, \ldots, \lambda_d\}$, $\Pi = \{\mu_1, \mu_2, \ldots, \mu_d\}$, and $\Gamma = \{\mu_2, \mu_3, \ldots, \mu_d\}$, where λ_1, μ_2 and μ_1 's are relative to stress of physicalization. The same of the physicalities μ_2 and λ_1 metry there that exercises J are also being the physical transition μ_2 and λ_2 , metry the physical stress stress of the obstimute Λ_2 and Λ_2 by Λ_2 is Λ_3 , we will show that the sum of end channel of Λ that is, $\Gamma_{2n} \leq 1$ and $\Gamma_{2n} \leq 1$. Note, we will show that the sum of end channel of Λ that Λ_2 .

We have

$$\Lambda = \prod_{xxx} \Gamma_{r\times m} \Leftrightarrow \lambda_j = \sum_{u=1}^{r} \pi_u \gamma_{uj} \Leftrightarrow \lambda_{ij} = \sum_{u=1}^{r} \pi_{iu} \gamma_{uj}$$

The sum of elements in jth column of A

$$\mathbf{1}' \lambda_j = \mathbf{1}' (\sum_{u=1}^r \pi_u \gamma_{uj}) = \sum_{u=1}^r \mathbf{1}' \pi_u \gamma_{uj} \le \sum_{u=1}^r \gamma_{uj} = \mathbf{1}' \gamma_j \le 1.$$

Therefore, A can work as a MC-matrix in the thinning operation.

Next we prove that the third property is true for vectors Z_{xx1} , Y_{xx1} and X_{xx1} . We use the moment generating function (mgf) technique to prove that $Z \stackrel{d}{=} \Lambda * X$.

Given that Y = y, $Z = \Pi * Y = \sum_{u=1}^{n} W_u$, where $W_u = \pi_u * y_u$ is a variate of Multinomial(y_u, π_u). It is clear that W_u 's are independent of each other given Y = y for u = 1, 2, ..., r, and their moment generating functions (mgf) are given by $M_{W_i}(t) = (1 - 1^i \pi_s + \pi'_s e^i)^{a_i}$. Similarly, given $X = x, Y = \sum_{j=1}^{\infty} U_j$ and $U_j = \gamma_j * x_j \sim Multinomial(x_j, \gamma_j)$ for j = 1, 2, ..., m are independent of each other with mgf $M_{\beta_j}(t^*) = (1 - 1^i \gamma_j + \gamma'_j e^{i^*})^{\gamma_j}$. Therefore, given X = x, the mgf of Z is given by

$$\begin{split} & \max_{\{0,1\}}(i) = B(P^{0}(Y,X) = i) \\ & = B(P^{0}(Y,X) = i) \\ & = B(B(P^{0}(Y,X) = i) \\ & = B(B(P^{0}(Y,X) = i)) \\ & = B(B(P^{0}(Y,X) = i)) \\ & = B(P^{0}(P^{0}(Y,X) = i)) \\ & = B(P^{0}(P^{0}(Y,X) = i)) \\ & = B(P^{0}(Y,X) = i) \\ & = B(P^{0}(Y,X$$

This means $Z \stackrel{d}{=} \Lambda * X$.

As far as the matrices Z_{xxk}, Y_{rxk}, X_{mxk} are concerned, according to the Def. 2.2,

$$Z_{xxk} = \prod_{xxr} * Y_{rxk} = [\Pi * Y_1, ..., \Pi * Y_k, ..., \Pi * Y_k],$$

and

$$Y_{rak} = \Gamma_{ram} * X_{mak} = [\Gamma * X_1, ..., \Gamma * X_r, ..., \Gamma * X_k].$$

It naturally follows that Z's ith column vector $Z_i = \Pi * Y_i \stackrel{d}{=} (\Pi\Gamma) * X_i = \Lambda * X_i$, hence $Z \stackrel{d}{=} \Lambda * X$.

Note: In order to describe the flat transitions between the three categorial wire also X_i was X. We denote the complete chandication indicator vectors of X_i Yand Z by $Z = (X_i - (X_i)Y_i = 0^{-1}) - (Y_i)^{-1}$ and $Z = (X_i - 1)Z_i$, measuring). We further denote that the corresponding FMC-matters see $\overline{\Pi}$ and $\overline{\Gamma}$, respectively. The sum of elements in each column of $\overline{\Pi}$ and $\overline{\Gamma}$ is equal to I. Therefore the transition from $\overline{X} \cap \overline{Y}$ is expressed as $\overline{\Pi} = \overline{\Gamma}$, \overline{Z} , which is equivalent to

$$Y = \Gamma * \overline{X} = \Gamma_m * X + \gamma_{m+1} * (1 - 1'X),$$

Similarly, the transition from \hat{Y} to \hat{Z} is given by $\hat{Z} = \hat{\Pi} * \hat{Y}$, and it is equivalent to

$$Z = \Pi * \hat{Y} = \Pi_r * Y + \pi_{r+1} * (1 - 1'Y)$$

Hence, the transition from X to Z can be fully described as $\overline{Z} = \overline{\Pi} * (\overline{\Gamma} * \overline{X}) \stackrel{d}{=} (\overline{\Pi} \widetilde{\Gamma}) * \overline{X}$. In the Def. 2.1, let $Y = \pi * N$, it is easy to obtain the conditional expectation and variance of Y given N are given by

$$E(Y|N) = \pi N,$$
 (2.3)

$$Var(Y|N) = V_*N$$
, (2.4)

where V_{π} is defined as a diagonal matrix derived from a vector π , that is, $V_{\pi} \triangleq \text{diag}(\pi) - \pi \pi'$. Similarly, in the Def. 2.2, let $Y_{n+1} = \Pi_{n+r} * N_{r+1}$, the expectation

and variance of Y given N can be given by

$$E(Y|N) = \Pi N$$
, (2.5)

$$Var(Y|N) = \sum_{i=1}^{n} N_i V_{m_i}.$$
 (2.6)

Similar to the Kronecker product for matrices in algebra, we also define the Kronecker thinning operation @ which may be useful in the future development based on the generalized thinning operation. The Kronecker thinning operation is defined as

where π is a vector of dimension s_i and N is a matrix of dimension $m \times k$. Specially, when m = 1, $\pi_{n+1} \oplus N_{1\times k} = \pi_{n+1} \star N_{1\times k}$

-2.

$$I \otimes n \triangleq (\pi_1 * n, \pi_2 * n, ..., \pi_r * n)_{e\times r}$$

where $\Pi = [\pi_1, \pi_2, \cdots, \pi_r]$ is a $s \times r$ matrix, and n is a non-negative integer.

3.

$$\begin{split} \Pi \circledast N \triangleq \left(\begin{array}{cccc} \Pi \circledast n_1 & \Pi \circledast n_{12} & \cdots & \Pi \circledast n_{14} \\ \Pi \circledast n_2 & \Pi \circledast n_{22} & \cdots & \Pi \circledast n_{24} \\ \vdots & \vdots & \ddots & \vdots \\ \Pi \circledast n_{n3} & \Pi \circledast n_{n3} & \cdots & \Pi \circledast n_{nk} \end{array} \right)_{servit} \end{split}$$

where Π is a $s \times r$ matrix, and N is a $m \times k$ matrix.

Example:

Suppose that we have longitudinal categorical data Y_0 of dimension s and its inherent variable T_{ij} of dimension r for i = 1, 2, ..., I and j = 1, 2, ..., I. We denote that $\tilde{T}_{ij} = (T_{ij}^0, 1 - 1T_{ij})'$ and $\tilde{Y}_{ij} = (Y_{ij}^0, 1 - 1Y_{ij})'$. There are three different ways to denote the minimization between the bosered and the inherent response.

Case I: If our focus is on the transition between Y_{ij} and T_{ij} , it can be written as

$$\begin{array}{rcl} Y_{ij} &=& \Pi * \widehat{T}_{ij} \\ &=& \Pi_r * T_{ij} + \pi_{r+1} * (1 - \mathbf{1}^r T_{ij}) \\ &=& \sum_{u=1}^r \pi_u * T_{ij}(u) + \pi_{r+1} * (1 - \mathbf{1}^r T_{ij}). \end{array}$$

or in an alternative way.

$$\widehat{Y}_{ij} = \widehat{\Pi} * \widehat{T}_{ij}.$$

Case II: If we are interested in the transition between $(Y_i)_{i \in I}$ and $(T_i)_{i \in I}$, we can

write it as

$$Y_i = \Pi * \widehat{T}_i = (\Pi * \widehat{T}_{i2}, \Pi * \widehat{T}_{i2}, \dots, \Pi_i * \widehat{T}_{iJ})_{e \times J},$$

or

 $\hat{\Sigma} = \hat{\Pi} * \hat{T}_{i}$

where $Y_i = [Y_{i1}, Y_{i2}, ..., Y_{iJ}]$ and $\widehat{Y}_i = [\widehat{Y}_{i1}, \widehat{Y}_{i2}, ..., \widehat{Y}_{iJ}]$, similarly, $T_i = [T_{i1}, T_{i2}, ..., T_{iJ}]$ and $\widehat{T}_i = [\widehat{T}_{i1}, \widehat{T}_{i2}, ..., \widehat{T}_{iJ}]$.

Case III: If we are interested in the transition between Y_{sxJxJ} and $T_{r\times JxJ},$ then we

have

 $Y = \Pi * \widehat{T}$,

$$\hat{Y} = \hat{\Pi} * \hat{T}$$
,

where the matrix in the ith folder of T, \hat{T} , Y and \hat{Y} are, respectively, T_i , \hat{T}_i , Y_i , and \hat{Y}_i .

2.3 Classification Error Models

In this extens, we introduce a confliction error model with an explicit expression based on the generalized thinning operation. Let Y_{n+1} and T_{n+1} represent the observed and the interest multimodial variables, were the -N Additionality(N_j). Then $\eta \triangleq E(T) = N_T$. Let $\tilde{T} = (T, N-1T)$ and $\tilde{Y} = (P, N-1T)$ denote the full vectors of classification variables T and Y. The unic-ionification model for multimonial data can be expressed as

$$Y = \Pi * T = \Pi_r * T + \pi_{r+1} * (N - \mathbf{1}^{*}T).$$
 (2.7)

In this thesis, we refer to model (2.7) as explicit multinomial mixelassification (BMMC) model for multinomial data constantiated with classification errors. For the mixelassified enterprised data with N = 1, model (2.7) in named the explicit mixelassification (EMC) model. For binomial variable T and Y, that is r = s = 1, we refer to model (2.7) run the explicit binomial mixelashies (EBMC) model.

Based on the thinning operation, we can also build the marginal EMMC models for each element of Y. For the jth element Y_{j_2} it can be written as

$$Y_j = \sum_{n=1}^{r} \pi_{jn} * T_n + \pi_{j,r+1} * (N - \mathbf{1}^t T),$$

where $\Pi_{j} = (\pi_{j1}, \pi_{j2}, ..., \pi_{jr+1})^r$ is the vector composed by elements in the jth row of Π . The marginal model for the reported count \widehat{Y}_{s+1} of the subjects classified into the (s + 1)th observed category of Y is given as

$$\tilde{Y}_{s+1} = \sum_{n=1}^{r} \pi_{s+1,n} * T_n + (1 - \mathbf{1}' \pi_{r+1}) * (N - \mathbf{1}'T).$$

Notice that in the joint misclassification model (2.7), $\tilde{Y}_{s+1} = N - \mathbf{1}'Y$.

In should be pointed out that the e amoption minimization module are not compare to denote the minimization between the two compared valuations). The is because, given T, the Y/s from the magnetial module are independent and there are an on M_{12} and M_{12} may be a the magnetial module are independent by M_{12} and M_{12} may be a the short module of M_{12} . For example, in the energe of M = 1denotes M_{12} and M_{12} may be a the short mode M_{12} may be also variable. However, if we are only intermedial is a specific entrapy, for sample, the denotes M_{12} many M_{12} may be a simple of the denote has more many subjects are classified into this company. The β_{12} has anglinal model can completely denote the denolativation of all of the strengthen into the strengthen M_{12} .

Generally, the mean and variance of a misclassified categorical or multinational variable may be used in developing estimating equations to estimate model parameters. So, we give the mean and variance of Y as follows:

$$\mu = E(Y)$$

 $= E[\Pi_{e} * T + \pi_{r+1} * (N - 1^{*}T)]$
 $= N \pi_{r+1} + (\Pi_{e} - \pi_{r+1}Y)\eta$
 $= N[\pi_{r+1} + (\Pi_{e} - \pi_{r+1}Y)p],$ (2.8)

where $\eta = E(T) = Np$, and

$$'ar(Y) = E[Var(Y|T]) + 'ar(B|X|T]]$$

 $= E[\sum_{i=1}^{N} V_{i}\pi_{i}^{2}] + Var([\Pi_{i} - \pi_{i+1}T]T)$
 $= \sum_{i=1}^{N} V_{i}\pi_{i}\pi_{i} + V_{i}\pi_{i+1}(N - Ti) + (\Pi_{i-1} - \pi_{i}T)Var(T)(\Pi_{i-1} - \pi_{i}T)'$
 $= \sum_{i=1}^{N} N_{i}\pi_{i}\pi_{i} + NV\pi_{i}\pi_{i+1}(-1Y)$
 $+ N(\Pi_{i} - \pi_{i}T)^{2} f_{i}^{2} (\Pi_{i} - \pi_{i}T)'.$ (2.9)

Let $q = \pi_{r+1} + (\Pi_r - \pi_{r+1}X)p_i$ it can be shown from the following mgf of Y that the error-prone variable Y also follows a multinomial distribution with a probability vector q, that is $Y \sim Multinomial(N, q)$. The mgf of Y can be calculated as

where q is the vector of multinomial probabilities of dimension s.

| | True Level (T) | | | |
|----------------------|----------------|------------|-------------|--|
| Classified Level (Y) | H (1) | M (2) | L(3) | |
| H (1) | π_{11} | π_{12} | π_{13} | |
| M(2) | π_{21} | π_{22} | π_{23} | |
| L(3) | π_{21} | π_{32} | T 33 | |

Table 2.2: Balanced misclassification of air quality in an environmental study.

Therefore, it follows that

$$E(Y) = Nq.$$
 (2.10)

 $Var(Y) = NV_{y}$ (2.11)

Comparing the expression (2.9) with expression (2.11), V_q should be equal to $\sum_{i=1}^{r} V_{q_i} p_i + V_{q_i} \dots (1 - Y_0) + (\Pi_r - \Psi_{r+1}Y)V_r (\Pi_r - \Psi_{r+1}Y)'$.

Generally, the true wardsh T and the observed washeb Y is the EDMC model (2) drives have special mathemet of comparises (x + 1 = x + 1), herein this for x - x. We refer to this case as the balanced minimum factometrization of the star wave star of the star interaction of the designation data due to imported measurement instruments and proceeders.

In some cases, r < s, which implies that there are more observed categories than the true categories. We name this type of classification as the unbalanced misclassification (UBMC). For example, in diagnosis of a kind of epidemic disease given in Table

| Diagnosis of test (Y) | Disease (T) | | | |
|-----------------------|--------------|--------------|--------------|--|
| | Positive (1) | Negative (2) | Suspected(3) | |
| Infected (1) | T 11 | T 12 | π13 | |
| Healthy (2) | 7 21 | T 22 | π_{23} | |

Table 2.3: Unbalanced misclassification with r < s in an example of asthma study.

2.3. an individual may be diagonal or Painting" which mass "interted", or "Weight with the "which hights," struct, "diagonal "waves diagonal waves "waves the interted on the interted waves and the structure that and the structure of the structure that and the structure of the structure the structure of the structure and the structure of the structure and the structure of the structure and the structure and the structure of the structure of the structure and the structure of the structure and the structure and the structure of the structure of the structure of the structure and the structure and the structure of the structure is structure of the structure in the structure of the structure of

In addition, model (2.7) can be used to describe the relationship between latent genetypes and manifest phenotypes. For example, it is known that there are four blood types for humans, that is, A, B, AB and O, and there types of related genes, i.e. A, B and O. All of possible genetype of an individual are AA, AB, BB, AO, BO and OO. In genetic, it is well known that both the genes A and B are dominator

| | Genotype (T) | | | | | |
|----------------|--------------|----|----|----|----|----|
| Blood type (Y) | AA | AB | BB | AO | BO | 00 |
| A | 1 | 0 | 0 | 1 | 0 | 0 |
| В | 0 | 0 | 1 | 0 | 1 | 0 |
| AB | 0 | 1 | 0 | 0 | 0 | 0 |
| 0 | 0 | 0 | 0 | 0 | 0 | 1 |

Table 2.4: Blood types and their genotypes, an example of unbalanced misclassifica-

tion with s < r

the gene O. Therefore, the relationship between the genetypes and blood types can be described by Table 2.4. Actually, in practice, the blood type of an individual may be misclassified due to the mintakes by unexperience examiners, which means some τ_{co} for $\mu \neq may be not zeros.$

From the discussions above, it can be seen that the new EMMC model (2.7) can be widely used to characterize different classification patterns for categorical or multinomial data.

2.4 Count Error Models

In epidemiologic studies, data like the total of patients infected by a kind of epidemic disease in different areas are frequently used to evaluate the environmental effects on population health. These data are often collected by some survivance systems. For example, the new cancer cases by state in USA can be obtained from SEER. But the data are prone to errors due to different reasons mentioned in Section 1.2 of Chapter 1.

Statistical analysis which takes measurement errors in count data into consideration is of gravit interest. However, as mentionse in Chapter 1, there is not yet a count error model which can accommodate both the overnamerated and undermanerated count data for imperfect semilivity and specificity. Therefore, in this section, we develow the measurement error models for count data.

2.4.1 Multinomial count error model

In the previous section, we discussed the charafteria result outfor for any probabilistic data. For the minimum of multi-model data, we assume that the and multimum data. For the minimum data we assume that the experimental expendition of the star of the minimum data in a remarkly are. However, the propulsion size may be auknown in the intercommutmental expension of the star $h = (a_1, b_1)$, where a_2 are the mass and warms or A_1 , respectively. Then the respective of the star of the star of the star of A_1 respectively. The star propulsion count of disease cases Y and the true room T in an equal star with an automa and automate counted in the model be

 $Y = \Pi * \hat{T} = \Pi_r * T + \pi_{r+1} * (N - 1^{t}T)$ (2.12)

We name this model as the multinomial count error model for a random N.

From expressions (2.11) and (2.12) in section 2.3, we derived the conditional expectation and variance of Y given the population size N to be:

$$E(Y|N) = N[\pi_r + (\Pi_r - \pi_{r+1}\mathbf{1}')p],$$

$$Var(Y|N) = \sum_{i=1}^{r} NV_{\pi_i}p_i + NV_{\pi_{r+1}}(1 - \mathbf{1}'p)$$

 $+N(\Pi_r - \pi_{r+1}\mathbf{1}')Var(T|N)(\Pi_r - \pi_{r+1}\mathbf{1}')$

where $Var(T|N) = NV_p$. So the unconditional expectation and variance of Y based on the assumptions on N can be given as

$$\mu = E[E(Y|N)] = \phi[\pi_r + (\Pi_r - \pi_{r+1}\mathbf{1}')p],$$
 (2.13)

and

$$\begin{split} V_{01}(Y) &= E[V_{01}(Y)(Y)] + V_{02}(Y)(Y)] \\ &= E[\sum_{i=1}^{N}N_{iij}(Y) + NV_{iiij}(1 - Y_j) \\ &+ (U_{ii} - U_{ii}(Y)) + (V_{ii}(U_{ii}) - U_{iii}(Y)) \\ &+ V_{02}[N[\pi_{ii} + (U_{ii} - u_{iii}(Y))] \\ &= \phi[\sum_{i}^{N}[L_{ij} + V_{iiij}(1 - Y_i)] \\ &+ (U_{ii} - u_{iii}(Y))_{ij}^{2}(U_{ii} - u_{iii}(Y)] \\ &+ (U_{ii} - u_{iii}(Y))_{ij}^{2}(U_{ii} - u_{iii}(Y)) \\ &+ (U_{ii} - u_{iii}(Y))_{ij}^{2}(V[\pi_{ii} + (U_{ii} - u_{iii}(Y))] \\ &+ (U_{ii} - u_{iii}(Y))_{ij}^{2}(U_{ii}) \\ &+ (U_{ii} - U_{iii}(Y))_{ij$$

Alternatively, if we let $q = \pi_{r+1} + (\Pi_r - \pi_{r+1}T)p$, we can get that, given N = n, $Y \sim$ multinomial(n, q) from the previous section. Therefore, the unconditional variance of Y has another form, that is

$$Var(Y) = E[Var(Y|N)] + Var[E(Y|N)]$$

 $= E[NV_q + Var(qN)]$
 $= \phi V_q + qVar(N)q'.$ (2.14)

and

Now we consider a special case that the population size $N \sim Poisson(\phi)$. Suppose that T_j represents the count of subjects belonging to the jth category among the population in an area, where j = 1, 2, ..., r+1. So gives $N = n, T \sim$ multinomial(n, p)and $T_j \sim h(n, p_j)$, where $\sum_{j=1}^{M} p_j = 1$, hence $\sum_{j=1}^{M} T_j = N$.

The joint moment generating function of T can be developed as follows:

$$\begin{split} & f_{T}(1) &= \mathcal{D}(e^{T_{T}}) \\ &= e^{T_{T}} \\ &= \frac{e^{T_{T}}}{T_{T}} \\ &= e^{T_{T}} \\ &= \frac{e^{T_{T}}}{T_{T}} \\ &= \frac{e^{T_{T}}}{T_{T}}$$

This and implies that, unconditionally, $T_j j = 1, 2, ..., r+1$ are independent Falsen variables and $T_j = Piccons(q_i = \phi_j)$. Similarly, we can also conclude that Y_i the count of onlyster calorided in this of the longery, for $l = 1, 2, \ldots, r+1$, are also independent Poisson variables and $Y_i = Piccons(q_i = \phi_j)$. This means that under the assumptions that the population size N follows a Falseon distribution, the counts of subjects fully in the difference comprises are subsequed or if each other.

In an open area, there are two situations leading to a dynamic population size. The first is migration including immigration and emigration, and the other is the natural gravity of excesses of the population size has be birth and dusts. In both situations, the population size N can be mounted but remains. If a near in interesting the count of peeple who are attached by an epidemic disease, for example, actinas in an open directive, then the population can be partitioned in the rest subspondingentical symmetry of the strength symmetry of the symmetry of the symmetry of the symmetry of the sympetry of the symmetry of the symmetry of the population and N. We will general some smalls for this kind of data in the following paragraphs.

As shown by Figure 2.1, T denotes the true count of infected subjects in an area, and Y be the observed count of reported disease cases from some registration systems. Suppose that, given the population size $N = n, T \sim b(n, p)$, where, p is the true disease rate in this serve. The relationship between T and Y is given by

$$Y = \pi^{+} * T + (1 - \pi^{-}) * (N - T),$$
 (2.15)

where π^+ is the sensitivity and π^- is the specificity in Table 2.5. We can conclude that given $N = n, Y[N = n \sim 6(n_0)$, where $q = 1 - q^- + (q^- + \pi^- - 1))$ is the reported discase rate in this area. The marginal distribution of Y is a Poisson distribution, that $i_1 Y \sim P(\mu - q_2)$. Actually, $T \stackrel{d}{=} p \cdot N$, similarly $Y \stackrel{d}{=} q \cdot N$.

| Reported cases | Disease cases | | | | | |
|----------------|-------------------|-----------------------------------|--------------|----------------|--|--|
| | Healthy $(1 - p)$ | Count $(N - T)$ | Infected (p) | Count (T) | | |
| Negative | 1. | $N - T - (1 - \pi^{-}) * (N - T)$ | $1 - \pi^+$ | $T-\pi^+*T$ | | |
| Positive | $1 - \pi^{-}$ | $(1 - \pi^{-}) * (N - T)$ | π^+ | $\pi^+ \ast T$ | | |

Table 2.5: Example of misclassified disease cases



Figure 2.1: The True and Reported Disease Cases in An Area

41

2.4.2 Corrected additive count error models

$$Y = \pi^{+} * T + (1 - \pi^{-}) * T^{0}$$
 (2.16)

In the expression (2.16), the count of infected subjects being correctly classified $\pi^{+} + T$ is independent of the count of leadily subjects being miclassified into infected category $(1 - \pi^{-}) * T^0$. If we let $e = (1 - \pi^{-}) * T^0$, the new count error model can be rewritten as

$$Y = \pi^{+} * T + e,$$
 (2.17)

where π^+ is the sensitivity. We call this model as the corrected additive model when it is compared with the additive model (1.18) which was discussed by Cameron and Trivedi (1998). Therefore, we can apply the model (117) to model the microscolid disons cares in a more with unknown population into: We compose that the unknown population size fidness a Poisson distribution. This is a popular and researched semantic in the fidness a Poisson distribution. This is a popular and researched semantic in the lass characteristic and the term moder of microscolid semantic of the material popular in any based-degl adout in expectation. In this distribution constrained probably subsponding in the spectra of the number of the spectra of the spectra of the spectra of the spectra of the number of the spectra orang for example, the environmental expression. The addition ence $\sim 10^{-10}$ cm ($^{-10}$ set) we have into the information of the spectra of the spectra of the model of the list the information of the spectra of the spectra of the model of the list in the information of the spectra of the spectra of the spectra of the list in the information of the spectra of the spectra of the spectra of the list in the information of the spectra of the spectra of the spectra of the list of the spectra of the spec

The expectation of Y is given by the following expression:

$$\mu = E(Y) = E[E(Y|T)] = \pi^{+}\eta + \psi,$$
 (2.18)

and the variance of Y is formulated by

$$Var(Y) = Var[E(Y|T)] + E[Var(Y|T)]$$

 $= Var(\pi^*T + \psi) + E[\pi(1 - \pi)T + \psi]$
 $= (\pi^*)^2 Var(T) + \pi^*(1 - \pi^*)\eta + \psi$
 $= \mu + (\pi^*)^2 [Var(T) - \eta].$

It is easy to see that in the corrected additive count error model (2.17), the expectation of Y can be greater than the expectation of T, that is, $\mu > \eta$ when $\psi > (1 - \pi^+)\eta$. On the other hand, the expectation of V and be madler than the expectation of V_1 is the $\mu_{i} < \psi$ show the $(i - i - v)_{V_1}$. Therefore, the current dashifts error model (147) can assummability the thick the current model that with hyperfect matrixing $(v = c_1)$ and impacted specificity ($\phi > 0$). In addition, $|\psi| = 0$ bound et a (this impact) is a straight or $|\psi| = 0$ bound to the shead proper Therefore, this model can accummability the random dashed in both induced properties of the problem straight or $|\psi| = 0$.

It should be painted out that, in the currented additive source more mode (127), the summaption that propagations in different Neurann distribution may be without dware we focus on a specific subspecification. For example, if we have the total population of N is a distribution, but we have note into the leading parameters of the propagation of the state of the propagation in the total population of the propagation of the leading respectively are state of proparial distribution. The distribution of the leading transpectively are state the propagation of the gauge curres T as in the subspectivity are state independent variables. In this situation, the mission of the new bioinfordied by model (121) of when there the propalability ρ .

Hence, can we do not how the distribution of the population air of a mass in a intermedial year is precisely. It is a population and the time of the healthy and the halehed groups are independent of each other. The court of the misreported dimension can among buildy adoptiquicities $c = e^{-x} + 2^{-2}$ is dues an among to be a separational. The heavier would be thing in the second to the population dimension variable to a Phison distribution when the size of the healthy subpopulation is very large. Therefore, the corrected additive energy model (2.17) will applicable when the difficulture of the terminal population when the distribution of the population are in valuess.

Chapter 3

Longitudinal Transition Models for Categorical Data and Count Data

3.1 Transition Models for Categorical Data

3.1.1 A transition model for dynamic categorical data

In this section, we develop a transition model for dynamic categorical data based on the generalized thinning operation. This transition model has similar structure as the excilicit misclassification model (2.7) with N = 1 in Section 2.3 of Chapter 2.

In a longitudinal study, we let $(a_{1}$ -house the baseline absorvation of the comproted variable $T_{ij} = (T_{ij})_{ij}, \dots, T_{ij})_{ij}$, $(f_{ij})_{ij}$ for the hardpect in a longitudinal study, where $i = 1, ..., 1, ..., T_{ij}$, $(i_{ij})_{ij}$, $(i_{ij}$

Table 3.1: Transition probabilities from $T_{i,j-1}$ to T_{ij}

| | T_{ij-1} | | | | |
|----------|------------------------|-------------------------|--|--------------------------------|--------------------------------|
| T_{ij} | 1 | 2 | | τ | r+1 |
| 1 | $\tilde{\eta}_{(l,l)}$ | $\bar{\eta}_{j(1,2)}$ | | $\tilde{\eta}_{ij(1,r)}$ | $1-1'\hat\eta_{ij(1,\cdot)}$ |
| 2 | ι. 19γ(2.1) | $\tilde{\eta}_{j(2,2)}$ | | $\hat{\eta}_{ij(r,2)}$ | $1-1'\hat{\eta}_{ij(2,\cdot)}$ |
| | | | | | |
| r | $\eta_{(0,1)}$ | $\bar{\eta}_{(0+2)}$ | | $\tilde{\eta}_{ (\tau,\tau)}$ | $1-1'\hat{\eta}_{(j(r,\cdot)}$ |
| r+1 | 15 ₍₁₁₎ | \$1(2) | | $\bar{\eta}_{ij(r)}$ | $1-1'\bar{\eta}_{ij}$ |

 $\bar{\eta}_{(0)1} = P(T_{(0)1} = 1)|X_{1_2-1} = 0\rangle$, that is, the probability of the transition from the state r + 1 at time j - 1 to the shi state at time j, where u = 1, 2, ..., r. In addition, the probability of transiting from state u at time j - 1 to state r + 1 at the next time point is equal to $1 - Y_{0_0(u)}$. Similarly, the probability that a subject with state r + 1at time j - 1 larges holes takes a state state time point jait. If M_{ij}

Table 3.1 shows the transition probabilities of the ith subject's state from the (j - 1)th time point to the jth time point. Similar to the misclassification problem for categorical data described in Section 2.3 of the previous Chapter, the full transition matrix is given by

$$\widetilde{\Lambda}_{ij} = \begin{bmatrix} \hat{\eta}_{ij} & \hat{\eta}_{ij} \\ \hat{\Lambda}_{ij} = \begin{bmatrix} \hat{\eta}_{ij} & \hat{\eta}_{ij} \\ \Gamma - \Gamma \hat{\eta}_{ijj} & 1 - \Gamma \hat{\eta}_{ijj} \end{bmatrix} = \begin{bmatrix} \hat{\eta}_{ij}, 1, 2, \cdots & \hat{\eta}_{ij}, 2, \cdots$$

and a simplified transition matrix can be defined as

$$\Lambda_{ij} = [\hat{\eta}_{ij}, \hat{\eta}_{ij}] = \begin{pmatrix} \hat{\eta}_{ij}(1_{ij}) & \cdots & \hat{\eta}_{ij}(1_{ij}) & \hat{\eta}_{ij}(1_{ij}) \\ \hat{\eta}_{ij}(1_{ij}) & \cdots & \hat{\eta}_{ij}(1_{ij}) & \hat{\eta}_{ij}(1_{ij}) \\ \vdots & \ddots & \vdots & \vdots \\ \hat{\eta}_{ij}(1_{ij}) & \cdots & \hat{\eta}_{ij}(1_{ij}) & \cdots & \hat{\eta}_{ij}(1_{ij}) \end{pmatrix},$$
(3.2)

By defining $\widetilde{T}_{ij} = (T_{ij}, 1-1^*T_{ij})$, the new transition model based on the generalized thinning operation can be defined as

$$T_{ij} = \Lambda * \hat{T}_{i,j-1}$$

= $\hat{\eta}_{ij} * T_{i,j-1} + \hat{\eta}_{ij} * (1 - 1^* T_{i,j-1}).$ (3.3)

The first part on the right side of the model (3.3), $\hat{\eta}_{ij} * T_{i,j-1}$, denotes the transition from the first r states, and the second part $\hat{\eta}_{ij} * (1 - \mathbf{i} T_{i,j-1})$ represents the transition from the last state r + 1 at time point j = 1.

We next present some useful results of calculations about expectations, variances, and covariances. Firstly, based on model (3.3), it is easy to see that the conditional expectation of T_{ij} given the previous $T_{i,j-1}$ is

$$E(T_{ij}|T_{i,j-1}) = \tilde{\eta}_{ij}T_{i,j-1} + \tilde{\eta}_{ij}(1 - \mathbf{1}^{*}T_{i,j-1})$$

= $(\tilde{\eta}_{ij} - \tilde{\eta}_{ij}\mathbf{1}^{*})T_{i,j-1} + \tilde{\eta}_{ij}.$ (3.4)

We further have the expectation of T_{ij} given T_{ik} for k < j

$$E(T_{ij}|T_{ik}) = \prod_{u=k+1}^{j} (\bar{\eta}_{u} - \bar{\eta}_{u}\mathbf{1}')T_{ik} + \sum_{l=k+1}^{j} \prod_{u=l+1}^{j} (\bar{\eta}_{u} - \bar{\eta}_{u}\mathbf{1}')\bar{\eta}_{l}.$$
 (3.5)

It should be noticed that, in this section, $\prod_{a=0}^{l} (\hat{\eta}_{a} - \hat{\eta}_{a} \mathbf{1}^{\prime})$ for $k \leq j$ is defined as $(\hat{\eta}_{ij} - \hat{\eta}_{ij} \mathbf{1}^{\prime})(\hat{\eta}_{i,j-1} - \hat{\eta}_{i,j-1} \mathbf{1}^{\prime}) \cdots (\hat{\eta}_{ik} - \hat{\eta}_{ik} \mathbf{1}^{\prime})$ but not $(\hat{\eta}_{ik} - \hat{\eta}_{ik} \mathbf{1}^{\prime})(\hat{\eta}_{i,i+1} - \hat{\eta}_{i,k+1} \mathbf{1}^{\prime}) \cdots (\hat{\eta}_{ij} - \hat{\eta}_{ij} \mathbf{1}^{\prime})$.

The unconditional expectations can be given as

$$\eta_{ij} \triangleq E(T_{ij})$$

= $(\bar{\eta}_{ij} - \bar{\eta}_{ij} \mathbf{1}^*) \eta_{ij-1} + \bar{\eta}_{ij}$ (3.6)

$$= \prod_{u=d+1}^{l} (\hat{\eta}_{u} - \hat{\eta}_{u}\mathbf{1}') \hat{\eta}_{ik} + \sum_{l=d+1}^{l} \prod_{u=l+1}^{l} (\hat{\eta}_{u} - \hat{\eta}_{u}\mathbf{1}') \hat{\eta}_{l}. \quad (3.7)$$

For any k < j, the expectation of the pairwise product $T_{ij}T_{ik}^{r}$ is

$$E(T_{ij}T_{ik}^{*}) = E[E(T_{ij}|T_{ik})T_{ik}]$$

= $\prod_{n=k+1}^{j} (\hat{\eta}_{nr} - \eta_{0r}\mathbf{1}^{*})E(T_{ik}T_{ik}^{*}) + \sum_{l=k+1}^{j} \prod_{r=l+1}^{j} (\hat{\eta}_{tr} - \eta_{rr}\mathbf{1}^{*})\hat{\eta}_{l}q_{ik}^{*}.$ (3.8)

Hence, the covariance between T_{ii} and T_{ik} is

$$Cor(T_0, T_0) = E(T_0, T_0) - E(T_0)E(T_0)$$

 $= \prod_{n=1}^{d} (\delta_0 - \delta_0, T)B(T_0, T_0) + \sum_{n=1+1}^{d} \prod_{n=1}^{d} (\delta_0 - \delta_0, T)\delta_n d_0'$
 $- \prod_{n=1+1}^{d} (\delta_0 - \delta_0, T)B(u_0' + \sum_{n=1+1+n=1}^{d} \prod_{n=1}^{d} (\delta_n - \delta_n, T)\delta_n d_0'$
 $= \prod_{n=1+1}^{d} (\delta_0 - \delta_0, T)Vor(T_0).$ (1.9)

48

For categorical variable $T_{ij} \sim multinomial(1, \eta_0)$, it is obvious that

$$Var(T_{ij}) = V_{\eta_{ij}} = diag(\eta_{ij}) - \eta_{ij}\eta'_{ij}$$
.

If we rewrite $T_i = (T_{i1}^i, T_{i2}^j, ..., T_{iJ}^i)_{i,J \times 1}$ and let $\eta_i = E(T_i)$, the variance-covariance matrix of T_i can be written as

$$\Sigma_{0} = \begin{pmatrix} \Sigma_{01} & \Sigma_{02} & \cdots & \Sigma_{0J} \\ \Sigma_{01} & \Sigma_{02} & \cdots & \Sigma_{0J} \\ \vdots & \vdots & \ddots & \vdots \\ \Sigma_{0J} & \Sigma_{0J} & \cdots & \Sigma_{0J} \end{pmatrix}_{rl \neq rJ},$$
 (3.10)

where $\Sigma_{jk} = \Sigma_{kj}^{s} = Coe(T_{ij}, T_{jk})$ for $j \neq k$, and $\Sigma_{jj} = Var(T_{ij}) = V_{\eta_{ij}}$.

Model (3.3) cm accommodate various transition models based on different assumptions on $\tilde{\Lambda}_{ij}$ or Λ_{ij} . In addition, Λ_{ij} can be a constant matrix over time for any subject, or it can be a matrix function of some covariates, even time-varying covariates. For example, in the full transition matrix Λ_{ij} , we can suppose that

$$\tilde{\eta}_{l}(s,r) = \frac{exp(x'_{lj}\beta_{l} + \gamma_{sr})}{1 + \sum_{l=1}^{r} exp(x'_{lj}\beta_{l} + \gamma_{lr})},$$
 (3.11)

and

$$\eta_{ij(n)} = \frac{exp(x'_{ij}\beta_n)}{1 + \sum_{i=1}^{r} exp(x'_{ij}\beta_i)}.$$
 (3.12)

In expression (3.11), $x_{ij} = (x_{ij(0)}, \dots, x_{ij(p)})'$ is a vector consisting of p explanatory variables. The parameter matrix

$$\mathcal{B} = \begin{pmatrix} \beta_{11} & \cdots & \beta_{1r} \\ \vdots & \cdots & \vdots \\ \beta_{pl} & \cdots & \beta_{pr} \end{pmatrix}$$

45

with $\beta_u = (\beta_{1u}, \dots, \beta_{pu})'$ denoting the effects of covariates, and

$$\Upsilon = \begin{pmatrix} \gamma_{11} & \cdots & \gamma_{1r} \\ \vdots & \cdots & \vdots \\ \gamma_{11} & \cdots & \gamma_{1r} \end{pmatrix}.$$

denotes the dynamic dependence.

Indeed, the transition model (3.3) can accommodate the following model (3.13-3.14) in nature. The latter one is just an alternative form of the special case of model (3.3) based on the assumptions (3.11) and (3.12). The model is riven by

$$\eta_{ij}^{c} = E(T_{ij}|T_{ij-1} = t_{i,j-1}, x_{ij}) = \begin{pmatrix} \eta_{ij}^{c}(z) \\ \vdots \\ \eta_{ij}^{c}(z) \\ \vdots \\ \vdots \\ \vdots \\ \vdots \\ \eta_{ij}^{c}(z) \end{pmatrix}$$
, (3.13)

where η_{ij}^{i} is the conditional expectation given the prior state of the process and current values of covariates. The element of η_{i}^{i} is defined by

$$t'_{p(u)} = \frac{exp(x'_{ij}\beta_u + t'_{i,j-1}\gamma_w)}{1 + \sum_{l=1}^{r} exp(x'_{ij}\beta_l + t'_{i,j-1}\gamma_b)}$$
(3.14)

for j = 1, 2, ..., J, where $\gamma_w = (\gamma_{wl}, ..., \gamma_w)'$ consists of the uth row of matrix Υ . This model leads to more complicated derivation of moments compared with model (3.3).

To address the issue of estimation, we next present the first estimating method based on model (3.3) with assumption (3.11) and (3.12). The interested parameters include all the elements of \mathscr{B} and Υ . Let $\theta = (Vec(\mathscr{B})', Vec(\mathbf{Y})')'$, where Vec is the operation of vectorizing a matrix. The GEE approach [Liang and Zeger, (1986)] has the estimating equations given by

$$\sum_{i=1}^{l} \frac{\partial \eta'_{i}}{\partial \theta} W_{i}^{-1}(t_{i} - \eta_{i}) = 0, \quad (3.15)$$

where W_i is the "working" covariance matrix of T_i . If the true covariance matrix Σ_i in (3.10) is used, the GEE becomes the GQL method [Sutradhar (2003)]. The derivatives involved in $\partial \eta_i / \partial \theta$ are given as

$$\frac{\partial \eta_{0}(z)}{\partial \tilde{h}_{mn}} = \sum_{k=1}^{n} (\frac{\partial \tilde{\eta}_{0,mn}}{\partial \tilde{h}_{mn}} - \frac{\partial \tilde{\eta}_{0,mn}}{\partial \tilde{h}_{mn}}) \eta_{k,\ell-10} + \sum_{k=1}^{n} (\tilde{\eta}_{0,\ell+1}) - \tilde{\eta}_{0,\ell+1}) \frac{\partial \eta_{k,\ell-10}}{\partial \tilde{h}_{mn}} + \frac{\partial \eta_{k,m}}{\partial \tilde{h}_{mn}}$$

$$= \sum_{k=1}^{n} [\tilde{\eta}_{0,\ell+1}(1 - \tilde{\eta}_{0,\ell+1}) - \tilde{\eta}_{0,\ell+1}(1 - \tilde{\eta}_{0,\ell+1})] x_{\ell}(u_{\ell}) x_{\ell}(u_{\ell}) x_{\ell}(u_{\ell})$$

$$+ \sum_{k=1}^{n} [\tilde{\eta}_{0,\ell+1}(1 - \tilde{\eta}_{0,\ell+1}) - \frac{\partial \eta_{k,\ell+1}}{\partial \tilde{h}_{mn}} + \eta_{k,0}(1 - \tilde{\eta}_{0,\ell+1})] x_{\ell}(u_{\ell}), \quad (3.16)$$

$$\frac{\partial \eta_{2}(z)}{\partial \beta_{nm}} = \sum_{k=1}^{n} [\hat{\eta}_{ij}(z_{0})\hat{\eta}_{ij}(z_{1} - \hat{\eta}_{ij}(z_{k}))\hat{\eta}_{ij}(z_{1})]^{2} q_{(0)k} \hat{\eta}_{k}(z_{1}-1)k) \\ + \sum_{k=1}^{k} [\hat{\eta}_{ij}(z_{k},k)(1 - \hat{\eta}_{ij}(z_{k}))] \frac{\partial \eta_{k}(z_{1}-1)k}{\partial \beta_{nm}}, \quad (3.17)$$

$$\frac{\partial \eta_{ij(\mathbf{x})}}{\partial \gamma_{\mathbf{x}k}} \ = \ \sum_{u=1}^r (\frac{\partial \tilde{\eta}_{ij(u,v)}}{\partial \gamma_{uk}} - \frac{\partial \tilde{\eta}_{ij(u)}}{\partial \gamma_{uk}}) \eta_{i,j-1(v)} + \sum_{u=1}^r (\tilde{\eta}_{ij(u,v)} - \tilde{\eta}_{ij(u)}) \frac{\partial \eta_{i,j-1(v)}}{\partial \gamma_{uk}} + \frac{\partial \tilde{\eta}_{ij(u)}}{\partial \gamma_{uk}}$$

$$= \tilde{\eta}_{ij(u,k)} (1 - \tilde{\eta}_{ij(u,k)}) \eta_{i,j-1(k)} + \sum_{u=1}^{n} [\tilde{\eta}_{ij(u,v)} - \tilde{\eta}_{ij(u)}] \frac{\partial \eta_{i,j-1(v)}}{\partial \gamma_{uk}}, \quad (3.18)$$

$$\frac{\partial \eta_{kj(n)}}{\partial \gamma_{nk}} = -\eta_{ij(n,k)} \eta_{ij(n,k)} \eta_{k,j-1(k)} + \sum_{n=1}^{r} [\eta_{ij(n,k)} - \eta_{ij(n)}] \frac{\partial \eta_{k,j-1(n)}}{\partial \gamma_{nk}},$$
 (3.19)

$$\begin{array}{rcl} \frac{\partial \partial_{0}(x,x)}{\partial s_{k,n}} & = \theta_{0}(x,x) \left(1-\theta_{0}(x,x)\right) z_{0}(x_{0}) \\ \frac{\partial \theta_{0}(x,x)}{\partial s_{k,n}} & = -\theta_{0}(x,0) \theta_{0}(x,0) z_{0}(x_{0}), \\ \frac{\partial \theta_{0}(x,x)}{\partial s_{k,n}} & = \theta_{0}(x,x) \left(1-\theta_{0}(x,x)x\right), \\ \frac{\partial \theta_{0}(x,x)}{\partial s_{k,n}} & = 0 & m \neq k \mbox{ for } a_{0}(x,x) \\ \frac{\partial \theta_{0}(x,x)}{\partial s_{k,n}} & = 0 & -\theta_{0}(x,x) \theta_{0}(x,x), \\ \frac{\partial \theta_{0}(x,x)}{\partial s_{k,n}} & = -\theta_{0}(x,x) \theta_{0}(x,x) \\ \frac{\partial \theta_{0}(x,x)}{\partial s_{k,n}} & = \theta_{0}(x,x) \theta_{0}(x,x), \\ \frac{\partial \theta_{0}(x,x)}{\partial s_{k,n}} & = 0 & (maxirs). \end{array}$$

The most efficient estimates of the model parameters can be obtained by applying the maximum likelihood approach. Based on model (3.3), the likelihood function given observations T = t is

$$L(\theta) = \prod_{i=1}^{l} \prod_{j=1}^{d} g_{ij|j-1}, \quad (3.20)$$

where

$$\begin{split} g_{(\beta|j)-1} &= \left[\prod_{u=1}^{r} \prod_{q=0}^{r} \eta_{(q)(u_{d})}^{(l_{1}_{1},l_{1}_{d})} \right] \left[\prod_{u=1}^{r} (1-1'\tilde{\eta}_{(\beta|\ell_{1},d)})^{(1-1'\eta_{d}(l_{1,d}-1))}\right] \\ &\times \left[\prod_{u=1}^{r} \eta_{(q)(0)}^{(l_{1}_{1},l_{1}_{d})} - 1'^{(l_{1}_{1},l_{1}-1')}\right] \left[(1-1'\eta_{0})^{(1-1'\eta_{d}(l_{1}-1')_{d}-1)} \right]. \end{split}$$

Then the log-likelihood function is

$$\begin{split} \ell(\theta) &= \sum_{i=1}^{I} \sum_{j=1}^{J} \sum_{v=1}^{c} \sum_{w=1}^{c} \sum_{(w)=1}^{s} t_{(\psi)(0)} t_{i,j-1i} log(\tilde{\eta}_{(j)(u,v)}) + (1 - 1^{i} t_{ij}) t_{i,j-2i(i)} log(1 - 1^{i} \tilde{\eta}_{(i)(u)})) \\ &+ (1 - 1^{i} t_{ij}) (1 - 1^{i} t_{i,j-1}) log(1 - 1^{i} \tilde{\eta}_{ij}) + \sum_{w=1}^{s} t_{ij(w)} (1 - 1^{i} t_{i,j-1}) log(\tilde{\eta}_{(i)(v)}) . \end{split}$$

where

To maximize the function $\ell(\theta)$, we solve the following equations:

$$\frac{\partial \ell(\theta)}{\partial \beta_{\mu}} = \sum_{i=1}^{J} \sum_{j=1}^{J} \sum_{n=1}^{r} (t_{ij}(n) - \hat{\eta}_{ij(n,i)}) t_{i,j-1(i)} + (t_{ij(n)} - \hat{\eta}_{ij(n)}) (1 - 1^{i} t_{i,j-1})] x'_{ij}$$

$$= \sum_{i=1}^{J} \sum_{j=1}^{m} \sum_{i=1}^{r} (t_{ij(n)} - E(T_{ij(n)}|T_{i,j-1} = t_{i,j-1})] x'_{ij} = 0, \quad (3.21)$$

$$\frac{\partial \ell(\theta)}{\partial \gamma_{bw}} = \sum_{i=1}^{J} \sum_{j=1}^{J} (t_{ij(u)} - \tilde{\eta}_{ij(u,i)}) t_{i,j-1(v)} = 0.$$
 (3.22)

However, the expressions (3.13-3.14) lead to a simpler development of ML approach than model (3.3). In the likelihood function based on (3.13) and (3.14),

$$g_{i,jj-1} = (1 - 1'q_{ij}^{i})^{1-1'u_{ij}} \prod_{u=1}^{i} [\eta_{ij(u)}^{i}]^{t_{ij(u)}}$$

$$= \frac{\prod_{u=1}^{i} exp[\sum_{i=1}^{i} (x_{ij}\beta_u + t'_{ij-1}\gamma_u)t_{iju}]}{1 + \sum_{i=1}^{i} exp(x_{ij}\beta_u + t'_{ij-1}\gamma_u)},$$

and the log-likelihood function is

$$\ell(\theta) = \sum_{i=1}^{I} \sum_{j=1}^{J} \left[\sum_{k=1}^{r} (x_{ik} \beta_j + t_{i,j-1}^{i} \gamma_k) t_{ijk} - log \{1 + \sum_{m=1}^{r} exp(x_{ij} \beta_n + t_{i,j-1}^{i} \gamma_n)\} \right].$$
 (3.23)

This leads to a series of simpler score equations

$$\frac{\partial \ell(\theta)}{\partial \beta_n} = \sum_{i=1}^{I} \sum_{j=1}^{J} (t_{ij(n)} - \eta^i_{ij(n)}) x^i_{ij} = 0,$$
 (3.24)

$$\frac{\partial \ell(\theta)}{\partial \gamma_u} = \sum_{u=1}^{I} \sum_{j=1}^{J} (t_{ij(u)} - \eta_{ij(u)}^c) t_{i,j-1} = 0,$$
 (3.25)

and they are equivalent to equations (3.21) and (3.22).

It can be seen that, in the first order transition model, the psirwise products $t_{i1}t'_{i2}, \dots, t_{ij}t'_{im}, \dots, t_{i,J-1}t'_{iJ}$ along with the first order responses $t_{i1}, \dots, t_{iJ}, \dots, t_{iJ}$ provide sufficient information for the estimation of \mathscr{R} and Υ . This finding implies that estimators based on the second-order GQL (GQL2) may be as efficient as the ML estimators. This suggests the promising application of GQL2 in the case that the full likelihood function is difficult to develop.

3.1.2 The transition model for dynamic binary data

Analogous to model (3.3), the dynamic binary data model can be written as

$$T_{ij} = \bar{\eta}_{ij} * T_{i,j-1} + \bar{\eta}_{ij} * (1 - T_{i,j-1}),$$
 (3.26)

with the baseline observations t_{i0} in a longitudinal study, where i = 1, 2, ..., I, and j = 1, 2, ..., J.

Under assumptions that $\bar{\eta}_{ij} = \gamma_{ij} + (1 - \gamma_{ij})\xi_{ij}$ and $\bar{\eta}_{ij} = (1 - \gamma_{ij})\xi_{ij}$, for j = 1, 2, ..., J, model (3.26) becomes the thinning-operation-based linear transition model of the following linear binary dynamic model given by

$$T_{ij} = b_{ij}T_{i,j-1} + (1 - b_{ij})\epsilon_{ij}$$
 (3.27)

[Tong (1990)]. In model (3.27), it is also assumed that $b_{ij} \sim b(1, \gamma_{ij})$, and b_{ij} is independent of ϵ_{ij} .

Let $\tilde{\eta}_{ij} = \frac{\exp(\zeta_{ij})}{1 \exp(\zeta_{ij}) + i_j}$ and $\tilde{\eta}_{ij} = \frac{\exp(\zeta_{ij})}{1 \exp(\zeta_{ij})}$, model (3.26) will be the thinning-operation-version of the non-linear binary dynamic model which is given by

$$t_{ij}^{i} = P(T_{ij} = 1|T_{i,j-1} = t_{i,j-1})$$

= $\frac{exp(x'_{ij}\beta + t_{i,j-1}\gamma)}{1 + exp(x'_{ij}\beta + t_{i,j-1}\gamma)}$, for j=1,2,..., J, (3.28)

[Amemiya (1985): Manski (1987)]. In fact, $\eta_{ij}^{t} = \bar{\eta}_{ij}t_{ij-1} + \bar{\eta}_{ij}(1 - t_{ij-1})$, is the conditional expectation $E(T_{ij}|T_{ij+1} = t_{ij-1})$ derived from model (3.26). This is a special case of model (3.14) for longitudinal binary data. The mean and variance of T_{ij} based on model (3.26) are given by

$$\eta_{ij} \triangleq E(T_{ij}) = \eta_{i,j-1}(\bar{\eta}_{ij} - \bar{\eta}_{ij}) + \bar{\eta}_{ij},$$
 (3.29)

$$\sigma_{ij}^2 \triangleq Var(T_{ij}) = \eta_{ij}(1 - \eta_{ij}),$$
 (3.30)

and the covariance between T_{ii} and T_{in} is

$$\sigma_{iju} \triangleq Cov(T_{ij}, T_{iu}) = Var(T_{iu}) \prod_{k=u+1}^{j} (\bar{\eta}_{ik} - \bar{\eta}_{jk}), \text{ for } u < j,$$
 (3.31)

where $\eta_{i0} = t_{i0}$.

Becoming the most-linear dynamic model (12.29) was applied by Struchleur and Hermit (1207) is subjustical althous takens data set. They developed there approaches to notizante the model parameters J and γ , manify, the generalized space likelihood (OQL), the second order OQL (OQL2) and maximum likelihood (DA) approaches. Its the OQL2 approach, the parameters of the model and wood order statistics of the responses lists the estimating procedures and advanced likelihood experison is the dependence of the line of the parameters of the properties in the optimum of OQL (OQL2) standard. Use the local of symmetry model (12.7). The default and Static procedures make the the local of symmetries are pion holow.

1. GQL approach

The parameters $\theta = (\beta^{q}, \gamma)^{i}$ are estimated by solving the estimating equations [Sutradhar (2003)]

$$\sum_{i=1}^{l} \frac{\partial \eta_{i}^{\prime}}{\partial \theta} \Sigma_{i}^{-1}(t_{i} - \eta_{i}) = 0. \quad (3.32)$$

Once we have $\hat{\theta}_{COL}$, its covariance matrix can be estimated by

$$\hat{V}(\hat{\theta}_{OQL}) = \left(\sum_{l=1}^{l} \frac{\partial \eta'_{l}}{\partial \theta} \sum_{i}^{-1} \frac{\partial \eta_{i}}{\partial \theta}\right)^{-1}|_{\theta = \hat{\theta}_{OQL}}.$$
 (3.33)

2. ML approach

The likelihood function of the observations $t=\{t_{ij}, i=1,2,\ldots, I, \text{ and } j=1,2,\ldots, J\}$ is given by

$$L(\theta) = \prod_{i=1}^{l} \prod_{j=1}^{d} g_{i,j(j-1)},$$
 (3.34)

where

$$g_{i,j|j-1} = \hat{\eta}_{ij}^{t_{ij}t_{i,j-1}} (1 - \hat{\eta}_2)^{(1-1)t_{ij}|\eta_{i,j-1}} \hat{\eta}_{ij}^{t_{ij}(1-t_{i,j-1})} (1 - \hat{\eta}_{ij})^{(1-t_{ij})(1-t_{i,j-1})},$$

Then the log-likelihood function is

$$\begin{split} \ell(\theta) &= \sum_{i=1}^{l} \sum_{j=1}^{j} \{ t_{ij} t_{i,j-1} log(\bar{\eta}_{ij}) + (1 - t_{ij}) t_{i,j-1} log(1 - \bar{\eta}_{ii}) \\ &+ t_{ij} (1 - t_{i,j-1}) log(\bar{\eta}_{ij}) + (1 - t_{ij}) (1 - t_{i,j-1}) log(1 - \bar{\eta}_{ij}) \}. \end{split}$$

To maximize the function $\ell(\theta)$, we solve the following equations:

$$\frac{\partial \theta(\theta)}{\partial \delta_{\theta}} = \sum_{i=1}^{J} \sum_{j=1}^{J} [(t_{ij} - \tilde{\eta}_{ij})t_{i,j-1} + (t_{ij} - \tilde{\eta}_{ij})(1 - t_{i,j-1})]t'_{ij(u)}$$

$$= \sum_{i=1}^{J} \sum_{j=1}^{J} [(t_{ij} - \eta''_{ij})t'_{ij(u)} = 0, \text{ for } u = 1, 2, ..., p, \quad (3.35)$$

$$\frac{\partial \ell(\theta)}{\partial \gamma} = \sum_{i=1}^{J} \sum_{j=1}^{J} (t_{ij} - \bar{\eta}_{ij}) t_{i,j-1}^{i} = 0.$$
 (3.36)

If one wants to derive the ML method based on model (3.28), the likelihood function is given by

$$L(\theta) = \prod_{i=1}^{J} \prod_{j=1}^{J} g_{i,j|j-1},$$

where $g_{i_2|j=1} = (\eta_{i_j}^c)^{t_{i_j}} (1 - \eta_{i_j}^c)^{1-t_{i_j}}$, and the log-likelihood function is

$$\ell(\theta) = \sum_{i=1}^{I} \sum_{j=1}^{J} t_{ij}(x_{ij}^{i}\beta + \gamma t_{i,j-1}) - \sum_{i=1}^{I} \sum_{j=1}^{J} log[1 + exp(x_{ij}^{i}\beta + \gamma t_{i,j-1})].$$

56

This yields equivalent score equations to (3.35) and (3.36).

The covariance matrix of $\hat{\theta}_{HL}$ can be estimated by

$$\hat{V}(\hat{\theta}_{ML}) = (I_T^{-1})_{\theta = \hat{\theta}_{ML}},$$
 (3.37)

where

$$I_{T} = \left[-E\left\{\frac{\partial^{2}\ell(\theta)}{\partial\theta\partial\theta'}\right\}\right]$$

is the Fisher Information matrix.

3. GQL2 approach

To utilize more information of data, Sutradhar and Farrell (2007) suggested that it would be more efficient to combine the second order statistics of the responses into the estimating procedure. The estimating equations [Sutradhar and Farrell (2007)] are given by

$$\sum_{i=1}^{I} \frac{\partial \varphi'_{i}}{\partial \theta} \Omega_{i}^{-1}(h_{i} - \phi_{i}) = 0, \quad (3.38)$$

where h_i is the observation of $H_i = (T_i^i, H_i^i)$, and $\varphi_i = E(H_i)$. Here $T_i = (T_{i1}, ..., T_{iJ})^i$, and $R_i = (T_{i1}T_{i2}, ..., T_{in}T_{in}, ..., T_{i,J-1}T_{iJ})^i$. Ω_i is the covariance matrix of H_i , that is,

$$Ω_i = Cov(H_i) = \begin{pmatrix} Cov(T_i) & Cov(T_i, R_i) \\ Cov(R_i, T_i) & Cov(R_i) \end{pmatrix}_{J(J+1)/2 \times J(J+1)/2}.$$
(3.39)

The GQL2 approach involves some moments up to order four, say

$$\begin{split} E(T_{ij}T_{ij}) &= \sigma_{aaj} + \gamma_{0i}\eta_{ij}, & law \neq j, \\ E(T_{ij}T_{ik}T_{ik}) &= \sum_{ij} (\prod_{j=1}^{n} \eta_{ij}g_{ij-1})_{i_{ij}=1,i_{ij}=1,i_{ij}=1,i_{ij}=1,i_{ij}}, & lot different j, u, v, \\ E(T_{ij}T_{ik}T_{ik}T_{ij}) &= \sum_{ij} (\prod_{j=1}^{j} \eta_{ij}g_{ij})_{i_{ij}=1,i_{ij}=1,i_{ij}=1,i_{ij}=1,i_{ij}=1,i_{ij}}, & lot different j, u, v, l, \\ \end{split}$$
where $g_{i,ij-1} = (g_{ij}^r)^{i_0}(1 - g_{ij}^r)^{i_d}u_i$ for j = 1, 2, ..., J, and $\sigma_{iaj} = Cov(T_{ij}, T_{ai}) \sum_{i_j^r}$ indicates the summation over all $t_{ai} = 0$ for $k \neq u, v, j$, and similarly $\sum_{i_j^r}$ indicates the summation over all $t_{ai} = 0$ for $k \neq l, u, v, j$.

The covariance matrix of $\hat{\theta}_{COL2}$ is estimated by

$$\hat{V}(\hat{\theta}_{GQE2}) = \left(\sum_{i=1}^{I} \frac{\partial \phi_i^c}{\partial \theta} \Omega_i^{-1} \frac{\partial \phi_i}{\partial \theta}\right)^{-1} |_{\theta = \hat{\theta}_{GQE2}}.$$
 (3.40)

As far as the matrix Ω_c (3.39) is concerned, some authors used a "working" covariance matrix, for example, neurality based covariance [Zhao and Prentice, (1990)], independence covariance [Satradhar, (2003)]. But these working covariance matrix may results in loss of efficiency.

To conduct statistical inference, for example, constructing confidence interval or testing (psychiasis, one needs the asymptotic distributions of these estimators \hat{n}_{OLL} , \hat{n}_{OLL} , and \hat{n}_{ML} . Under some mild regularity conditions [Newsy and McFadden (1990)], is follow from equation (3.23) and (3.38), (3.35-30) that as $1 - \infty$,

$$\sqrt{I}(\hat{\theta}_{OQL} - \theta) \sim N\left(0, \left\{\sum_{i=1}^{J} \frac{\partial u_{i}^{\prime}}{\partial \theta} \sum_{i}^{-1} \frac{\partial u_{i}}{\partial \theta}\right\}^{-1}\right),$$
 (3.41)

$$\sqrt{I}(\hat{\theta}_{OQL2} - \theta) \sim N\left(0, \left\{\sum_{i=1}^{J} \frac{\partial \varphi'_i}{\partial \theta} \Omega_i^{-1} \frac{\partial \varphi_i}{\partial \theta}\right\}^{-1}\right),$$
 (3.42)

and

$$\sqrt{I}(\hat{\theta}_{ML} - \theta) \sim N(0, I_T^{-1}).$$
 (3.43)

3.2 Longitudinal Models for Count Data

As mentioned in Chapter I, a lot of authors discussed nonlinear transition models [Besag (1974); Wong (1986); Zeger and Qaqish (1988); Diggle et al. (2002)]. In practice, some count data may have linear relationship among the responses T_{ij} 's, for example, the dynamic population sizes in a district. Among the population of size $T_{i,j-1}$ in the prior year, some poople may die or more out in the (j-1)th year, and the rest are still living in this area. At the same time, there may be some newborns or immigration in this sees.

Another example is the prevolves count of an epidemic dimense with its default of the forwards one count of the perpedicion at a spin type. For T_{Q-1} denote the prevalence count in the (j-1)th year and T_Q denote the prevalence count in the dynamic dimension of the dynamic dimension dimension of the dynamic dimension dim

In the following subsections, we introduce two new models for dynamic count data which characterize the linear dependence among follow-up observations.

3.2.1 Non-stationary AR(1) model

Let T_{ij} denote the count response in district i in the jth year, i = 1, 2, ..., J and j = 1, 2, ..., J. McKemire (1988) and Sutradhar (2003) discussed a stationary AR(1) model for count data. The model is given by

$$T_{ij} = \gamma * T_{i,j-1} + \epsilon_{ij}, \qquad (3.44)$$

where v is the limits in this map of this processing and t_{22} denote the base interface density of the second with this is a spherical second se

In this thesis, we generalize model (3.44) to a new non-stationary AR(1) (NS-AR(1)) model which allows for time-waying covariates and unrestricted expectation of the additive error. The new model is given by

$$T_{ii} = \gamma * T_{i,i-1} + D_{ii}$$
 (3.45)

This model has a similar expression as (3.44) but different assumptions. It is assumed that $D_{ij} \sim Poisson(t_{ij})$, where t_{ij} may be a function of $t_{i,j-1}$ and some explanatory variables. However, given $T_{ij-1} = t_{ij-1}$, D_{ij} is independent of $\gamma * t_{i,j-1}$. It is obvious that, given the prior observations $T_{ij-1} = t_{i,j-1}$, the conditional expectation of T_{ij} can be formulated by

$$\eta_{ij}^{\epsilon} = E(T_{ij}|T_{i,j-1} = t_{i,j-1}) = \xi_{ij} + \gamma t_{i,j-1}.$$
 (3.46)

In practice, the new model (3.45) can be exploited to model different types of data sets with various background, then the assumptions about ξ_{ij} may vary accordingly. Two examples are presented in the following paragraphs.

- 1. For the dynamic population, T_{ij} denotes the population in are the ijk lysu in district i. 1. In model 54(k), the first term $\gamma + T_{ij-1}$, consists of people whose their in district i from the (j - 1)k by set to below. By preparadat new residents due to hirk and immigration. $T_{ij-1} = \gamma + T_{ij-1}$ is the number of people who due energyized advange that j = 0. (b) years, that is simulation, the number of people included in D_{ij} due to newborns may be related to the previous population and T_{ij-1} . Therefore, U_i can be a function of the previous population in the T_{ij-1} and the one observation U_i due to the population of the previous population in the T_{ij-1} . Therefore, U_i can be a function of the previous population in the T_{ij-1} .
- 2. For the data about prevalues of a dimuon, T_{ij} denotes the prevalues count of the dimuon ensues at the juwn in distarts i. The first part in model (Ad) ref₁₋₁ in composed parts are the (i - 1)th juwn can dura will melfruig the disease in the (j_1) juwn. D_{ij} includes new dimension count includes peopler who bermous indiced in the (j_1) juwn and particular includes M_{ij} denotes the other districts. $T_{ij-1} = \gamma + T_{ij-1}$ includes particular who did at to zero out of district in the (j_1) juwn before the energy. D_i on he reasonably assumed to be independent of the prior district radius (j_{i+1}) .

Alternatively, the term $\gamma T_{L_{L^+}}$ in the true complex can also be interpreted a test dynamic dependence of $T_{L_{L^+}}$ term (see the fact example, the case assumes that $\gamma + T_{L_{L^+}}$ can complexify durations the dynamic dependence, the D_0 can interpret of the model of the term (see the same test of the term (see the same test) methods are the term (see the same test) and the D_0^{-1} see mutually independent and they are independent of the previous descructions $T_{L_{L^+}} \lambda_{A}$ for far of μ_{A} (see the same test) and the D_0^{-1} see mutually independent and they are independent of the previous descructions $T_{L_{L^+}} \lambda_{A}$ For the NS-AR(1) model (3.45), it can be concluded from Section 2.4 in Chapter 3 that a Poisson variable $T_{i,j-1}$ leads to a Poisson-distributed $\gamma * T_{i,j-1}$, then it further leads to a Poisson variable $T_{i,j}$. There are three different cases of model (3.44).

- Case 1. The zero baseline observation $t_{c0} = 0$ leads to a unconditionally Poisson variable $T_{0,i}$ hence leads to a Poisson sequence $\{T_{ij} \sim Poisson(\eta_i)\}$). This can be used to model new cases of a discusse in different periods from the outbreak of the discuss in small areas. However, in a large district, it is difficult to go back to the outbreak of sum discuss, for example, castr.
- Case 2. Suppose that we have a non-zero baseline t_{db}. It leads to a Binomial variable γ * T_{db}. Therefore we get a non-Poisson variable T_{d1} and hence a sequence of non-Poisson variables {T_{d1}}.
- Case 3. If the baseline observations t_{40} 's are not available, we can assume that T_{41} follows a Poisson distribution with expectation ξ_{41} with a regression intercept β_{41} describing the baseline expectation. This assumption results in a sequence of Poisson variables $\{T_{41}\}$.

In the second case with non-zero baseline observations, the likelihood function of the data becomes very complicated, which will be shown in the following subsection.

3.2.2 Linear transition model

In some cases, T_{ij} and $T_{i,j-1}$ may not follow the NS-AR(1) model (3.45), but the conditional mean structure (3.46) describing the relationship between these two may still hold. For example, to model the incidence count of an epidemic disease in district i, the count of the new discase cases in the jth year T_{ij} may not follow the NS-AB(1) model (L-M). Even the dynamic population data may not exactly follow the NS-AR(1) model due to the mixture of migration with death and blath. Tabling this into consideration, the linear transmission (LT) model based on the conditional many entructure (L-M) becomes an appropriate alternative of the NS-AR(1) model (L-M).

The LT model is given by

$$T_{ij}|T_{i,j-1} = t_{i,j-1} \sim Poisson(\eta_{i,j}^{i} = \xi_{ij} + \gamma t_{i,j-1}).$$
 (3.47)

This model can be viewed as a special case of the linear formhar a work (2001) by Blandelly. Guildi and Waldward (2001) is which there is no subject-specific matom disc. However, our model has different interpretations from the URM. For the incidence count of a discuss, main model (11.07), t_0 denotes the the URM with the discuss at the prime model (12.07), t_0 denotes the discussion discussion of the discuss. The term $s_{1,0,1-1}$ of the discuss the discussion discussion of T_0 on $T_{0,1-1}$ and t_0 can be assumed in be a function of covariants and are resonanced factors which contribute to the expectation of the covariance of the discuss.

When the LT model is applied to data sets in the two examples in the previous subsection, the terms $\gamma_{1,j-1}$, ξ_0 and $(1 - \gamma)\epsilon_{i,j-1}$ have similar interpretations to those under the NSA.R(1) model presented in Section 1.2.1. However, the term $\gamma_{1,j-1}$ can also completely accommodate the dynamic dependence of T_{ij} on the prior observation t_{i-1} .

It is apparent that the conditional distribution of T_{ij} given the prior observation $t_{i,j-1}$ is a Poisson distribution. However, encept T_{i0} with baseline $t_{i0} = 0$, the marginal distribution of T_{ij} is not Poisson, for j = 1, 2, ..., J.

3.2.3 Moments of the NS-AR(1) and LT models

In this subsection, we provide the calculations of some measures of the response of our the NS-RM(1) could (14-8) and the Li mode (14-7). These measures are used in the construction of estimating expansion for the CQL and CQL² approaches. Since higher orders manufac, for randop, the third and function for monoting, are also provide the for exploring approaches, the provide the relevant constraints at the out of this association. One may matter that some measures under both NS-RR(1) model and LT model that the same requerests.

Under the NS-AR(1) model and LT model, we have the same expression of the expectation of T_{ij} which is given by

$$\eta_{ij} = \xi_{ij} + \gamma \eta_{i,j-1} \Rightarrow \eta_{ij} = \sum_{k=n+1}^{j} \xi_{ik} \gamma^{j-k} + \gamma^{j-n} \eta_n$$
, for $u < j$, (3.48)

where $\eta_{i0} = t_{i0}$. We also have the same expression of the expectation of the pairwise product $T_{i1}T_{i0}$ as follows:

$$\begin{split} E(T_{ij}T_{in}) &= & \eta_{in}\xi_{ij} + \gamma E(T_{in}T_{ij-1}) = \eta_{in} \int_{k=0}^{j-k-1} \gamma^k \xi_{ij-k} + \gamma^{j-\kappa} E(T_{in}^2) \\ &\Rightarrow & Cov(T_{in},T_{ij}) = \gamma^{j-\kappa}\sigma_{in}^2 \\ &\Rightarrow & Cov(T_{in},T_{ij}) = \min\{1, \gamma^{j-\kappa}\overline{\sigma}_{in}^2\}. \end{split}$$

The expectations of T_{ij}^2 under NS-AR(1) and LT models are, respectively, given

$$\begin{split} \text{NS-AR(1) model:} \qquad & E(T_{ij}^2) = \eta_{ij} + \eta_{ij}^2 + \gamma_i^2 [E(T_{i,j-1}^2) - \eta_{i,j-1} - \eta_{i,j-1}^2] \\ & \Rightarrow \ \ \sigma_{ij}^2 = \eta_{ij} + \gamma_i^2 (\sigma_{i,j-1}^2 - \eta_{i,j-1}), \\ \text{LT model:} \qquad & E(T_{ij}^2) = \eta_{ij} + \eta_{ij}^2 + \gamma_i^2 [E(T_{i,j-1}^2) - \eta_{i,j-1}^2] \\ & \Rightarrow \ \ \sigma_{ij}^2 = \eta_{ij} + \gamma_i^2 \sigma_{i,j-1}^2. \end{split}$$

Under both NS-AR(1) model and LT model, some third and forth order moments with the same expressions are given by

Under the NS-AR(1) model, the third and fourth order moments can also be

by

calculated as follows

$$\begin{split} & E(T_{i}^{*}) = u_{i} + 2d_{i}^{*} + d_{i}^{*} + T_{i}^{*}(T_{i}^{*})_{i}^{*} - u_{i}^{*} - d_{i}^{*})_{i}^{*} \\ & \qquad + 3\gamma^{2}(1 + d_{i}^{*} - \gamma)(E(T_{i}^{*})_{i}^{*}) - u_{i}^{*} - \tau^{2}d_{i}^{*}, d \\ & \qquad + \gamma^{2}(1 + d_{i}^{*} - \gamma)(E(T_{i}^{*})_{i}^{*}) - u_{i}^{*} - \tau^{2}d_{i}^{*}, d \\ & \qquad + \gamma^{2}(1 + d_{i}^{*} - \gamma)(E(T_{i}^{*})_{i}^{*}) - u_{i}^{*} - \tau^{2}d_{i}^{*}, d \\ & \qquad + \gamma^{2}(1 + d_{i}^{*} - \gamma)(E(T_{i}^{*})_{i}^{*}) - u_{i}^{*} - \tau^{2}d_{i}^{*}, d \\ & \qquad + \gamma^{2}(1 + d_{i}^{*} - \gamma)(E(T_{i}^{*})_{i}^{*}) - u_{i}^{*}d_{i}^{*}, d \\ & \qquad + (E(T_{i}^{*}T_{i}^{*})) - (E(T_{i}^{*})_{i}^{*}) - (E(T_{i}^{*})_{i}^{*}) \\ & \qquad = \sum_{i=1}^{N} A(T_{i}^{*})^{2i(i+1)} + \gamma^{2i(i+1)}E(T_{i}^{*})^{2i(i+1)} + \gamma^{2i(i$$

66

whereas under the LT model these moments can are

$$\begin{split} & R(T_{i}^{*}) = R_{i}^{*} d_{i}^{*} 2 d_{i}^{*} d_{i}^{*} (r_{i}^{*})^{*} \\ & = \theta_{i} + 2 d_{i}^{*} + 4 \gamma (1 + 4 (U(RT_{i}, ..., r_{i}^{*}), ..., r_{i}^{*}), ..., r_{i}^{*}) R(T_{i}^{*}, ..., r_{i}^{*}) \\ & R(T_{i}) = R_{i}^{*} d_{i}^{*} + 2 h_{i}^{*} (r_{i} + 6 h_{i}^{*})^{*} + (d_{i}^{*})^{*} \\ & = \eta_{i} + 2 d_{i}^{*} + 6 h_{i}^{*} + (d_{i}^{*})^{*} + (d_{i}^{*})^{*} \\ & = \eta_{i} + 2 d_{i}^{*} + 6 d_{i}^{*} + d_{i}^{*} + (T_{i}^{*}) + 2 e_{i}^{*} + 2 h_{i}^{*} + h_{i}^{*} (T_{i}^{*}), ..., - d_{i}^{*}, ..., 1 \\ & = \frac{1}{2 h_{i}^{*}} R_{i}^{*} R_{i}^{*} + 1 + 2 e_{i}^{*} (R_{i}^{*} T_{i}^{*}), ..., + (T_{i}^{*}), ..., + (T_{i}^{*}) \\ & = \frac{1}{2 h_{i}^{*}} R_{i}^{*} R_{i}^{*} - 1 + 2 e_{i}^{*} (R_{i}^{*} T_{i}^{*}), ..., + (T_{i}^{*}), ..., + (T_{i}^{*$$

3.2.4 Estimation of the model parameters

The parameters of interest in these two models are $\theta = (\beta^{i}, \gamma)^{i}$, where β represents the effects of covariates, and γ is the dynamic dependence parameter reflecting the correlation between the current outcome and the prior outcome.

3.2.4.1 Generalized quasi-likelihood method

The generalized quasi-likelihood method (GQL) [Sutradhar, (2003)] is to obtain the estimates by solving the estimating equations:

$$\sum_{i=1}^{l} \frac{\partial \eta_i'}{\partial \theta} \Sigma_i^{-1}(t_i - \eta_i) = 0, \quad (3.49)$$

where $\partial \eta_i / \partial \theta$ is the first derivative matrix of η_i with respect to θ and is of dimension $(p + 1) \times J$. p is the dimension of x_{iii} in ξ_{ii} . Among $\partial \eta_i / \partial \theta$,

$$\frac{\partial \eta_{kj}}{\partial \beta_k} = exp(x'_{ij}\beta)x_{ij(k)} + \gamma \frac{\partial \eta_{k,j-1}}{\partial \beta_k}$$
(3.50)

for k = 1, ..., p, j = 1, ..., J, and

$$\frac{\partial \eta_{kj}}{\partial \gamma} = \eta_{k,j-1} + \gamma \frac{\partial \eta_{k,j-1}}{\partial \gamma}$$
(3.51)

for j = 1, ..., J, $\Sigma_{i} = \Lambda_{i}W_{i}\Lambda_{i}$ is the variance-covariance matrix of T_{ii} where $\Lambda_{i} = diag(\sigma_{01}, ..., \sigma_{ij})$ and W_{ii} is the true correlation structure of T_{ii} if W_{ii} is replaced by a guarant "avorking" correlation structure W_{ii} the GQL approach becomes the GEE approach [Liang and Zegar (1986)]. Once we have the estimate θ_{QQL} , its corresponding covariance matrix can be estimated by

$$\hat{V}(\hat{\theta}_{OQL}) = \left(\sum_{s=1}^{I} \frac{\partial y_s^{s}}{\partial \theta} \Sigma_s^{-1} \frac{\partial y_b}{\partial \theta}\right)^{-1}|_{\theta = \hat{\theta}_{OQL}}.$$
 (3.52)

3.2.4.2 GQL2 approach

To obtain more efficient estimators of the model parameters, Sutradhar and Farrell (2007) developed the GQL2 approach which utilizes both the first and second order statistics of responses under the model (3.28) for dynamic binary data. In this section, we develop the GQL2 approach by exploiting the first and second order dynamic count responses. The GQL2 estimating equations are given by

$$\sum_{i=1}^{I} \frac{\partial \varphi'_i}{\partial \theta} \Omega_i^{-1}(h_i - \varphi_i) = 0 \quad (3.53)$$

[Sutradhar and Farrell, (2007)], where $H_i = (T_i^*, E_i^* \vee ink) T_i = (T_{i1}, ..., T_{iJ})^*$ and $R_i = (T_{i1}^*, ..., T_{iJ}^*, T_{iJ}^* T_{iJ}^*, ..., T_{iJ}^*, T_{iJ}^*)^*$, h_i is the observation of H_i and $\eta_i = E(H_i)$ is the expectation of H_i . The variance-covariance matrix of H_i is given by

$$\Omega_i = \begin{pmatrix} Cov(T_i) & Cov(T_i, R_i) \\ Cov(R_i, T_i) & Cov(R_i) \end{pmatrix}_{J(J+3)/2 \times J(J+3)/2}$$

Some quantities required in this approach such as φ_i , $Cov(T_i)$, $Cov(R_i, T_i)$ and $Cov(R_i)$ can be easily calculated based on the moments calculated in Section 3.2.3. Some result derivative required in (2.33) are size below.

Under the NS-AR(1) model.

$$\begin{array}{rcl} \frac{\partial \mathcal{E}(T_{ij}^2)}{\partial \overline{\eta}} &=& \frac{\partial \eta_{ij}}{\partial 1}(1+2\eta_{ij})+\gamma^2[\frac{\partial \mathcal{E}(T_{ij-1}^2)}{\partial \overline{\eta}}-\frac{\partial \eta_{ij-1}}{\partial 1}(1+2\eta_{i,j-1})],\\ \\ \frac{\partial \mathcal{E}(T_{ij}^2)}{\partial \overline{\gamma}} &=& \frac{\partial \eta_{ij}}{\partial 1}(1+2\eta_{ij})+\gamma^2[\frac{\partial \mathcal{E}(T_{ij-1}^2)}{\partial \gamma}-\frac{\partial \eta_{ij-1}}{\partial \gamma}(1+2\eta_{i,j-1})],\\ \\ &+\gamma[\mathcal{E}(T_{ij-1}^2)-\eta_{ij-1}-\eta_{ij-1}], \end{array}$$

Under the LT model,

$$\begin{array}{ll} \displaystyle \frac{\partial \mathcal{E}(T_{ij}^2)}{\partial \beta} &=& \displaystyle \frac{\partial \eta_{ij}}{\partial \beta}(1+2\eta_{ij})+\gamma^2 (\frac{\partial \mathcal{E}(T_{i,j-1}^2)}{\partial \beta}-2\eta_{i,j-1}\frac{\partial \eta_{i,j-1}}{\partial \beta}), \\ \displaystyle \frac{\partial \mathcal{E}(T_{ij}^2)}{\partial \gamma} &=& \displaystyle \frac{\partial \eta_{ij}}{\partial \gamma}(1+2\eta_{ij})+\gamma^2 (\frac{\partial \mathcal{E}(T_{i,j-1}^2)}{\partial \gamma}-2\eta_{i,j-1}\frac{\partial \eta_{i,j-1}}{\partial \gamma})+2\gamma [\mathcal{E}(T_{i,j-1}^2)-\eta_{i,j-1}^2)], \end{array}$$

Under both of the NS-AR(1) and LT models.

$$\begin{split} \frac{\partial \mathcal{E}(T_{n,T_{2}}^{-1})}{\partial \sigma} &= \frac{\partial \tilde{m}_{0}}{\partial \sigma} + \eta_{n} \mathcal{L}_{0} \mathcal{L}_{0} + \frac{\partial \mathcal{E}(T_{n}^{-1}\mathcal{L}_{2}, \omega)}{\partial \sigma} \\ &= \frac{\partial \tilde{m}_{0}}{\partial \sigma} \sum_{j \to +1} \mathcal{L}_{j} \mathcal{L}_$$

Once the estimate $\hat{\theta}_{OQ62}$ is obtained, the corresponding covariance matrix can be stimated by

$$\hat{V}(\hat{\theta}_{GQE2}) = \left(\sum_{i=1}^{n} \frac{\partial \varphi_{i}^{i}}{\partial \theta} \Omega_{i}^{-1} \frac{\partial \varphi_{i}}{\partial \theta}\right)^{-1} |_{\theta = \hat{\theta}_{GQE2}},$$
 (3.54)

As mentioned in the Section 3.1.2, it is shown by Standaue and Farrel (2021) that the GQL method possible as efficient estimators as the HL approaches sentor that output methods are difficult estimations in the HL approaches meththod output assumed OQL. Similarly, the OQL method in the first order transition model (1.4.7) is also expected to produce abusts as efficient as the MLFs for model memory, which can be demonstrated in our numeric status. Therefore, we do not constart simulations under the GQL method in the first Table of the therm methods with the GQL method under the LT model. We expect that the therm methods with cationative with are simular efficiency.

3.2.4.3 Maximized likelihood method

In this subsection, we develop parameter estimation based on the likelihood approach. Under NS-AR(1) model, the conditional probability that $T_{ij} = t_{ij}$ given the prior observation $t_{i,j-1}$ is

$$\frac{P(l_0^{-1} - e_l(T_{l-1} - t_{l-1}))}{\sum_{k_l=0}^{m(l_l-1_{l-1})} P(T_0 = t_0|T_{l-1} - t_{l-1}, K_0 = k_0)P(T_{l-1} - t_{l-1})} \\ = \sum_{k_l=0}^{m(l_l-1_{l-1})} P(N_0 = t_0 - k_0)\frac{t_{l-1}}{k_0(t_{l-1} - t_0)^{\gamma-1}}^{\lambda_0} (1 - \gamma^{t_{l-1}-k_0}) \\ = \sum_{k_l=0}^{m(l_l-1_{l-1})} \frac{t_{l-1}^{-1}e_l}{(t_l-t_0)^{k_l}} \frac{t_{l-1}}{k_0(t_{l-1} - t_0)^{\gamma-1}}^{\lambda_0} (1 - \gamma^{t_{l-1}-k_0})$$
(3.55)

It is very difficult to calculate the joint likelihood of T_{ij} with the observation t_{ij} because of the complicated conditional probability (2.55). Therefore, the ML astimators of model parameters are difficult to obtained under the NS-AR(1) model. In the momeric studies, we only conduct simulations to check the performance of GQL and to GQL anotheds for the NS-AR(1) model.

As far as the LT model is concerned, the joint likelihood given observations T=t is written as

$$L(\theta|T = t) = \prod_{i=1}^{t} f(t_i)$$

 $= \prod_{i=1}^{t} \prod_{j=1}^{t} f(t_i) \{t_{ij} = t_i\}$
 $= \prod_{i=1}^{t} \prod_{j=1}^{t} \frac{f_i(t_j)^{t_{ij}} exp(-t_i^{t_j})}{t_{ij}!},$ (3.56)

where $T = \{T_{ij}, i = 1, 2, ..., I, j = 1, 2, ..., J\}$, its observations are $t = \{t_{ij}, i = 1, 2, ..., I, j = 1, 2, ..., J\}$ with baseline observations $t_{i0}, i = 1, 2, ..., I$, and $\eta_{i1} = exp(x_{i1}\beta) + \gamma t_{i0}$. The log-likelihood can be expressed as

$$\ell \theta$$
 = $\sum_{i=1}^{I} \sum_{j=1}^{J} t_{ij} log[exp(x_{ij}\beta) + \gamma t_{i,j-1}] - \sum_{i=1}^{I} \sum_{j=1}^{J} [exp(x_{ij}\beta) + \gamma t_{i,j-1}].$ (3.57)

The ML estimates $\hat{\theta}_{ML}$ are obtained by solving the equations:

$$\sum_{i=1}^{I} \sum_{j=1}^{J} (t_{ij}/\eta_{ij}^{c} - 1) exp(x_{ij}^{c}\beta)x_{ij} = 0, \quad (3.58)$$

$$\sum_{i=1}^{J} \sum_{j=1}^{J} (t_{ij}/\eta_{ij}^{c} - 1)t_{i,j-1} = 0. \quad (3.59)$$

To estimate the covariance matrix of $\hat{\theta}_{ML}$, one need to calculate the Fisher information matrix which can be given by

$$l = -E\left\{\frac{|\phi'(\phi)|}{|\phi'(\phi)|}\right\} = \left\{\frac{|\phi'(\phi)|}{|\phi'(\phi)|} - E\left\{\frac{|\phi'(\phi)|}{|\phi'(\phi)|}\right\}\right\}$$

$$= E\left[\sum_{l=1}^{d} \left(\frac{|\phi'(\phi)|}{|\phi'(\phi)|} - \frac{|\phi'(\phi)|}{|\phi'(\phi)|}\right)\right]$$

$$= E\left[\sum_{l=1}^{d} \left(\frac{|\phi'(\phi)|}{|\phi'(\phi)|} - \frac{|\phi'(\phi)|}{|\phi'(\phi)|}\right)\right]$$

$$= E\left[\sum_{l=1}^{d} \left(\frac{|\phi'(\phi)|}{|\phi'(\phi)|} - \frac{|\phi'(\phi)|}{|\phi'(\phi)|}\right)\right]. (3.6)$$

Then, covariance matrix of $\hat{\theta}_{ML}$ can be estimated by

$$\hat{V}(\hat{\theta}_{ML}) = I \left[\sum_{i=1}^{I} \sum_{j=1}^{J} \left(\begin{array}{c} \frac{t_{ij}^2}{q_{ij}^2} x_{ij} x_{ij}^2}{\frac{t_{ij}}{q_{ij}^2} t_{ij-1} x_{ij}} \\ \frac{t_{ij}}{q_{ij}^2} t_{ij-1} x_{ij}^2 & \frac{t_{ij+1}^2}{q_{ij}^2} \end{array} \right) \right]_{\theta=\hat{\theta}_{ML}}^{-1}$$
(3.61)

It can be some from (2.186) and (2.100), the ML estimating equations under the DT model (3.147) analy involve the first and second order responses, which is very similar to (3.15.3.33) and/or dynamic binary model (3.28). This messes that the GO(23 spproaches employing the first and the second responses tends to yield almost identical estimates to the ML estimates. Therefore, the GO(23 under the LT model may be the optimal GO(20) approach.

3.2.5 Simulation studies

In this metric, we conduct simulations to examine some intermedia lenses in deticular distremut one different model articing and dispose of countries. For they we will dock the performance of the proposed estimation approaches. Usable to bevolution of the effect of the strength of the strength of the distret of the strength of the strength of the strength onder the MS-ARI(1) model to lenses of the complexity of the likelihood function. Similarly, and the distret of H model (LHC), 1000 minutations on the stude be OQL and ML methods. The remains that we do not exolute a simulation for the OQL and prodution that the OQL approach. In the theory of the likelihood function, the the OQL approach, the the OQL OQL estimations are expected to be almost same as the OQL and ML estimator.

Next, it is noticed that excert equivalent of the baseline observation is also of importance producing sublick emittants of model parameters in transition models infining the IS-8.24(1) model (\pm (\pm)) and the LT model (\pm (\pm)). Inprecise, there may be different datases of the hashing absorbing in the second second second second part for the theoretic parameters of the second second second second second second in an intractic procession. 300 simulations about misperification of baseline downed sources combard 10 Section 22.25.1

It can be seen that the NS-AR(1) model and the LT model are very similar in many aspects including model structure, conditional and unconditional expectation structures as well as some higher moments shown in Section 3.2.3. These similarities may lead to challenges in distinguishing them. Therefore, it is meaningful to check the possible misspecification of the two models. For this purpose, 500 simulations are carried out in Section 3.2.5.4.

3.2.5.1 Designs

In the simulation studies in Section 3.2.5.2 and Section 3.2.5.3, we let $\xi = exp(\Lambda_i + \Lambda_F v_{QII} + \Lambda_F v_{QII})$ for both the NS-AR(1) model (3.44), and 1.7 model (3.47). However, we set up different designs of the two constants x_{QCII} and x_{QCII} under the two models as below.

Design I: In the NSAR(1) model, we have chosen the samples size I = 00, and number of repeated observations J = 4. Three values are chosen for $\gamma : 0.2, 0.5$, and 0.8. The baseline observations d_{22} are sampled from a Fluore intrimitivity, that is $t_{dis} \sim Pietsone(10)$. Among the two time dependent covariates follows, the first case discuss a categorical variable, for example the economic level, six quality level.

$$x_{ij(0)} = \begin{cases} 1, & j=1,2 \text{ and } i=1,...,1/2; \\ 0, & j=3,4 \text{ and } i=1,...,1/2; \\ 0, & j=1,2 \text{ and } i=n/2+1,...,1 \\ 1, & j=3,4 \text{ and } i=n/2+1,...,1 \end{cases}$$

and the second covariate is a subject-constant but time-varying variable which represents a consituous variable.

$$x_{intri} \sim N(j/10 - 0.1, 1), j = 1, 2, 3, 4$$
 for all $i = 1, 2, ..., 60$.

Design II: In the LT model, we choose the same sample size I = 60, but different

number of repeated measurements J = 6. For different values are assigned to γ : 0.05, 0.2, 0.5, and 0.9. The baseline observations t_0 are also sampled from the Poisson distribution *Poisson*(50). Similar to two covariates in the NS-AR(1) model. the two time-varying councilies are given by

$$\begin{array}{cccc} -1, & j{=}1,2 \mbox{ and } i{=}1,\ldots,1/3; \\ 0, & j{=}3,4 \mbox{ and } i{=}1,\ldots,1/3; \\ 1, & j{=}5,6 \mbox{ and } i{=}1,\ldots,1/3; \\ 0, & j{=}1,2,3 \mbox{ and } i{=}1/3+1,\ldots,2/3; \\ 1, & j{=}1,2,3 \mbox{ and } i{=}1/3+1,\ldots,2/3; \\ 1, & j{=}1,2,3 \mbox{ and } i{=}2/3+1,\ldots,1; \\ 0, & j{=}4,5,6 \mbox{ and } i{=}2/3+1,\ldots,1; \end{array}$$

and

$$r_{ij(2)} \sim \begin{cases} N((j-1)/10, 1), & \text{for } j = 1, ..., 4, \text{and} i = 1, ..., I; \\ N(3/10, 1), & j=5; \text{ and } i=1, ..., I, \\ N(2/10, 1), & j=6; \text{ and } i=1, ..., I. \end{cases}$$

In Section 3.2.5.4, we use Design II for both the NS-AR(1) model and the LT model to compare the misspecification between them.

3.2.5.2 Estimation of model parameters

In this section, 1000 simulation are conducted to check the performance of the proposed approaches in stituaring interested parameters in the NS-AR(1) model (3.42) and LT model (3.47). The results under the NS-AR(1) model and LT model are given in Table 3.2 and Table 3.3, respectively.

In each case of γ , we applied the GQL and GQL2 approaches to estimate the model parameters $\theta = (\beta', \gamma)'$, where $\beta = (\beta_0, \beta_1, \beta_2)'$ are the regression coefficients, γ is the dynamic dependence parameter. From Table 3.2, it can be seen that both the GQL and GQL2 approaches yield approximately unbiased estimates of model parameters. For example, when $\gamma = 0.2$, we have the GQL estimates of $\hat{\beta}_{OOL} =$ (1.0009, -1.0010, 0.9988)', and the GQL estimate of γ is 0.1999. In the same case, $\hat{\beta}_{OOL3} = (1.0093, -1.0000, 0.9985)'$ and $\hat{\gamma}_{OOL3} = 0.1997$. They are very close to the true values $\beta = (1, -1, 1)$ and $\gamma = 0.2$. The estimated standard errors (ESE) derived from GQL estimator (3.52) and GQL2 estimator (3.54) are almost identical to their corresponding simulated standard errors (SSE). Therefore, the coverage probabilities Another point is that the GQL2 approach tends to have slightly higher efficiency than the GQL approach when the ESE's and SSE's are concerned. For example, in the case that $\gamma = 0.8$, the (SSE, ESE) of $\hat{\beta}_i$ are (0.1899, 0.1874) under the GQL approach, and (0.1891, 0.1870) under the GQL2 approach. However, in most cases the GQL performs almost as well as the GQL2 as far as the SM, SSE, ESE and CPr

Under the LT model, for elificate values of the dynamic dependence permeters ∂D_{00} and ML approaches are used to emission model parameters $\sigma = (f, r)^{-1}$ with $\beta = (\dot{\alpha}_{i}, \beta_{i}, \beta_{i})$. The simulation results are given in Table 4.3. It is shown that the GQL and ML approaches juid approximately ability of the dynamic of P. for rample, $m_{i} = 0.5$, we have the GQL estimation of $\dot{\alpha}_{i}_{i} = (1.006, -1.0007, 0.9007)$ with SNFs (1044), 0.0403, 0.0413 and 2.0257, 0.1307, 0.0005, 0.0003, onld to GQL minutes $\sigma + 1$ is GPH = 0.527 (0.1257, 0.0005, 0.0003), and the GQL minutes $\sigma + 1$ is GPH = 0.527 (0.1257, 0.0055, 0.0003), and the GQL SPL = 0.0003 (0.0013), SPL = 0.0033 (0.0013),

| | $\gamma = 0.2$ | | $\gamma =$ | 0.5 | $\gamma = 0.8$ | | |
|-----------------------|----------------|---------|------------|---------|----------------|---------|--|
| Quantity | GQL | GQL2 | GQL | GQL2 | GQL | GQL2 | |
| $SM(\beta_0)$ | 1.0009 | 1.0093 | 0.9949 | 0.9951 | 0.9976 | 0.9983 | |
| SSE | 0.0701 | 0.0692 | 0.1032 | 0.1031 | 0.0964 | 0.0961 | |
| ESE | 0.0698 | 0.0691 | 0.1019 | 0.1018 | 0.0966 | 0.0965 | |
| CPr | 0.942 | 0.946 | 0.943 | 0.941 | 0.940 | 0.944 | |
| $SM(\beta_1)$ | -1.0010 | -1.0000 | -1.0163 | -1.0160 | -1.0128 | -1.0127 | |
| SSE | 0.0572 | 0.0564 | 0.1825 | 0.1823 | 0.1899 | 0.1891 | |
| ESE | 0.0545 | 0.0541 | 0.1796 | 0.1793 | 0.1874 | 0.1870 | |
| CPr | 0.941 | 0.944 | 0.952 | 0.952 | 0.950 | 0.952 | |
| $SM(\beta_2)$ | 0.9988 | 0.9985 | 0.9986 | 0.9984 | 0.9991 | 0.9987 | |
| SSE | 0.0347 | 0.0346 | 0.0567 | 0.0566 | 0.0409 | 0.0408 | |
| ESE | 0.0346 | 0.0345 | 0.0563 | 0.0562 | 0.0418 | 0.0417 | |
| CPr | 0.944 | 0.951 | 0.951 | 0.949 | 0.951 | 0.953 | |
| $\mathrm{SM}(\gamma)$ | 0.1999 | 0.1997 | 0.4999 | 0.4999 | 0.7999 | 0.7999 | |
| SSE | 0.0080 | 0.0079 | 0.0083 | 0.0083 | 0.0053 | 0.0053 | |
| ESE | 0.0079 | 0.0079 | 0.0083 | 0.0083 | 0.0054 | 0.0054 | |
| CPr | 0.949 | 0.949 | 0.950 | 0.954 | 0.955 | 0.956 | |

Table 3.2: Simulation results for the NS-AR(1) model with the true values of param-

eters $\beta = (\underline{1, -1, 1})$

estimates are $\hat{\beta}_{ML} = (1.0014, -1.0002, 0.9989)'$ with SSE's (0.1020, 0.0819, 0.0425)and ESE (0.1045, 0.0830, 0.0425) and $\hat{\gamma}_{ML} = 0.0493$ with a SSE 0.0098 and an ESE 0.0098. These estimates of θ are very close to the true values $\theta = (1, -1, 1, 0.5)$, and these ISE's are also very close to the corresponding SSE's.

It also on he seem through Table 3.1 that ML approach much to have slightly higher efficiency that the GQL approach, thus is, the SSE's (SSE's) under the ML approach are a little smaller than the corresponding SSE's (SSE's) under the GQL approach. However, in most cases, the GQL approach, Beckers estimates that are very device to the ML estimations as far at the SSL SSL SSE and CP was encoursed. This implies that the first order responses $\{\tau_{ki}\}$ include almost all information about the model parameters θ . This is also vely we did not conduct the simulation about the OQL2 approach on the LT model.

In summary, the proposed approaches can effectively estimate the model parameters in the NS-AR(1) model and the LT model. The GQL approach yields highly efficient estimates of parameters. This implies meaningful application of this method because of its simplicity compared with the GQL2 method under the NS-AR(1) model and the ML methods.

3.2.5.3 Misspecified baseline observations

To check the influence of possible misspecification of baseline observations on parameter astimation, in this subsection, we conduct 500 simulations to check the performance of the proposed approaches for NS-AR(1) model (3.45) and LT model (3.47) taking the mis-specified baseline observations into consideration. In simulation, we choose $\beta = a(\beta_{1,0}\beta_{2,0}) = (1, -1, 1)/(and \gamma = 0.46)$. The true baseline observations

| | $\gamma =$ | 0.05 | $\gamma = 0.2$ | | $\gamma = 0.5$ | | $\gamma = 0.9$ | |
|---------------|------------|---------|----------------|---------|----------------|---------|----------------|---------|
| Quantity | GQL | ML | GQL | ML | GQL | ML | GQL | ML |
| $SM(\beta_0)$ | 1.0025 | 1.0029 | 1.0007 | 1.0024 | 1.0004 | 1.0014 | 0.9909 | 0.9934 |
| SSE | 0.0599 | 0.0600 | 0.0701 | 0.0692 | 0.1054 | 0.1020 | 0.1956 | 0.1928 |
| ESE | 0.0597 | 0.0595 | 0.0098 | 0.0691 | 0.1070 | 0.1045 | 0.1876 | 0.1847 |
| CPr | 0.948 | 0.950 | 0.942 | 0.946 | 0.950 | 0.949 | 0.938 | 0.935 |
| $SM(\beta_1)$ | -0.9990 | -0.9988 | -1.0010 | -1.0000 | -1.0007 | -1.0002 | -1.0047 | -1.0032 |
| SSE | 0.0539 | 0.0539 | 0.0572 | 0.0564 | 0.0843 | 0.0819 | 0.1585 | 0.1564 |
| ESE | 0.0522 | 0.0521 | 0.0545 | 0.0541 | 0.0856 | 0.0835 | 0.1507 | 0.1484 |
| CPr | 0.937 | 0.937 | 0.941 | 0.944 | 0.962 | 0.955 | 0.937 | 0.934 |
| $SM(\beta_2)$ | 0.9972 | 0.9971 | 0.9988 | 0.9985 | 0.9989 | 0.9989 | 1.0013 | 1.0008 |
| SSE | 0.0278 | 0.0278 | 0.0347 | 0.0346 | 0.0431 | 0.0425 | 0.0653 | 0.0647 |
| ESE | 0.0277 | 0.0277 | 0.0346 | 0.0345 | 0.0430 | 0.0425 | 0.0643 | 0.0639 |
| CPr | 0.954 | 0.954 | 0.944 | 0.951 | 0.950 | 0.949 | 0.939 | 0.946 |
| $SM(\gamma)$ | 0.0499 | 0.0498 | 0.1999 | 0.1997 | 0.4994 | 0.4953 | 0.8994 | 0.8994 |
| SSE | 0.0054 | 0.0053 | 0.0050 | 0.0079 | 0.0100 | 0.0099 | 0.0083 | 0.0082 |
| ESE | 0.0053 | 0.0052 | 0.0079 | 0.0079 | 0.0098 | 0.0098 | 0.0083 | 0.0082 |
| CPr | 0.944 | 0.939 | 0.949 | 0.949 | 0.942 | 0.950 | 0.951 | 0.946 |

Table 3.3: Simulation results for the LT model with the true values of parameters

 $\beta = (1, -1, 1)$

 t_0 are generated from *Poisson*(50) under both the NS-AR(1) model and the LT model, whereas, in conducting statistical inference for the two models, we assume that all the baseline observations are mis-specified to be 50. The simulation results in presence of the mis-specified baseline observations under the NS-AR(1) model and LT model are given Table 3.4.

As discussed in Section 3.2.1, the baseline observations have much influence on the NS-AR(1) model. Actually, missmerified baseline observations are expected to have significant influence on the statistical inference based on both the NS-AR(1) model and the LT model. This is because the missnerified baseline observations load to around expectations no. hence lead to incorrect no for i = 2,....J. Further, all moments of the remonse are incorrect due to the missnerification of baseline observations. Therefore, statistical inferences based on the estimating countion-based approaches and the ML approach which highly depends on the accuracy of data are not reliable any more. This can be demonstrated from Table 3.4. For example, under the NS-AR(1) model, the GQL estimate of $\beta_0 = 1$ is 0.9000, and the corresponding GOL2 estimate is 0.9411. The CPr's under the two approaches are 0.900 and 0.912. which are significantly lower than the nominal level 0.95. Similarly, under the LT model, $\hat{\beta}_{waves} = 0.9471$ and $\hat{\beta}_{waves} = 0.9524$ both of which have significant biases from the true value 1. The CPr's of β_0 under the GQL approach is 0.936 which is much smaller than 0.95. Further note that the biases of A. A. and 4 under both are enables are significant relative to the biases of these estimates employing correct baseline observations, and the CPr's in the case of mis-specified observations are simificantly biased from 0.95.

Based on the simulation results in Table 3.4. we conclude that the baseline ob-

| Table 3.4: Mis- | specifying the | baseline o | deservation (| $t_{s0} = 50 \ w$ | hen $t_{40} \sim .$ | Pois(50) with |
|-----------------|----------------|------------|---------------|-------------------|---------------------|---------------|
| | | | | | | |

| | | |
|------|--|--|
| | | |
| | | |

| | | Correct | | | Misspecified | | | | |
|--------|-----|---------|-----------|-----------|--------------|---------|-----------|-----------|--------|
| | | ß | β_1 | β_2 | γ | β | β_1 | β_2 | γ |
| Method | | | | | NS-AR(| 1) mode | 4 | | |
| GQL | SM | 0.9954 | -1.0129 | 1.0000 | 0.6494 | 0.9300 | -1.1531 | 1.0279 | 0.6573 |
| | SSE | 0.0956 | 0.1833 | 0.0422 | 0.0069 | 0.0966 | 0.2055 | 0.0410 | 0.0069 |
| | ESE | 0.0971 | 0.1876 | 0.0433 | 0.0069 | 0.0979 | 0.2185 | 0.0428 | 0.0009 |
| | CPr | 0.940 | 0.960 | 0.946 | 0.950 | 0.900 | 0.974 | 0.900 | 0.814 |
| GQL2 | SM | 0.9959 | -1.0132 | 0.9997 | 0.6495 | 0.9411 | -1.0829 | 1.0243 | 0.6557 |
| | SSE | 0.0967 | 0.1830 | 0.0422 | 0.0069 | 0.0977 | 0.1956 | 0.0419 | 0.0069 |
| | ESE | 0.0969 | 0.1866 | 0.0433 | 0.0059 | 0.0974 | 0.2048 | 0.0427 | 0.0070 |
| | CPr | 0.942 | 0.962 | 0.946 | 0.946 | 0.912 | 0.978 | 0.910 | 0.862 |
| | | | | | LT : | nodel | | | |
| GQL | SM | 0.9978 | -1.0024 | 0.9974 | 0.6494 | 0.9471 | -1.0293 | 1.0168 | 0.6536 |
| | SSE | 0.1352 | 0.1070 | 0.0525 | 0.0094 | 0.1391 | 0.1094 | 0.0539 | 0.0094 |
| | ESE | 0.1369 | 0.1077 | 0.0516 | 0.0094 | 0.1376 | 0.1093 | 0.0508 | 0.0095 |
| | CPr | 0.956 | 0.958 | 0.960 | 0.954 | 0.936 | 0.948 | 0.906 | 0.938 |
| ML | SM | 1.0012 | -1.0000 | 0.9968 | 0.6493 | 0.9524 | -1.0256 | 1.0157 | 0.6535 |
| | SSE | 0.1303 | 0.1040 | 0.0516 | 0.0094 | 0.1344 | 0.1052 | 0.0529 | 0.0094 |
| | ESE | 0.1340 | 0.1054 | 0.0510 | 0.0094 | 0.1352 | 0.1072 | 0.0502 | 0.0094 |
| | CPr | 0.962 | 0.964 | 0.958 | 0.952 | 0.952 | 0.960 | 0.916 | 0.934 |

servations do influence the statistical inferences about models. Therefore, one should be careful to choose baseline values in practice.

3.2.5.4 Misspecification of models

In this section, 500 simulation are conducted to check the influence of the misspecification of models on the estimates of model effects. We consider two cases of the missnerification of models.

- Case 1. Under the true NS-AR(1) model (3.45), we mis-specify the model as the LT model (3.47), of which the simulation results are given in the Table 3.5.
- Case 2. Under the true LT model (3.47), we mis-specify the model as the NS-AR(1) model (3.45), of which the results are given in Table 3.6.

It can be seen from the two tables that, the statiants under all sympaches in both Car I and Case III are reliably improvements of the theory of the SF Case I and Scale III are reliably the statistical sympactic sympactic sympactic and the ML emission ($M_{\rm He}, S_{\rm HM} = 0.0086, \pm 0.0086, \pm 0.0086, 1.0076, \pm 0.0086, 0.0076, 1.0008, 0.00076, 1.0008, 0.00076, 1.0018, 0.00186, 0.00076, 0.00186,$ As for set the QQ2 separatic moder the minoperfield NS-Ab(1)) model Table 36 is conversels. He QQ2 arisations may have adding space biasance that the QQ4 extinuits have, separably for large waves of γ_{1} for intranse, when $\gamma = 0.3$, the QQ4 and $\gamma_{2} = 0.006$. This may be due to the different expectations of QQ3 estimatlagging z = 0.006. This may be due to the different expectations of Q24 as simulations of z_{1} is 0.013, which has be bus the test the test magnetized in the QQ24 as similar question (Q34) are the images of the Q24 as similar questions of z_{2}^{2} moder the the QQ24 as similar question (Q34) are the test questions of z_{2}^{2} moder the solutions of the Q242 as similar question (Q34) are similar to the values of quantum T. Tak is beaux that the fits order responses (Te₄) and the pairwise products of the mapping (T₂T_n), which have the same expectations mader the two models, have mader included above infinite in timteration above the mathematical products on quantum test of the spectra of the simulation of the quantum test of the mathematical products on quantum test of the simulation of the simulation of the test of the simulation of the simulation of the simulation of the test of the simulation of the test of the simulation of the simulation of the test of the simulation of the simulation of the simulation of the simulation of the test of the simulation of the simulation of the simulation of the simulation of the test of the simulation of the sis

However, the arcimited standard stress and coverage probabilities of 1953 (Teb to the neutrinoisy concentration is the stress that cover particular to a stress that the stress in the stress that the stress is the stress that the stress str

are, respectively, 0.999 and 1.000, which are all much greater than 0.95.

As for a Gase II is ensemed, the GQL estimators and the GQL estimators of anta-hole errors obtained from [3,32] and [3,56], respectively, total beam nonversity underweinhand families errors as γ information, which hash to how CPV than the nonmial local GS. For intensor, when $\gamma = 0.9$ is Table 3.6, the EEG Gauge is 0.600, and is in such smaller than the webs of the corresponding SIE 0.1966. Similarly the value of ESE of $\beta_{\rm AUVAR}$ is 0.4000, which the corresponding SIE is a much larger value 1.1564. The GV/s for $\beta_{\rm AUVAR}$ is 0.4000, which the corresponding SIE of this discretion of the Gauge and the state of the corresponding SIE is a much sequence of the larger value 1.1564. The GV/s for $\beta_{\rm AUVAR}$ is for the two spectras the low corresponding the phenomenon may be breases much error information is used in containing attandard errors and its constructing confidences intrevals used and approaches in the first order responses and pairwise potentia of the first order improves a phase in pairwise potentia of the first order responses and pairwise potentia of the first order improves for the phase intervals of the order of the order of channel mode of the order of channel desception.

In summary, the misuperification of models described in Case I and Case II do not affect the unbiasedness of estimates of model parameters. However, the mispaceification may lead to severely biased estimation of standard errors of $\hat{\theta}_i$ then lead to poor confidence intervals.

| | $\gamma =$ | 0.05 | $\gamma = 0.2$ | | $\gamma = 0.5$ | | $\gamma = 0.9$ | |
|---------------|------------|---------|----------------|---------|----------------|---------|----------------|---------|
| Quantity | GQL | ML | GQL | ML | GQL | ML | GQL | ML |
| $SM(\beta_0)$ | 0.9970 | 0.9972 | 0.9977 | 0.9987 | 0.9984 | 0.9996 | 0.9953 | 0.9958 |
| SSE | 0.0590 | 0.0588 | 0.0702 | 0.0699 | 0.0900 | 0.0894 | 0.1001 | 0.0999 |
| ESE | 0.0599 | 0.0597 | 0.0720 | 0.0712 | 0.1065 | 0.1050 | 0.1925 | 0.1920 |
| CPr | 0.953 | 0.953 | 0.958 | 0.956 | 0.980 | 0.976 | 0.999 | 1.000 |
| $SM(\beta_1)$ | -0.9998 | -0.9997 | -1.0005 | -0.9999 | -1.0002 | -0.9995 | -1.0036 | -1.0034 |
| SSE | 0.0523 | 0.0521 | 0.0575 | 0.0571 | 0.0723 | 0.0720 | 0.0853 | 0.0852 |
| ESE | 0.0523 | 0.0522 | 0.0599 | 0.0594 | 0.0851 | 0.0839 | 0.1510 | 0.1506 |
| CPr | 0.947 | 0.949 | 0.951 | 0.956 | 0.975 | 0.978 | 0.999 | 0.999 |
| $SM(\beta_2)$ | 0.9999 | 0.9999 | 0.9988 | 0.9986 | 0.9996 | 0.9993 | 0.9999 | 0.9999 |
| SSE | 0.0270 | 0.0270 | 0.0318 | 0.0317 | 0.0372 | 0.0370 | 0.0362 | 0.0352 |
| ESE | 0.0277 | 0.0277 | 0.0320 | 0.0318 | 0.0430 | 0.0428 | 0.0665 | 0.0665 |
| CPr | 0.958 | 0.958 | 0.955 | 0.953 | 0.972 | 0.976 | 0.998 | 0.998 |
| $SM(\gamma)$ | 0.0500 | 0.0500 | 0.2003 | 0.2002 | 0.4999 | 0.4998 | 0.9000 | 0.9000 |
| SSE | 0.0053 | 0.0053 | 0.0077 | 0.0076 | 0.0070 | 0.0070 | 0.0032 | 0.0032 |
| ESE | 0.0053 | 0.0053 | 0.0082 | 0.0082 | 0.0097 | 0.0097 | 0.0082 | 0.0082 |
| CPr | 0.943 | 0.939 | 0.960 | 0.965 | 0.994 | 0.994 | 1.000 | 1.000 |

Table 3.5: Misspecified LT model under true NS-AR(1) model, where $\beta = (1, -1, 1)$.

85

| | $\gamma =$ | 0.05 | $\gamma = 0.2$ | | $\gamma = 0.5$ | | $\gamma = 0.9$ | |
|---------------|------------|---------|----------------|---------|----------------|---------|----------------|---------|
| Quantity | GQL | GQL2 | GQL | GQL2 | GQL | GQL2 | GQL | GQL2 |
| $SM(\beta_0)$ | 0.9980 | 0.9986 | 0.9947 | 0.9950 | 1.0009 | 1.0066 | 0.9913 | 0.9845 |
| SSE | 0.0693 | 0.0690 | 0.0794 | 0.0791 | 0.1169 | 0.1170 | 0.1966 | 0.1964 |
| ESE | 0.0660 | 0.0657 | 0.0736 | 0.0730 | 0.0911 | 0.0908 | 0.0891 | 0.0890 |
| CPr | 0.946 | 0.944 | 0.936 | 0.934 | 0.865 | 0.864 | 0.624 | 0.624 |
| $SM(\beta_1)$ | -0.9966 | -0.9970 | -1.0022 | -1.0016 | -1.0239 | -0.9825 | -1.0945 | -1.0099 |
| SSE | 0.1016 | 0.1009 | 0.1296 | 0.1294 | 0.2598 | 0.2490 | 0.7070 | 0.6753 |
| ESE | 0.1030 | 0.1011 | 0.1284 | 0.1279 | 0.1884 | 0.1811 | 0.2080 | 0.1731 |
| CPr | 0.954 | 0.952 | 0.956 | 0.952 | 0.876 | 0.848 | 0.646 | 0.646 |
| $SM(\beta_2)$ | 0.9992 | 0.9993 | 1.0025 | 1.0021 | 0.9988 | 0.9970 | 0.9983 | 1.0022 |
| SSE | 0.0325 | 0.0323 | 0.0364 | 0.0361 | 0.0527 | 0.0529 | 0.0769 | 0.0765 |
| ESE | 0.0318 | 0.0315 | 0.0357 | 0.0352 | 0.0425 | 0.0423 | 0.0385 | 0.0384 |
| CPr | 0.942 | 0.940 | 0.940 | 0.939 | 0.880 | 0.866 | 0.674 | 0.682 |
| $SM(\gamma)$ | 0.0499 | 0.0496 | 0.1995 | 0.1997 | 0.4997 | 0.4983 | 0.8992 | 0.8993 |
| SSE | 0.0053 | 0.0053 | 0.0082 | 0.0081 | 0.0110 | 0.0110 | 0.0101 | 0.0101 |
| ESE | 0.0050 | 0.0050 | 0.0077 | 0.0076 | 0.0081 | 0.0081 | 0.0039 | 0.0039 |
| CPr | 0.934 | 0.937 | 0.934 | 0.932 | 0.852 | 0.842 | 0.548 | 0.546 |

Table 3.6: Misspecified NS-AR(1) model under true LT model, where .

85

Chapter 4

Modeling Misclassified Longitudinal Categorical Data

4.1 Overview

Beginses that T_{ij} is the true has underwalds comparison transmost and Y_{ij} the denormal response for subject is of the jult in $g_{ij}(i_1 - 1)_{i_1,\dots,i_l}$ and $j = 1, 2,\dots, J$. It is assumed that the true comparison trapmass T_{ij} follows a multinemial distribution, that is, $T_{ij} \rightarrow Mathematical, <math>n_{ij_1}$. We also summer that the dynamic theorem of T_{ij} follows a multinemized transmoster and the G_{ij} state G_{ij} (G_{ij}) (G_{i is constant over time and subjects

$$\widehat{\Pi} = \begin{pmatrix} \pi_{11} & \pi_{12} & \cdots & \pi_{1,r+1} \\ \pi_{21} & \pi_{22} & \cdots & \pi_{2,r+1} \\ \vdots & \vdots & \ddots & \vdots \\ \pi_{r+1,1} & \pi_{r+1,2} & \cdots & \pi_{r+1,r+1} \end{pmatrix},$$

where $\pi_{ur} = P(Y_{ij(u)} = 1|T_{ij(r)} = 1), \pi_{u,r+1} = P(Y_{ij(u)} = 1|\mathbf{1}^{r}T_{ij} = 0). \pi_{s+1,r} =$ $P(\mathbf{1}'Y_{ij} = 0|T_{ij(v)} = 1) = 1 - \sum_{i=1}^{s} \pi_{kv_i}$ and $\pi_{s+1,r+1} = P(\mathbf{1}'Y_{ij} = 0|\mathbf{1}'T_{ij} = 0) =$ $1 - \sum_{i=1}^{s} \pi_{k,r+1}$ for u = 1, 2, ..., s and v = 1, 2, ..., r. The MC matrix II is given by

$$\Pi = \begin{pmatrix} \pi_{11} & \pi_{12} & \cdots & \pi_{1,r+1} \\ \pi_{21} & \pi_{22} & \cdots & \pi_{2,r+1} \\ \vdots & \vdots & \ddots & \vdots \\ \pi_{r,1} & \pi_{r,2} & \cdots & \pi_{r,r+1} \end{pmatrix} \triangleq [\pi_1, \pi_2, \dots, \pi_{r+1}],$$

where π 's are vectors of dimension s. If $1'\pi = 1$ for all i = 1, 2, ..., r + 1. If will become the FMC-matrix describes the misclassification from r + 1 inherent categories to s observed categories. Let $\Pi_r = [\pi_1, \pi_2, ..., \pi_r]$ be the submatrix of Π deleting the last column Text.

The misclassification model for the longitudinal data Y_{ij} and T_{ij} is given by

$$Y_{ij} = \Pi * \tilde{T}_{ij}$$

 $= \Pi_r * T_{ij} + \pi_{r+1} * (1 - 1^r T_{ij})$ (4.1)
 $= \sum_{u=1}^{r} \pi_s * T_{ij(u)} + \pi_{r+1} * (1 - 1^r T_{ij}).$

According to the definition of the generalized thinning operation, the expectation

of the observed categorical variable Y_{ij} in this model is given by the following formulas

$$\mu_{ij} = E(Y_{ij})$$

 $= E[\Pi_r * T_{ij} + \pi_{r+1} * (1 - 1^*T_{ij})]$
 $= (\Pi_r - \pi_{r+1} 1^i)' \eta_{ij} + \pi_{r+1}.$ (4.2)

The covariance matrix of Y_{ij} is

$$\begin{split} & (\mathbf{r}(i_{0}) = Vor(R(i_{0}), \mathbf{f}_{0}) + \mathcal{B}Vor(\mathbf{r}_{0}), \mathbf{f}_{0}) \\ & = Vor(R(i_{0}), \mathbf{f}_{0}) \sum_{i=1}^{N} \mathcal{B}_{i_{0}}(\mathbf{r}_{0}), \mathbf{f}_{i_{0}} = \mathcal{B}_{i_{0}}(\mathbf{r}_{0}), \mathbf{f}_{i_{0}}(\mathbf{r}_{0}), \mathbf{f}_{i_{0}}(\mathbf{r}$$

As shown in Section 2.3 of Chapter 2, $Y_{ij} \sim Maltinomial(1, \mu_{ij})$. Hence the covariance of Y_{ij} can be written in an alternative form

$$Var(Y_{ij}) = V_{\sigma_{ij}} = diag(\mu_{ij}) - \mu_{ij}\mu'_{ij}$$
 (4.3)

The covariance between Y_{ij} and Y_{in} is given by

$$\begin{split} Cov(Y_{ij},Y_{ik}) &= E(Y_{ij}Y_{ik}^{*}) - \mu_{ij}\mu_{ik}^{\prime} \\ &= E\{[\pi_{i+1}-(\Pi_{r}-\pi_{i+1}Y)T_{ij}]]\pi_{i+1}-(\Pi_{r}-\pi_{i+1}Y)T_{ik}]'\} \\ &- [\pi_{i+1}-(\Pi_{r}-\pi_{i+1}Y)\eta_{ij}][\pi_{i+1}-(\Pi_{r}-\pi_{i+1}Y)\eta_{ij}]' \\ &= (\Pi_{r}-\pi_{r+1}Y)Cov(T_{ij},T_{ik})(\Pi_{r}-\pi_{r+1}Y)'. \end{split}$$

These moments will be used to develop the GEE, GQL methods to estimate model parameters. Actually, it is much easier to calculate theses moments based on the explicit model (4.1) than that based on the classic descriptive misclassification model.

We now discuss the maximum likelihood (ML) approach which produces efficient estimators of the interested parameters. Suppose that we have observations of the manifest variable Y and the latent T, that is, $y = [u_i, -1, ..., I and j = 1, ..., I]$ and $t = (t_{ij}, i = 1, ..., I and j = 0, ..., I)$, where t_{ik} we baseline observations and assumed to be known. The complete likelihoof function is given by

$$L(\theta|_{\mathcal{B}}, t) = \prod_{i=1}^{t} f(y_{ii}|t_{i})f(t_{i})$$

 $= \prod_{i=1}^{t} \prod_{j=1}^{t} f(y_{ij}|t_{ij})\prod_{i=1}^{t} \prod_{j=1}^{t'} g_{i_{i},i_{j}j-1},$ (4.4)

where $g_{i,ij-1}$ is given in section 1 of Chapter34. Under some regularity conditions, such as all elements of FMC-matrix $\overline{\Pi}$ are within the interval (0, 1), which implies that $0 < r_{uv} < 1$, for $u = 1, \dots, s + 1$ and $v = 1, \dots, r + 1$, the conditional likelihood function of observations y_{ij} given t_{ij} is given by

$$f(y_0|q_0) = \left[(1 - \Gamma_{\pi_{+1}})^{(1-\Gamma_{\pi_{+1}})} \prod_{i=1}^{i} (1 - \Gamma_{\pi_{+1}})^{imp} \right]^{i=\Gamma_{\pi_{+1}}}$$

 $\times \prod_{i=1}^{i} \left[(1 - \Gamma_{\pi_{+1}})^{(1-\Gamma_{\pi_{+1}})} \prod_{i=1}^{i} \prod_{i=1}^{imp} \right]^{imp}$
 $= \left[\prod_{i=1}^{i} \prod_{i=1}^{i} \pi_{i}^{imp} m_{i=1}^{imp} \right] \left[\prod_{i=1}^{i} (1 - \Gamma_{\pi_{+1}})^{(1-\Gamma_{\pi_{+1}})(imp)} m_{i=1}^{imp} \right]$
 $\times \left[\prod_{i=1}^{i} \pi_{i}^{imp} m_{i=1}^{imp} \right] \left[(1 - \Gamma_{\pi_{+1}})^{(1-\Gamma_{\pi_{+1}})(imp)} m_{i=1}^{imp} \right]. (4.5)$

From expressions (4.4) and (4.5), it can be seen that it is difficult to calculate the marginal likelihood of the observed data g. Hence the ML estimates ramot be obtained by directly maximize the likelihood function of g. In this case, the expectation k maximization (EM) algorithm is helpful to calculate the ML estimates in an iterative provednes.

The log-likelihood function of complete data can be expressed as

$$\begin{split} \ell(\theta) &= \sum_{i=1}^{t} \sum_{j=1}^{t} \left[\sum_{i=1}^{t} [x_{ij}^{ij} (x_{ij}^{ij} + t_{ij}^{ij}, \tau_{ij}) k_{ij} (\alpha_{ij} - log) (1 + \sum_{i=1}^{t} exp(x_{ij}^{ij} x_{ij}^{ij}, \tau_{ij}^{ij}, \tau_{ij})) \right] \\ &+ \sum_{i=1}^{t} \sum_{j=1}^{t} \sum_{i=1}^{t} \sum_{i=1}^{t} \sum_{i=1}^{t} \sum_{i=1}^{t} exp(x_{ij}^{ij} x_{ij}^{ij}, \tau_{ij}^{ij}) exp(1 - 1Y_{ij}) exp(1 - 1Y_{ij}) (1 -$$

In this function, the values $t_{ij}, i = 1, ..., I$ and j = 1, ..., J are not observable. The EM algorithm starts with an initial value $\theta^{(0)}$. Denoting $\theta^{(0)}$ as the estimate of θ at the kth iteration, the (k + 1)th iteration of the EM can be developed below.

E-step: Find the expected complete-data log-likelihood function if θ were $\theta^{(k)}$: $Q(\theta|\theta^{(k)}, y) = E_T(\ell(\theta)|Y = y, \theta^{(k)})$

M-step: Determine $\theta^{(k+1)}$ by maximizing this expected log-likelihood function $Q(\theta|\theta^{(k)}, y)$. If the regularity conditions for π_{aa} 's are violated, for example, $\pi_{aa} = 0$, we have $y_{0(\eta)}t_{0(\eta)}(\log(\pi_{aa}) = 0$, in the log-likelihood function $\ell(\theta)$. Therefore the ML estimating recordence is still apolicable.

We consider two special cases in the following sections. In Section 4.2, we conduct the analysis of the misclassified longitudinal binary data. In section 4.4, the misclassified binary data with non-ignorable missing information are analyzed.

4.2 Misclassified Longitudinal Binary Data

In epideminique stations unda au techti authana prevention and ronthir pregnato, e abdril - dissues attein auto-dock-enternical basen da hein distanziani protectival by some proferma spostramations emiglende by the parents. Quantizamient are wicklyb und beams they are industrying simple and economical these measures that beam intertion and the station of the stations of the stationary of the distance profermation of each during beamset in the stationary of the distance stations, it is challenging associations for parents to identify separations of schemeling finant antitempointens. All of these means results in chamitations moves on influentions matterns, but here exists, we achieved parents the chamitation model and matternities the interview of targets from and regularizations. All of these means results in chamitations models and methods to identify with mini-industrial branching associations and are not relative informers to taking measurement errors that screents.

| | Asthma (T_{ij}) | | | | |
|------------------------------|-------------------|-------------|--|--|--|
| Diagnosis of test (Y_{ij}) | Infected (1) | Healthy (0) | | | |
| Positive (1) | π+ | c | | | |
| Negative (0) | ϵ^+ | π. | | | |

Table 4.1: Misclassified Asthma Status

4.2.1 Model description

Let V_{ij} denote the disgoned starting (partition-1, negatives) for the rdt dubl to the duble bias point of T_{ij} denote the corresponding true starting (indextol-1, healthy-ed), for i = 1, 2, ..., l and j = 1, 2, ..., J. The parallel diagrams for a duble starting and the starting of the correct diagram bias for a starting of the correct diagram bias for a bias bias of the correct diagram bias of the diagram bias of the correct diagram bias of the correct diagram bias of the diagram bias of the correct diagram bias of the correct diagram bias of the diagram bias of the correct diagram bias of the dia

$$\overline{\Pi} = \begin{pmatrix} \pi^+ & \epsilon^- \\ \epsilon^+ & \pi^- \end{pmatrix}.$$
(4.7)

53
To accommodate the biomedical background, we develop the misclassification model for binary data based on the two quantities, namely, the sensitivity π^+ and the specificity π^- .

It is assumed that $T_{ij} \sim b(1, \eta_{ij})$, where $\eta_{ij} = P(T_{ij} = 1)$ is a function of some risk factors or covariates. The classification error model is expressed as

$$Y_{ij} = \pi^{+} * T_{ij} + (1 - \pi^{-}) * (1 - T_{ij}),$$
 (4.8)

where \star is the binomial thinning operation described in Chapter 2. (V_{ij}, T_{ij}) may take one of the four pairs of values: (1, 1), (0, 0), (0, 1) and (1, 0). The former two cases indicate the two correct diagnoses, respectively, while the latter two cases imply two type of mindiagnoses. This model can completely and explicitly describe the mincleastication form T_{ij} to T_{ij} .

Based on the model (4.8), it is easy to calculate the required momenta for the development of the estimation approaches. The expectation and variance of Y_{ij} are given by

$$E(Y_{ij}) = \mu_{ij} = 1 - \pi^{-} + (\pi^{-} + \pi^{+} - 1)\eta_{ij}$$

and

$$Var(Y_{ij}) = \pi^{+}(1 - \pi^{+})\eta_{ij} + \pi^{-}(1 - \pi^{-})(1 - \eta_{ij}) + (\pi^{+} + \pi^{-} - 1)^{2}Var(T_{ij})$$

= $\mu_{ij}(1 - \mu_{ij}).$

The covariance between Y_{ij} and Y_{in} for u < j, is formulated by

$$Cor(Y_{ij}, Y_{ab}) = E(Y_{ij}Y_{ab}) - E(Y_{ij})E(Y_{ab})$$

$$= E\{[\pi^{+}T_{ij} + (1 - \pi^{-})(1 - T_{ij})][\pi^{+}T_{ia} + (1 - \pi^{-})(1 - T_{ia})]\}$$

$$-1 - \pi^{-}) + (\pi^{-} + \pi^{+} - 1)q_{ij}][(1 - \pi^{-}) + (\pi^{-} + \pi^{+} - 1)q_{a}]$$

$$= (\pi^{-} + \pi^{+} - 1)^{2}cor(T_{ij}, T_{ab}),$$

where $(\pi^- + \pi^+ - 1)$ is the Youden's index which captures the performance of a diagnostic test [Youden (1950)].

To develop the GQL and GQL2 approaches for the estimates of interested parameters, one needs to compute the moments of the response up to order four. Keeping in mind $Y_{ij}^{2} = Y_{ij}$, the calculations involved are given by

$$\begin{split} & R(Y_0Y_{00}) = (1 - e^{-1})^2 + (1 - e^{-\tau})(e^{-\tau} + e^{-\tau} - 1)(y_{00} + y_{01}) + (e^{-\tau} + e^{-\tau})^2 E(I_0T_{00}), \\ & R(Y_0Y_0Y_0) = R(\prod_{k=0,0}^{-1} (1 - e^{-\tau})^k + (e^{-\tau} + e^{-\tau} - 1)T_{00}) \\ & = (1 - e^{-1})^k + (1 - e^{-\tau})^k (e^{-\tau} + e^{-\tau} - 1)(y_{00} + y_{00} + y_{00}) \\ & + (1 - e^{-\tau})^k (e^{-\tau} + e^{-\tau} - 1)^k E(T_0T_0X_0) + E(T_0T_0)) \\ & + (e^{-\tau} + e^{-\tau} - 1)^k E(T_0T_0X_0). \end{split}$$

$$\begin{split} & R(t_{i}^{T}(X_{i}^{T}X_{i}^{T})) \\ & = & \lim_{t \to \infty^{-1}} \left[1 - t^{--1} \left(1 + t^{--1} + s^{+1} f_{i} \right) \right] \\ & = & \lim_{t \to \infty^{-1}} \left[(1 - s^{-1})^{+1} \left(1 - s^{-1} \right)^{T} (1 - s^{--1} - s^{+1}) g_{ij} + g_{ij} +$$

It is clear that these moments of the observed response Y are linear combinations of the 1st to 4th order moments of the true response T. The corresponding quantities about the true response T involved in the formulas can be calculated under the assumed model (3.29) of T in Section 3.1.2.

The following covariances under model (4.8) are needed for the parameter estimation in the GQL2 (it is OGQL under the assumed model (3.28) of T) framework. They are given by

$$\begin{split} &Cort(Y_0, Y_m, Y_m) = (r^- * r^- 1)^2 (1 - \pi^-)^2 Cort(Y_0, T_m + T_m) \\ &- (r^- + \pi^- - 1)^2 Cort(Y_0, T_m, T_m) \\ &- (r^- * \pi^- - 1)^2 Cort(Y_0, T_m, T_m) \\ &+ (r^- + \pi^- - 1)^2 (1 - \pi^-)^2 Cort(Y_0 + T_m, T_m, T_m) \\ &+ (r^- + \pi^- - 1)^2 (1 - \pi^-)^2 Cort(Y_0 + T_m, T_m) \\ &+ (r^- + \pi^- - 1)^2 (1 - \pi^-)^2 Cort(Y_0 + T_m, T_m) \\ &+ T_m (1 - \pi^-)^2 Cort(Y_m, T_m, T_m) \end{split}$$

and

In addition, since $Y_{ij}^2 = Y_{ij}$, we have

$$Cov(Y_{ij}, Y_{iu}Y_{ij}) = E(Y_{ij}Y_{iu})(1 - E(Y_{ij})),$$

 $Cov(Y_{ij}Y_{iu}, Y_{iu}Y_{ij}) = E(Y_{ij}Y_{iu}Y_{iu}) - E(Y_{ij}Y_{iu})E(Y_{ij}Y_{iu})$

which are linear combinations of the moments of the true responses.

Note that if we have a perfect specificity $\pi^- = 1$, that is, the completely exact diagnoses among the healthy population, then the 1st to 4th moments given above can be greatly simplified as follows:

$$\mu_{ij} = \pi^+ \eta_{ij},$$

 $E(Y_{ij}Y_{ik}) = (\pi^+)^2 E(T_{ij}T_{ik}),$
 $E(Y_{ij}Y_{ik}Y_{ik}) = (\pi^+)^2 E(T_{ij}T_{ik}T_{ik}),$
 $E(Y_{ij}Y_{ik}Y_{ik}) = (\pi^+)^4 B(T_{ik}T_{ik}T_{ik}),$

It follows that the covariances can be simplified by

$$Cor(Y_{ij}, Y_{ik}) = (\pi^+)^2 Cor(T_{ij}, T_{ik}),$$

 $Cor(Y_{ij}, Y_{ik}Y_{ij}) = (\pi^+)^2 Cor(T_{ij}, T_{ik}T_{ij}),$
 $Cor(Y_{ij}, Y_{ik}Y_{ik}) = (\pi^+)^2 Cor(T_{ij}, T_{ik}T_{ik}),$
 $Cor(Y_{ij}Y_{ik}, Y_{ik}Y_{ij}) = (\pi^+)^4 Cor(T_{ij}T_{ik}, T_{ik}T_{ij}),$
 $Cor(Y_{ij}Y_{ik}, Y_{ik}Y_{ij}) = (\pi^+)^4 Cor(T_{ij}T_{ik}, T_{ik}T_{ij}),$

In addition, the variance of Y_{ij} can be simplified as follows:

$$Var(Y_{ij}) = \pi^+ \eta_{ij} (1 - \pi^+ \eta_{ij})$$

= $(\pi^+)^2 Var(T_{ij}) + \pi^+ (1 - \pi^+) \eta_{ij}$

Therefore, in case of perfect specificity, the estimating equations approaches become much simpler.

To be specific in developing estimation approaches, we assume that the true response T_{ii} follow the transition model (3.28). We rewrite it as

$$f'_{ij} = P(T_{ij} = 1|T_{i,j-1} = t_{i,j-1})$$

= $\frac{exp(x'_{ij}\beta + t_{i,j-1}\gamma)}{1 + exp(x'_{ij}\beta + t_{i,j-1}\gamma)}$, for j=1,2,..., m, (4.9)

The 1st-4th moments of T under this model can be found in Section 3.1.2.

4.2.2 Estimation of the model effects

To device the methodology see summa that the semility r^{*} and the predibity of r^{*} as theore, mind by the simplicity of the devices. In fact, is epseloadiopid studies, even if s^{*} and r^{*} are allows, sum resemble estimates can be obtained from previously similar rankes as from independent validation attackes of the data studies as dense, or how some nore exact chics examination of a relatively studi sample Bips, Baneyie and Malli (BBG). We further assume that for two responts follows the transition of (19) with baneba corrections $t_{0} = 0$.

4.2.2.1 GQL method

When the sensitivity π^+ and the specificity π^- are known, the parameters of interest are $\theta = (\beta^*, \gamma)^r$ from the transition model (4.9), where $\beta = (\beta_1, ..., \beta_p)^r$ is the vector of regression coefficients, and γ is the dynamic dependence parameter.

To estimate the model parameters based on the GQL method, we solve the fol-

lowing estimating equations

$$\sum_{i=1}^{I} \frac{\partial \mu'_{ij}}{\partial \theta} \Sigma_i^{-1}(y_i - \mu_i) = 0, \quad (4.10)$$

where $y_k = (y_{k1}, ..., y_{kJ})^*$, $\mu_e = (\mu_{k1}, ..., \mu_{kJ})^*$, and $\partial_{jk_i}/\partial \theta$ is the first order derivative matrix of μ_i with respect to θ of dimension $(p + 1) \times J$ for known π^+ and π^- . The jith element of $\partial_{jk_i}/\partial \theta$ is given by

$$\frac{\partial \bar{y}_{ij}}{\partial \beta_k} = (\pi^- + \pi^+ - 1) \{ \bar{\eta}_0 (1 - \bar{\eta}_0) \eta_{i,j-1} + \bar{\eta}_0 (1 - \bar{\eta}_0) (1 - \eta_{i,j-1}) \} x_{ij(k)} \\ + (\bar{\eta}_0 - \bar{\eta}_0) \frac{\partial \bar{y}_{i,j-1}}{\partial \beta_k} \qquad (4.11)$$

for k = 1, ..., p, j = 1, ..., J, and

$$\frac{\partial p_{ij}}{\partial \gamma} = (\pi^- + \pi^+ - 1) \sum_{a=1}^{j} \eta_{ia-1} \bar{\eta}_{ia} (1 - \bar{\eta}_a) \prod_{k=a+1}^{j} (\bar{\eta}_k - \bar{\eta}_k),$$
 (4.12)

for j = 2, ..., J and $\partial \eta_{21}/\partial \gamma = 0 \Rightarrow \partial \eta_{21}/\partial \gamma = 0$. Σ_i is the variance-covariance matrix of V_i . If Σ_i is replaced with a general "working" covariance matrix W_i in the case that Σ_i is unknown, the GQL approach becomes the GEE approach [Liang and Zeger (1986].

Once we have the estimate $\hat{\theta}_{GQE}$, the corresponding consistent estimate of the covariance matrix of $\hat{\theta}_{GQE}$ is given by

$$\hat{V}(\hat{\theta}_{GQL}) = \left(\sum_{k=1}^{I} \frac{\partial \mu'_{i}}{\partial \theta} \sum_{k}^{-1} \frac{\partial \mu_{i}}{\partial \theta}\right)^{-1} |_{\theta = \hat{\theta}_{GQL}},$$
 (4.13)

4.2.2.2 Maximum likelihood method

In this subsection, we develop the maximum likelihood (ML) approach under regularity conditions, such as imperfect sensitivity and specificity, that is $0 < \pi^+, \pi^- < 1$. The complete likelihood function given observations Y = y and T = t is formulated

by

$$L(\theta|g, t) = \prod_{i=1}^{l} f(g_i, t_i) \text{ nonsmber}$$
 (4.14)
 $= \prod_{i=1}^{l} f(g_i|t_i)f(t_i)$
 $= \prod_{i=1}^{l} \prod_{j=1}^{l} f(g_i|t_j) \prod_{i=1}^{l} \prod_{j=1}^{j} g_{i,(j)-1},$ (4.15)

where $g_{0|l-1}$ are given in Section 3.1.2, and

$$f(y_{ij}|t_{ij}) = [(\pi^+)^{y_{ij}}(1 - \pi^+)^{1-y_{ij}}]^{t_{ij}}[(\pi^-)^{1-y_{ij}}(1 - \pi^-)^{y_{ij}}]^{1-t_{ij}}$$

= $(\pi^+)^{y_{ij}t_{ij}}(1 - \pi^+)^{(1-y_{ij})t_{ij}}(\pi^-)^{(1-y_{ij})(1-t_{ij})}(1 - \pi^-)^{(1-t_{ij})y_{ij}}.$ (4.16)

The log-likelihood function can be expressed as

$$l(\theta, t, y) = \sum_{i=1}^{d} \zeta_{i}[\theta_{i}, y_{i}]$$

 $= \sum_{i=1}^{d} \sum_{j=1}^{d} \zeta_{i}[\theta_{i}, y_{j}] + \gamma t_{i,j-1}] - \sum_{i=1}^{d} \sum_{j=1}^{d} \log[1 + \exp\{t_{i}[\theta] + \gamma t_{i,j-1}]]$
 $+ \sum_{i=1}^{d} \sum_{j=1}^{d} \log_{i}[t_{i}] - t_{ij}[\log(1 - \pi^{*}) + \sum_{i=1}^{d} \sum_{j=1}^{d} (1 - y_{i})[t_{i}]\log\log(1 - \pi^{*})]$
 $+ \sum_{i=1}^{d} \sum_{j=1}^{d} \log_{i}[t_{i}]\log\pi^{*} + \sum_{i=1}^{d} \sum_{j=1}^{d} (1 - y_{i})(1 - t_{i})\log\pi^{*}.$ (4.17)

In $\ell(\theta; t, y)$, the values $t_{ij}, i = 1, ..., I, j = 1, ..., J$ are not observable. We, therefore, apply the EM algorithm to find the ML estimates of the model parameters. Given an initial value $\theta^{(0)}$, we denote $\theta^{(0)}$ as the estimate of θ at the kth iteration. The (k + 1) the training of the EM algorithm can be derived as follows: E-step: Find the expected complete-data log-likelihood function if θ were $\theta^{(k)}$:

$$\begin{split} & (Q(\theta^{(0)}, X)) = y_1(\theta^{(0)}) \\ &= \sum_{i=1}^{d} \sum_{j=1}^{d} Q_{ij}^{(0,1)} e_{ij}^{(0,1)} (e_{ij}^{(0,1)} + e_{ij}^{(0,1)}) \\ &= \sum_{i=1}^{d} \sum_{j=1}^{d} Q_{ij}^{(0,1)} e_{ij}^{(0,1)} (e_{ij}^{(0,1)} + e_{ij}^{(0,1)}) (e_{ij}^{(0,1)} + e_{ij}^{(0,1)}) \\ &= \sum_{i=1}^{d} \sum_{j=1}^{d} y_{ij}^{(0,1)} (e_{ij}^{(0,1)} + \sum_{j=1}^{d} (1 - y_{ij}) (e_{ij}^{(0,1)}) (e_{ij}^{(0,1)}) \\ &= \sum_{i=1}^{d} \sum_{j=1}^{d} y_{ij}^{(0,1)} (e_{ij}^{(0,1)}) (e_{ij}^{(0,1)} - x^{(0,1)}) \\ &+ \sum_{i=1}^{d} \sum_{j=1}^{d} y_{ij}^{(0,1)} (e_{ij}^{(0,1)}) (e_{ij}^{(0,1)} - x^{(0,1)}) \\ &+ \sum_{i=1}^{d} \sum_{j=1}^{d} y_{ij}^{(0,1)} (e_{ij}^{(0,1)}) (e_{ij}^{(0,1)} - x^{(0,1)}) \\ &+ \sum_{i=1}^{d} \sum_{j=1}^{d} y_{ij}^{(0,1)} (e_{ij}^{(0,1)}) (e_{ij}^{(0,1)} - x^{(0,1)}) \\ &+ \sum_{i=1}^{d} \sum_{j=1}^{d} y_{ij}^{(0,1)} (e_{ij}^{(0,1)} - x^{(0,1)}) \\ &+ \sum_{i=1}^{d} y_{ij}^{(0,1)} (e_{ij}^{(0,1)} - x^{(0,1)}) \\ &+ \sum$$

where

$$t_{ij}^{(k+1)} = E_{g_{(k)}}(T_{ij}|Y_i = y_i) = \frac{P_{g_{(k)}}(Y_i = y_i, T_{ij} = 1)}{P_{g_{(k)}}(Y_i = y_i)}$$

(4.19)

and

$$(l_{ql}t_{d-1})^{(k+1)} = E_{\phi w}(T_{0}T_{d-1}); (= y_k) = \frac{P_{\phi w}(Y_i = y_k, T_{d-1} = T_iT_{d-1} = 1)}{P_{\phi w}(Y_i = y_k)}$$
. (420)
Let $\Omega^{\phi} = \{1, 2, ..., w\}, \Omega^{\phi}_{j} = \Omega^{\phi} \setminus \{j\}$ and $\Omega_{j,w,1} = \Omega^{\phi} \setminus \{j, j = 1\}$. For
iven *i*, we denote $A = \{k : t_k = 1, k \in \Omega^{\phi}_{j-1}\}, k \in \Omega^{\phi}\}, A_j = \{k : t_k = 1, k \in \Omega^{\phi}_{j}\}$ and
 $l_{d-1} = \{k : t_k = 1, k \in \Omega^{\phi}_{j-1}\}, where t_j is the space f_j is the probabilities involved$

in (4.18) and (4.19) can be calculated as

$$\begin{split} &P(Y_i = y_i) \\ &= \sum_{i \neq j \neq i} P(Y_i = y_i) (T_i = t_i) P(T_i = t_{ij}) \prod_{j \neq j} P(T_{ij} = t_{ij}) I(P(T_{ij} = t_{ij})) P(T_{ij} = t_{ij}) \\ &= \sum_{i \neq j \neq i} P(Y_i) \sum_{i \neq j \neq i} I(T_i = T_i) \sum_{i \neq j \neq i} I(T_i) \prod_{i \neq j \neq i} I(T_i) \sum_{i \neq j \neq i} I(T_i) \prod_{i \neq j \neq i} I(T_i) \sum_{i \neq j \neq i} I(T_i) \prod_{i \neq$$

$$\begin{split} &P(Y_i = a_i, T_i = 1) \\ &= \sum_{\substack{A \in I^{(T)} \\ A \in I^{(T)}}} P(Y_i = a_i, (T_{i_1}, \dots, T_{i_j-1}, T_{i_j-1}, \dots, T_{i_j}) = (b_{i_1}, \dots, b_{i_j-1}, 1, b_{i_j+1}, \dots, b_{i_j}) \\ &= \sum_{\substack{A \in I^{(T)} \\ A \in I^{(T)}}} (P^{(T)} (P^{(T)} \sum_{\substack{A \in I^{(T)} \\ A \in I^{(T)}}} (P^{(T)} (P^{(T)} \sum_{\substack{A \in I^{(T)} \\ A \in I^{(T)}}} (P^{(T)} ($$

and

where |A| denotes the size of A, $\bar{A} = A \cup \{k + 1 : k \in A\}$, and $I_{\{k+1cA\}} = 1$ if $k + 1 \in A$, 0, otherwise. $2^{\Omega^{\#}} = \{B : B \subseteq \Omega^{\#}\}$ consisting of all subsets of $\Omega^{\#}$.

M-step: Determine $\theta^{(k+1)}$ by maximizing the expected log-likelihood. To do so, we solve equations:

$$S^{*}(\theta) = \frac{\partial Q}{\partial \theta} = \begin{pmatrix} \frac{\partial Q(\theta)\theta^{(0)}(y)}{\partial J} \\ \frac{\partial Q(\theta)\theta^{(0)}(y)}{\partial \gamma} \end{pmatrix} = 0,$$

where

$$\frac{\partial Q(\theta|\theta^{(k)}, y)}{\partial \beta} = \sum_{i=1}^{J} \sum_{j=1}^{J} [t_{ij}^{(k+1)} - \bar{\eta}_{ij} - (\bar{\eta}_{ij} - \bar{\eta}_{ij})t_{i,j-1}^{(k+1)}]x_{ij}, \quad (4.21)$$

$$\frac{\partial Q(\theta|\theta^{(k)}, y)}{\partial \gamma} = \sum_{i=1}^{J} \sum_{j=1}^{J} [(t_{ij}t_{i,j-1})^{(k+1)} - \bar{\eta}_{ij}t_{i,j-1}^{(k+1)}]. \quad (4.22)$$

Due the ML estimates $h_{0,1}$ are achieved from the DM algorithm, we can estimate the wintersevenessissme matrix of the estimates $\hat{\theta}_{0,1}$ by the interest of the absences Hocket inference on the first inference on the distribution matrix h_{-} . Denote the first and would estimate the distribution of the log-like high-matrix $(\theta_{0,1}, \eta_{0,2}) = d\hat{\theta}_{0,1}^{(0,1)} = d\hat{\theta}_{0,1}^{(0,1)$

$$\begin{split} &I_{T} = -E_{0}[B(\theta; Y, T)|y| - E_{0}[S(\theta; T, Y)|S'(\theta; T, Y)|y| + E_{0}[S(\theta; T, Y)|y| E_{0}[S(\theta; T, Y)|y| \\ &= E_{0}[B(\theta; Y, T)|y| - \sum_{i=1}^{i} E_{0}[S(\theta; T_{i}, Y_{i})S(\theta; T_{i}, Y_{i})|y_{i}] \\ &+ \sum_{i=1}^{i} E_{0}[S(\theta; T_{i}, Y_{i})|y_{i}] E_{0}[S(\theta; T_{i}, Y_{i})|y_{i}]. \end{split}$$

For multinomial variables, I_Y reduces to

$$\sum_{i=1}^{l} E_{\theta}[S_{i}(\theta; T_{i}, Y_{i})|y_{i}]E_{\theta}[S_{i}^{i}(\theta; T_{i}, Y_{i})|y_{i}] \qquad (4.23)$$

due to the fact that

$$E[B_i(\theta; Y_i, T_i)] = E[S_i(\theta; T_i, Y_i)S'_i(\theta; T_i, Y_i)].$$
 (4.24)

However, similarizes subset that or initiality at an infer arress derived from freeming (122) and the understanding of the two marked errors on MERC based D4 Mapertun. It may be because the differences between the observed values of the two subset of expansion (124) and maximum based on functions (124). For the difference of the Fulder that the observed or initiative of the Fulder to the difference of the fulder of the fuller difference of the fuller of the fuller difference of the fuller of the fuller difference o

information matrix as the expected complete-data log-likelihood function (4.17), that

18,

$$\begin{split} & h_{TY} = -\mathcal{E} \left(\frac{\partial^2 \ell(\theta)}{\partial \theta \partial \theta'} \right) \\ & = \sum_{i=1}^{2} \sum_{j=1}^{2} \left[\begin{array}{c} x_{ij} (\hat{u}_{ij}(1 - \hat{u}_{ij}) u_{ij-1} + \hat{u}_{ij}(1 - \hat{u}_{ij}) (1 - u_{ij-1})] x'_{ij} & \hat{u}_{ij} (1 - \hat{u}_{ij}) u_{ij-1} x'_{ij} \\ & \hat{u}_{ij} (1 - \hat{u}_{ij}) u_{ij-1} x'_{ij} & \hat{u}_{ij} (1 - \hat{u}_{ij}) u_{ij-1} x'_{ij} \end{array} \right] \end{split}$$

which can be evaluated at the final estimate $\hat{\theta}_{ML}$. Furthermore, we define another quantity as

$$\begin{split} I_Q &= -E\left(\frac{\partial^2 Q}{\partial \partial \theta \theta'}\right) \\ &= \sum_{i=1}^{I} \sum_{g=1}^{T} \begin{bmatrix} x_{ij} [\eta_{ij}(1-\bar{\eta}_{ij})(1-\bar{\eta}_{ij})^{(h+1)}_{i,j-1} + \bar{\eta}_{ij}(1-\bar{\eta}_{ij})x_{i,j-1}^{(h+1)} p'_{ij} & \bar{\eta}_{ij}(1-\bar{\eta}_{ij})x_{i,j-1}^{(h+1)} x_{ij} \\ \bar{\eta}_{ij}(1-\bar{\eta}_{ij})x_{i,j-1}^{(h+1)} p'_{ij} & \bar{\eta}_{ij}(1-\bar{\eta}_{ij})x_{i,j-1}^{(h+1)} \end{bmatrix} \end{split}$$

In DM algorithm, k_{i} can be estimated by employing $\xi^{(i+1)}$ from the last Lenge of an off in δ_{i}^{i} . GeV for the smalley of δ_{i}^{i} for last i_{i}^{i} , the deviation of the starting equal to $S(\tilde{h}_{0L}, t_{0,k})$ by replacing t_{i} and $t_{i}(t_{i-1})$ with $\xi^{(i+1)}_{i}$ and $(t_{i}(t_{i-1})^{i+1})$ from the hat. Early, respectively. Our simulations research that the estimated t_{i}^{i} the deviation of the starting of the deviation of the starting of the deviation of

$$I_{Y}^{c} = (I_{T,Y} - I_Q + \sum_{i=1}^{I} S_i^* S_i^*)|_{\theta_{WL}}.$$
 (4.25)

Therefore, the estimate of $\hat{\theta}_{NL}$ can be given by

$$\widehat{Var}(\hat{\theta}_{ML}) = (I_Y^*)^{-1}.$$
 (4.26)

It is shown from the simulation in the next section that the new estimator (4.26) can consistently estimate true $V(\hat{\theta}_{NL})$ in EM algorithm. In addition, this new retinuator is in good concordance with $\hat{V}(\hat{\theta}_{NQL})$ which will be given in the Section 4.2.2.

Remark:

1. Suppose that we have the observations of $T_{ij} = t_{ij}$ and $Y_{ij} = y_{ij}$. As mentioned in Section 4.1, if we have perfect specificity, then $(1-t_{ij})y_{ij} = 0$ and $1 - \pi^- = 0$. And we define that $\theta^0 = 1$. The conditional likelihood part in the full likelihood function then one be modified to accommodate this case, and it is given by

$$f(y_{ij}|t_{ij}) = (\pi^+)^{y_{ij}t_{ij}}(1 - \pi^+)^{(1-y_{ij})t_{ij}}(\pi^-)^{(1-y_{ij})(1-t_{ij})}(1 - \pi^-)^{(1-t_{ij})y_{ij}}$$

= $(\pi^+)^{y_{ij}t_{ij}}(1 - \pi^+)^{(1-y_{ij})t_{ij}}$ (4.27)

Then the corresponding log-likelihood function can be expressed as

$$\ell(\theta) = \sum_{i=1}^{\ell} \sum_{j=1}^{\ell} t_{ij} [x_{ij}^{i} \beta + \gamma t_{i,j-1}) - \sum_{i=1}^{\ell} \sum_{j=1}^{\ell} \log[1 + exp[x_{ij}^{i} \beta + \gamma t_{i,j-1})] \\ + \sum_{i=1}^{\ell} \sum_{j=1}^{\ell} y_{ij} t_{ij} \log \pi + \sum_{i=1}^{\ell} \sum_{j=1}^{\ell} (1 - y_{ij}) t_{ij} \log(1 - \pi^{+}) \qquad (4.28)$$

The EM algorithm can be developed based on the log-likelihood function (4.28).

2. In the case of perfect sensitivity, a similar EM procedure can be developed.

 The GQL approach developed in Section 4.2.2.1 automatically accommodates these two special cases.

4.2.2.3 GQL2 (OGQL) method

Here we develop the GQL2 approach, that is, the optimal GQL (OGQL) approach in Sutradhar and Farrell (2007) to estimate the model parameters under the for each right: matrixe model (12) for dynamic basicy data. This opposed, on also accumuload the one of perform sensitivy or specificity. As mutticed by Satzahlar and Farrell (2007), moler the dynamic dynamics model (4.9), the estimating equations have one hole by $M_{\rm eff}$ or $M_{\rm eff}$ or $M_{\rm eff}$ or $M_{\rm eff}$ or $M_{\rm eff}$ and the scored order grown. They demonstration that the estimators obtained by OGQL are almost as efficient as the ML estimator but with force assumptions on the model. However, their combinion was reached under the assumption that there is no michaedification in the dubt.

In this section, we dowedputhe GQU approach by explaining the fibra allow of order statistics of the barge responses with interdendicion the topic spins the second. It will be down that the GQL method behaves similar to the maximum likelihood approach. In simulation. Therefore, in presence of maintametization (102) approach motion to also also the OGQ approach motion the disk of the the OGQL spreases intensis to also also the OGQ approach motion the disk of the OGQL are made in statistication. There also also the OGQL approach motion the disk of the OGQL approach motion the also also the OGQ approach motion the disk of the OGQL approach motion that the of the OGQL approach motion the disk of the disk

Let f_i be the observation of $F_i = (Y_i^i, S_i^j)^i$, where $Y_i = (Y_{i1}, \dots, Y_{iJ})^i$, and $S_i = (Y_{i1}Y_{i2}, \dots, Y_{in}Y_{in}, \dots, Y_{i,J-1}Y_{iJ})^i$, then $\delta_i = E(F_i) = (\mu_i^i, \nu_j^i)^i$, where $\nu_i = E(S_i)$. Further, let

$$\Omega_i = \begin{pmatrix} Cov(Y_i) & Cov(Y_i, S_i) \\ \\ Cov(S_i, Y_i) & Cov(S_i) \end{pmatrix}$$

be the $J(J + 1)/2 \times J(J + 1)/2$ covariance matrix. The estimating equations take the same form as those in [Sutradhar and Farrell (2007)]. But the meaning of the components in our setting is different from the cases in [Sutradhar and Farrell (2007)]. Our OGQL approach is given as

$$\sum_{i=1}^{l} \frac{\partial \delta_i^c}{\partial \theta} \Omega_i^{-1} (f_i - \delta_i) = 0 \qquad (4.29)$$

The quantities required in this equation such as μ_i , ν_i , $Cov(Y_i)$, $Cov(S_i, Y_i)$ and $Cov(S_i)$ can be calculated based on the moments given in section 4.2.1.

Now, let ζ_i denote the expected value of the true response S_i , with elements $\zeta_{uv} = E(T_{iv}T_{iv})$. The elements of the first order derivative of δ_i with respect to θ are formulated by

$$\frac{\partial \delta_{nuv}}{\partial \theta} = (1 - \pi^+ - \pi^-)^2 \frac{\partial \zeta_{nuv}}{\partial \theta} - (1 - \pi^-)(1 - \pi^- - \pi^+) (\frac{\partial \eta_n}{\partial \theta} + \frac{\partial \eta_n}{\partial \theta}),$$

where $\frac{\partial q_{0}}{\partial s}$'s are given by expressions (2.2) and (2.3) in the paper [Sutradhar and Farrell (2007)]. $\frac{\partial q_{0}}{\partial s}$'s are given in the following equations

$$\frac{\partial \zeta_{iuss}}{\partial \beta} = \sum_{l_{10}, l_{10} \notin S^0} [g_{11}^* \prod_{j=2}^m g_{i,j|j-1}^* (\sum_{k=2}^m (t_{ik} - \lambda_{ik|k-1}^*) x_{ik} + (t_{i1} - \eta_{i1}) x_{il})]_{t_{in}=1, t_{ir}=1}, \quad (4.30)$$

$$\frac{\partial \zeta_{487}}{\partial \gamma} = \sum_{t_{10}, t_{10} \notin S^0} [g_{11}^{t_{11}} \prod_{j=2}^{m} g_{1,jj-1}^{s} \{ \sum_{k=2}^{m} (t_{ik} - \lambda_{ik(k-1)}^{s} t_{ik}) \}_{t_{10}=1, t_{20}=1}$$
 (4.31)

[Sutradhar and Farrell (2007)].

Once the estimate $\hat{\theta}_{OOQL}$ is obtained, the corresponding covariance matrix $V(\hat{\theta}_{OOQL})$ can be consistently estimated by

$$\hat{V}(\hat{\theta}_{OOQL}) = \left(\sum_{i=1}^{I} \frac{\partial \theta_{i}^{*}}{\partial \theta} \Omega_{i}^{-1} \frac{\partial \delta_{i}}{\partial \theta}\right)^{-1} |_{\theta = \hat{\theta}_{OOQL}}.$$
 (4.32)

4.2.3 Simulation Studies

As those in some other estimating-equations-based approaches, the estimators proposed in this thesis are all consistent and are asymptotically normally distributed. In this subsection, we study the proporties of the proposed estimators, namely, the QQL, OGQL and ML estimators, through Monte Carlo simulation. Random samples with a size similar to that of a subset of the BBSC study were generated based on different designs of covariates. Model parameters are then estimated by using the three proposed approaches.

4.2.3.1 Covariate designs

In the simulation study, we consider I = 560 independent subjects each with J = 4 repeated observations. The true data $l_{ij}, i = 1, 2, ..., 660$ and j = 1, 2, 3, 4, are generated following the dynamic model (4.5), and the observed data y_{ij} are generated following the mischastification model (4.8). To be opecific, y_{ij} can be generated following the proveduce doscribed below.

Once we have t_{ij}, we can firstly generate a binomial variable U from binomial(t_{ij}, π⁺).
 Secondly, we generate another binomial variable V from binomial(1-t_{ij}, 1-π⁻).

(3) Lastly, we get y_{ij} = U + V.

Sutradhar and Farrell (2007) conducted simulations for error-free binary data based on the following three covariate designs. In our simulation studies in this section, we also consider these three designs for the analysis of mis-measured longitudinal binary data. The three designs are given by:

Design 1: $x_{ii(1)} = 1$ and $x_{ii(2)} = j/4$ for i = 1, ..., 560 and j = 1, ..., 4.

Design 2: $x_{ij(1)} = 1$, (j = 1, 2); $x_{ij(2)} = 0$, (j = 3, 4), i = 1, ..., 140;

 $x_{ii(1)} = 1, j = 1, \dots, 4, i = 141, \dots, 420;$

$$x_{ij(1)} = 0$$
, $(j = 1, 2)$; $x_{ij(1)} = 1$, $(j = 3, 4)$, $i = 421, \dots, 560$.
 $x_{il(2)} = j/4$, $j = 1, \dots, 4$, $i = 1, \dots, 560$.

$$\begin{split} \text{Design} & \exists : r_{0(1)} = 1, \ (j = 1, 2; r_{0(2)} = 4, \ (j = 3, 0; i = 1, \dots, 10), \\ & r_{0(2)} = 1, \ j = 1, \dots, 4, \ i = 141, \dots, 630, \\ & r_{0(2)} = 1, \ (j = 1, 2; r_{0(2)} = -1, (j = 3, 1), i = 1, \dots, 10), \\ & r_{0(2)} = -65, \ (j = 1, 2); \ r_{0(2)} = -65, \ (j = 3, 4), \ (i = 1, \dots, 10), \\ & r_{0(2)} = -65, \ (j = 1, 2); \ r_{0(2)} = -65, \ (j = 3, 4), \ (i = 1, \dots, 10), \\ & r_{0(2)} = -65, \ (j = 1, 2); \ r_{0(2)} = -65, \ (j = 3, 4), \ (i = 1, \dots, 10), \\ & r_{0(2)} = -65, \ (j = 1, 2); \ r_{0(2)} = -5, \ (j = 3, 4), \ (i = 1, \dots, 10), \\ & r_{0(2)} = -65, \ (j = 1, 2); \ r_{0(2)} = -5, \ (j = 3, 4), \ (j = 1, \dots, 10), \\ & r_{0(2)} = -65, \ (j = 1, 2); \ r_{0(2)} = -1, \ (j = 1, \dots, 10), \\ & r_{0(2)} = -65, \ (j = 1, 2); \ r_{0(2)} = -1, \ (j = 1, \dots, 10), \\ & r_{0(2)} = -65, \ (j = 1, 2); \ r_{0(2)} = -1, \ (j = 1, \dots, 10), \\ & r_{0(2)} = -65, \ (j = 1, 2); \ r_{0(2)} = -1, \ (j = 1, 2); \ (j = 1$$

Simulation results corresponding to different settings of the countate designs and the choices of parameter values are presented in the subsequent subsections in the form of tables.

4.2.3.2 Estimation of the model parameters

In this subscrite, we carry out a simulation study to can use the three models of the state of

$$\beta = (\beta_1, \beta_2)^r = (1, 1)^r$$
;

 $\gamma = 1.5, 0, and - 1;$

$(\pi^+, \pi^-) = (0.95, 0.90)$ and (0.75, 0.80),

where $\gamma = 0$ implies dynamic independence. The higher values of sensitivity π^+ and the specificity π^- , that is, (0.56, 0.90) in the simulation, means loss classification error for an excellent diagnostic test, whereas the lower values of this two quantities, for example, (0.375, 0.80) in the simulation, implies more classification error for a poor diamontic test.

The simulation image (SB), simulated standard rener (SBS), estimated standard even (SBS), and accomp probabilities (CF) of the 46% confidence intervals for the interacted parameters β and γ are repetited in Tables 4.2-4.4 when (π^+,π^-) (80,6,030) under the constants design 1, 2, and 3, respectively. Similarly, buy expressed SM, SBS, Eq. (CY) of g and γ are repetited in Tables 4.2-4 when $(\pi^+,\pi^-) = (0.75,0.80)$ under the constants designs 1, 2, and 3, respectively. In therefore follows that a confidence interact on the constants of by under $\xi_{1,2}$ (SE3.6), under $\xi_{1,2}$ (SE3.6), where $\xi_{1,2}$ (SE3.6), where $\xi_{1,2}$ (SE3.6), using $\xi_{1,2}$ (SE3.6), where $\xi_{2,2}$ (SE3.6), where $\xi_{2,2}$ (SE3.6), using $\xi_{2,2}$ (SE3.6), where $\xi_{2,2}$ (SE3.6), where $\xi_{2,2}$ (SE3.6), we an adjusted near $\xi_{2,2}$ (SE3.6), where $\xi_{2,2}$ (SE3.6), we determine $\xi_{2,2}$ (SE3.6), we determine $\xi_{2,2}$ (SE3.6), where $\xi_{2,2}$ (SE3.6), we determine $\xi_{2,3}$ (SE3.6), we determine $\xi_{2,3}$ (SE3.6), we determine $\xi_{3,3}$ (SE3.6), we determine $\xi_{3,3}$ (SE3.6), where $\xi_{3,3}$ (SE3.6), we determine $\xi_{3,3}$ (SE3.6), we determine $\xi_{3,3}$ (SE3.6), where $\xi_{3,3}$ (

Furthermore, in outs of Table 5.2.15, three different estimations are compared blue ends of the GG (2004), and ML approxed. Namely, they are the ideal estimates (1) when the absorptions of the error-form response T are such, the mice contrastice (2) when the discretations of the three-form response T are such, the mice corrected softmates (2) taking characteristic mere the data $\{\mu_{2}\}$ has consideration. The assume model (10) can be used for both is fold minimum can full the mice estimates, the low mice of the size of the size of the size of minimum discretations are sized for the data of the size of minimum ends of the size of the size of the size of the size of minimum ends of the size of minimum ends of the size of the ends of the size of the ends of the size of the end of the size of the end of the size of the end of the size of the end of the size of the end of the size of the end of the size of the end of the size of the end of the size of the end of the size of the approaches which are given in Section 4.2.2 in this chapter.

Obviously, the ideal estimates (1) can be expected to perform the boxt when observations of the latent responses are available. However, this is not always the case in practice. Therefore, it is our main interest in this subsection to examine the performance of the naive estimates and the corrected estimates under the three estimation accreaced.

First, we dock the performance of the OQL OOQL and ML estimats under the dock duration, where the latent response was assend to be directly durarelable. It is dura from Tables 4.2.4 ° that all of the three approaches yield ignorable binars on the estimaton of $\theta = (f_{12}, r_{12})^2$. The OOQL method series the product all out the more markue as the ML model does. The Silverer, provides a fixed does. The Silverer provides a fixed does. The Silverer provides are highly competition, for at our the decremes of the Silver to the compounding Silver, and the CPA's new concerned. Similar conclusions are randoubly by finite dimensional fixed and the Silver series of the decrement of the Silver to the compounding Silver, and the CPA's new concerned. Similar conclusions more randoubly by finite dimension of the silver at single series of the outer short of the conclusion. However, the preformance of the inducations is not of our much interest, due to the lumited modeling of the bilard statistics is not of our much interest, due to the lumited modeling of the bilard statistics is provided.

One do un mán algeirów si to compare the performance of the nuive estimates andre the three estimation approaches. Generally, all of the three approaches produce highly biased estimates through the three covariate designs, and the simution get areas when (x^+, x^-) take low values, i.e. $(x^+, x^-) = (0.75, 0.36)$. To be specific, the main estimates of $\theta = (x^-, y^-)$, when $(x^+, x^-) = (0.75, 0.36)$, are greedy indementional. In addito, the compared occurs probabilities are baiedoiry here

Now, we more on to examine the profermance of the currently of minimum bare be $O(Q_1, O(Q_1, and Q_1, and Q_1)$ expressions. From 7 with the 14.3 v, we can see that all of the three approaches produce approximativy indicated relationst, with overage probabilistics advances against net around the observed $O(Q_1, appath)$ behaves the $O(Q_1, approximation of Q_2)$ and M_1 and M_2 approaches to the log and near other $O(Q_1, appath)$ and M_2 are strong the number $S(W_1)$ model. The overavel of $O(Q_1, approximation M_2)$ models to be dominanted only the number $S(W_1)$ model is $O(Q_1, approximation M_2)$. The the three $O(Q_1, approximation M_2)$ models are the observable of $O(Q_1, approximation M_2)$ models are the observable of $O(Q_1, approximation M_2)$. The theorem is the origin of the $O(Q_1, approximation M_2)$ models are the observable of the O(Q_2, approximation M_2) and M_2 . The theorem is the output of the MLS Y_1 . This is the immunof matching the matching of the ma by the results in the case that $\gamma = 0$ when $(\gamma^+, \gamma^-) = (35, 530)$ under dough 21 MeV of Def (1.100 + 0.101 + 0.

At his the net back, we compare the ideal emisators with the entroproduct corrected emissions where the three emission emissions. It can be seen that held, of the ideal emission is a strategistic emission of the emission probabilities are used 155. However, the SEE's of the ideal emission appears to be significant to the emission of the SEE's of the issues upper number, because the ideal emissions are appeared to be the lost answer the three hilds of emissions. However, the corrected emissions are different to the ideal eminimation, However, the corrected emissions are different to the ideal emimath, the same particulation minimatification errors may drive hoppen in most of ensets.

Now, before we reach our final conclusion, there are a few important points to be mentioned.

1. The OGQL and ML approaches perform about balentially in most of cases as far as the mbinaschere, estimated standard errors, and enverage probabilities of the 65° confidence intervals are conversel. This is because the OGQL approach utilizes as much information as the ML approach does, that is, both approaches we the first and second order statistics of the responses in their estimating equilators. However, the similarity of theoretic development, there we approache are found to the similarity of theoretic development, there we approaches are found to the similarity of theoretic development, then we approaches are found to the similarity of the similar development, then we approaches are found to the similarity of the similar development. The similarity of the similarity development, then we approaches are found to the similarity of the similar development. The similarity approaches are found to the similarity of the similar development. The similarity of the similarity development, then we approaches are found to the similarity of the similar development. The similarity development of the similarity of the similar development, the similar development are found to the similarity of the similarity development. to perform as poor as each other in the naive estimation. This further explains that the GGL and ML approaches may be surpassed by the GQL method in some cases of the naive estimation because the GQL approach uses only the first order statistics of the error-room responses, hence these energy information is used in the estimation.

2. Be the ideal and noise estimates of the parameters $\delta = (p,\gamma)$, the estimates $\lambda_{\rm estarchical products (applical) between for the correct anisotane under the ML approach,$ the DM approach is polydown between for the correct anisotane of metrics of Ver(Fac).For the ideal and noise estimates consistently estimated by the innerse of the $varience inflational materians <math>f_{\rm eff}^{-1}$ and $f_{\rm eff}^{-1}$, modeling, the the correct of hear dutational three (DM) and the DM approximates the transposed from the transposed for simulation (1) (2).

In contrains, the simulation reach in Table 12.4.7 disc that the three proour distancian agreement of the COG, and ML sprearbox, see highly competitive in both simulation where the host requests are known, or the minimum distance of meganesis in correctly and the OQL approximation of the errors (SEN) are conversel. On the ther hand, market mean simulation, therein generation of the simulation of the thermal method matheful errors (SEN) are conversel. On the thermal matter of the simulation inclusion of the simulations in this section. It is obtained in the poolsy, with the OQL approach being slightly better than the other two approaches. Throughout the simulations in this section, the OQL and ML methods profess poolsy with the OQL approach being slightly better than the other two approaches. Throughout the simulations in this section, the OQL and ML methods profess poolsy with the OQL approach being slightly better than the other two approaches. However, considering the complexity of the DM algorithm and for the corrected ML methods, where we recommend the owner of the construct OQC method matter is a possible of the DM algorithm and for the corrected ML method.

| | | | GQL | | | OGQL | | | ML | |
|-----|---------------|--------|--------|--------|--------|--------|--------|--------|--------|--------|
| 2 | Quantity | | (2) | (3) | (1) | (2) | (3) | (1) | (2) | (3) |
| 1.5 | $SM(\beta_1)$ | | 0.909 | 1.007 | 1.032 | 0.983 | 0.992 | 1.031 | 0.983 | 0.995 |
| | SSE | 0.171 | 0.164 | 0.314 | 0.160 | 0.144 | 0.251 | 0.160 | 0.144 | 0.250 |
| | ESE | 0.171 | 0.165 | 0.310 | 0.165 | 0.142 | 0.244 | 0.165 | 0.142 | 0.242 |
| | CPr | 0.954 | 0.174 | 0.940 | 0.963 | 0.850 | 0.944 | 0.962 | 0.854 | 0.946 |
| | $SM(\beta_2)$ | 1.002 | 0.696 | 1.011 | 0.989 | 0.771 | 1.052 | 0.989 | 0.771 | 1.054 |
| | SSE | 0.464 | 0.349 | 0.702 | 0.427 | 0.323 | 0.732 | 0.427 | 0.323 | 0.631 |
| | ESE | 0.465 | 0.352 | 0.704 | 0.425 | 0.318 | 0.716 | 0.425 | 0.318 | 0.616 |
| | CPr | 0.952 | 0.492 | 0.950 | 0.968 | 0.807 | 0.948 | 0.958 | 0.805 | 0.945 |
| | $SM(\gamma)$ | 1.505 | 1.402 | 1.512 | 1.582 | 0.852 | 1.501 | 1.582 | 0.850 | 1.501 |
| | SSE | 0.385 | 0.277 | 0.515 | 0.236 | 0.181 | 0.450 | 0.235 | 0.180 | 0.450 |
| | ESE | 0.379 | 0.275 | 0.511 | 0.240 | 0.177 | 0.437 | 0.239 | 0.177 | 0.435 |
| | CPr | 0.936 | 0.890 | 0.954 | 0.954 | 0.071 | 0.947 | 0.955 | 0.070 | 0.947 |
| 0 | $SM(\beta_1)$ | 1.009 | 0.950 | 1.010 | 1.005 | 0.975 | 1.005 | 1.005 | 0.975 | 1.005 |
| | SSE | 0.136 | 0.126 | 0.164 | 0.130 | 0.125 | 0.161 | 0.130 | 0.125 | 0.161 |
| | ESE | 0.132 | 0.128 | 0.167 | 0.132 | 0.127 | 0.164 | 0.132 | 0.127 | 0.164 |
| | CPr | 0.956 | 0.952 | 0.948 | 0.957 | 0.945 | 0.952 | 0.956 | 0.945 | 0.952 |
| | $SM(\beta_2)$ | 0.989 | 0.727 | 0.986 | 1.007 | 0.760 | 1.015 | 1.007 | 0.760 | 1.015 |
| | SSE | 0.389 | 0.350 | 0.456 | 0.273 | 0.241 | 0.360 | 0.273 | 0.241 | 0.359 |
| | ESE | 0.375 | 0.344 | 0.479 | 0.267 | 0.242 | 0.350 | 0.267 | 0.242 | 0.352 |
| | CPr | 0.942 | 0.869 | 0.954 | 0.956 | 0.827 | 0.953 | 0.955 | 0.824 | 0.952 |
| | $SM(\gamma)$ | 0.008 | 0.0280 | 0.013 | -0.009 | -0.003 | -0.013 | -0.009 | -0.003 | -0.013 |
| | SSE | 0.285 | 0.275 | 0.355 | 0.153 | 0.137 | 0.213 | 0.153 | 0.137 | 0.213 |
| | ESE | 0.275 | 0.255 | 0.343 | 0.148 | 0.135 | 0.212 | 0.148 | 0.135 | 0.213 |
| | CPr | 0.954 | 0.951 | 0.951 | 0.943 | 0.948 | 0.954 | 0.943 | 0.947 | 0.952 |
| -1 | $SM(\beta_1)$ | 1.007 | 0.955 | 1.010 | 1.005 | 0.949 | 1.008 | 1.006 | 0.949 | 1.008 |
| | SSE | 0.121 | 0.121 | 0.151 | 0.120 | 0.118 | 0.150 | 0.120 | 0.118 | 0.150 |
| | ESE | 0.123 | 0.120 | 0.151 | 0.122 | 0.119 | 0.150 | 0.122 | 0.119 | 0.151 |
| | CPr | 0.954 | 0.926 | 0.960 | 0.953 | 0.925 | 0.957 | 0.953 | 0.925 | 0.956 |
| | $SM(\beta_2)$ | 0.994 | 0.838 | 1.000 | 1.000 | 0.706 | 1.008 | 1.000 | 0.706 | 1.007 |
| | SSE | 0.269 | 0.269 | 0.325 | 0.213 | 0.206 | 0.276 | 0.213 | 0.206 | 0.275 |
| | ESE | 0.268 | 0.264 | 0.322 | 0.213 | 0.204 | 0.272 | 0.213 | 0.203 | 0.273 |
| | CPr | 0.946 | 0.894 | 0.952 | 0.950 | 0.684 | 0.949 | 0.950 | 0.682 | 0.950 |
| | $SM(\gamma)$ | -1.002 | -0.861 | -1.009 | -1.009 | -0.717 | -1.017 | -1.009 | -0.717 | -1.017 |
| | SSE | 0.209 | 0.214 | 0.249 | 0.122 | 0.117 | 0.169 | 0.122 | 0.117 | 0.169 |
| | ESE | 0.209 | 0.213 | 0.245 | 0.120 | 0.114 | 0.168 | 0.120 | 0.114 | 0.166 |
| | CPr | 0.952 | 0.908 | 0.942 | 0.954 | 0.296 | 0.946 | 0.955 | 0.295 | 0.947 |

Table 4.2: Simulation results under Design 1 with $(\pi^+,\pi^-)=(0.95,0.90)$ and the true values of parameters $\beta=(1,1)$

| | | | GQL | | | OGQL | | | ML | |
|----------|---------------|--------|--------|--------|--------|--------|--------|--------|--------|--------|
| γ | Quantity | | (2) | (3) | (1) | (2) | (3) | (1) | (2) | (3) |
| 1.5 | $SM(\beta_1)$ | 1.005 | 0.885 | 1.008 | 0.998 | 0.797 | 0.999 | 0.998 | 0.797 | 0.999 |
| | SSE | 0.124 | 0.103 | 0.161 | 0.113 | 0.099 | 0.143 | 0.112 | 0.099 | 0.143 |
| | ESE | 0.126 | 0.107 | 0.160 | 0.117 | 0.104 | 0.147 | 0.117 | 0.104 | 0.148 |
| | CPr | 0.950 | 0.553 | 0.959 | 0.956 | 0.482 | 0.967 | 0.954 | 0.486 | 0.969 |
| | $SM(\beta_2)$ | 0.997 | 1.153 | 1.010 | 1.019 | 1.217 | 1.027 | 1.019 | 1.217 | 1.030 |
| | SSE | 0.446 | 0.367 | 0.570 | 0.234 | 0.207 | 0.323 | 0.234 | 0.207 | 0.323 |
| | ESE | 0.440 | 0.368 | 0.564 | 0.238 | 0.205 | 0.330 | 0.239 | 0.209 | 0.333 |
| | CPr | 0.947 | 0.920 | 0.945 | 0.959 | 0.821 | 0.960 | 0.958 | 0.831 | 0.960 |
| | $SM(\gamma)$ | 1.541 | 0.851 | 1.559 | 1.498 | 0.777 | 1.505 | 1.498 | 0.777 | 1.501 |
| | SSE | 0.483 | 0.352 | 0.643 | 0.184 | 0.150 | 0.300 | 0.184 | 0.150 | 0.299 |
| | ESE | 0.475 | 0.365 | 0.627 | 0.191 | 0.156 | 0.313 | 0.192 | 0.158 | 0.313 |
| | CPr | 0.945 | 0.540 | 0.947 | 0.958 | 0.003 | 0.961 | 0.960 | 0.003 | 0.964 |
| 0 | $SM(\beta_1)$ | 1.003 | 0.875 | 1.006 | 1.002 | 0.873 | 1.003 | 1.002 | 0.873 | 1.003 |
| | SSE | 0.093 | 0.093 | 0.116 | 0.093 | 0.093 | 0.116 | 0.093 | 0.093 | 0.116 |
| | ESE | 0.094 | 0.092 | 0.114 | 0.094 | 0.091 | 0.114 | 0.094 | 0.091 | 0.114 |
| | CPr | 0.961 | 0.711 | 0.948 | 0.957 | 0.702 | 0.942 | 0.957 | 0.703 | 0.947 |
| | $SM(\beta_2)$ | 0.981 | 0.958 | 0.984 | 0.996 | 0.939 | 1.001 | 0.995 | 0.939 | 1.001 |
| | SSE | 0.277 | 0.261 | 0.331 | 0.173 | 0.166 | 0.225 | 0.173 | 0.166 | 0.225 |
| | ESE | 0.263 | 0.256 | 0.320 | 0.166 | 0.159 | 0.217 | 0.167 | 0.160 | 0.218 |
| | CPr | 0.941 | 0.949 | 0.950 | 0.934 | 0.925 | 0.953 | 0.935 | 0.928 | 0.952 |
| | $SM(\gamma)$ | 0.024 | -0.028 | 0.025 | 0.006 | -0.008 | 0.003 | 0.005 | -0.008 | 0.004 |
| | SSE | 0.271 | 0.258 | 0.328 | 0.134 | 0.121 | 0.179 | 0.134 | 0.121 | 0.179 |
| | ESE | 0.290 | 0.257 | 0.315 | 0.130 | 0.123 | 0.180 | 0.130 | 0.123 | 0.181 |
| | CPr | 0.940 | 0.945 | 0.951 | 0.938 | 0.954 | 0.952 | 0.935 | 0.954 | 0.950 |
| -1 | $SM(\beta_1)$ | 1.002 | 0.879 | 1.001 | 1.002 | 0.862 | 1.001 | 1.002 | 0.852 | 1.001 |
| | SSE | 0.091 | 0.689 | 0.110 | 0.057 | 0.084 | 0.107 | 0.087 | 0.084 | 0.107 |
| | ESE | 0.093 | 0.090 | 0.113 | 0.090 | 0.087 | 0.110 | 0.090 | 0.087 | 0.111 |
| | CPr | 0.957 | 0.725 | 0.955 | 0.961 | 0.653 | 0.961 | 0.961 | 0.652 | 0.960 |
| | $SM(\beta_2)$ | 0.999 | 0.973 | 0.992 | 1.003 | 0.853 | 0.999 | 1.003 | 0.853 | 0.999 |
| | SSE | 0.192 | 0.199 | 0.231 | 0.135 | 0.133 | 0.172 | 0.134 | 0.133 | 0.173 |
| | ESE | 0.202 | 0.206 | 0.239 | 0.141 | 0.137 | 0.181 | 0.141 | 0.137 | 0.181 |
| | CPr | 0.962 | 0.964 | 0.961 | 0.959 | 0.808 | 0.957 | 0.958 | 0.809 | 0.957 |
| | $SM(\gamma)$ | -1.000 | -0.874 | -0.992 | -1.005 | -0.725 | -1.001 | -1.005 | -0.724 | -1.001 |
| | SSE | 0.212 | 0.213 | 0.248 | 0.109 | 0.105 | 0.147 | 0.109 | 0.105 | 0.147 |
| | ESE | 0.214 | 0.219 | 0.250 | 0.111 | 0.108 | 0.153 | 0.111 | 0.108 | 0.153 |
| | CPr | 0.953 | 0.938 | 0.952 | 0.958 | 0.292 | 0.962 | 0.959 | 0.292 | 0.964 |

Table 4.3: Simulation results under Design 2 with $(\pi^+,\pi^-)=(0.95,0.90)$ and the true values of parameters $\beta=(1,1)$

| | | | GOL | | | OGOL | | | ML | |
|----------|---------------|--------|--------|--------|--------|--------|--------|--------|--------|--------|
| γ | Quantity | (1) | (2) | (3) | (1) | (2) | (3) | (1) | (2) | (3) |
| 1.5 | $SM(\beta_1)$ | 1.004 | 0.807 | 1.004 | 1.003 | 0.847 | 1.003 | 1.003 | 0.847 | 1.003 |
| | SSE | 0.072 | 0.065 | 0.093 | 0.059 | 0.061 | 0.088 | 0.059 | 0.061 | 0.058 |
| | ESE | 0.071 | 0.065 | 0.090 | 0.059 | 0.062 | 0.085 | 0.059 | 0.062 | 0.087 |
| | CPr | 0.942 | 0.162 | 0.943 | 0.942 | 0.299 | 0.948 | 0.942 | 0.300 | 0.947 |
| | $SM(\beta_2)$ | 1.001 | 0.697 | 1.002 | 1.003 | 0.885 | 1.001 | 1.003 | 0.885 | 1.002 |
| | SSE | 0.169 | 0.153 | 0.205 | 0.143 | 0.119 | 0.178 | 0.143 | 0.119 | 0.178 |
| | ESE | 0.165 | 0.152 | 0.204 | 0.139 | 0.123 | 0.179 | 0.139 | 0.124 | 0.181 |
| | CPr | 0.945 | 0.492 | 0.947 | 0.935 | 0.855 | 0.946 | 0.937 | 0.858 | 0.943 |
| | $SM(\gamma)$ | 1.506 | 1.402 | 1.512 | 1.502 | 1.128 | 1.511 | 1.502 | 1.128 | 1.511 |
| | SSE | 0.184 | 0.175 | 0.214 | 0.123 | 0.107 | 0.167 | 0.123 | 0.107 | 0.166 |
| | ESE | 0.179 | 0.175 | 0.211 | 0.122 | 0.109 | 0.167 | 0.122 | 0.108 | 0.167 |
| | CPr | 0.943 | 0.901 | 0.952 | 0.951 | 0.074 | 0.951 | 0.953 | 0.073 | 0.943 |
| 0 | $SM(\beta_1)$ | 1.005 | 0.868 | 1.005 | 1.004 | 0.886 | 1.005 | 1.004 | 0.886 | 1.005 |
| | SSE | 0.068 | 0.062 | 0.079 | 0.051 | 0.056 | 0.073 | 0.061 | 0.056 | 0.073 |
| | ESE | 0.066 | 0.063 | 0.080 | 0.050 | 0.057 | 0.074 | 0.050 | 0.057 | 0.074 |
| | CPr | 0.937 | 0.454 | 0.948 | 0.946 | 0.474 | 0.950 | 0.946 | 0.474 | 0.954 |
| | $SM(\beta_2)$ | 0.999 | 0.817 | 0.999 | 0.999 | 0.870 | 1.001 | 0.999 | 0.870 | 1.001 |
| | SSE | 0.131 | 0.125 | 0.154 | 0.108 | 0.103 | 0.134 | 0.108 | 0.103 | 0.134 |
| | ESE | 0.128 | 0.126 | 0.154 | 0.106 | 0.102 | 0.133 | 0.106 | 0.102 | 0.133 |
| | CPr | 0.938 | 0.680 | 0.951 | 0.941 | 0.735 | 0.956 | 0.941 | 0.735 | 0.956 |
| | $SM(\gamma)$ | -0.001 | 0.132 | 0.001 | -0.001 | -0.056 | -0.002 | -0.001 | -0.056 | -0.002 |
| | SSE | 0.145 | 0.143 | 0.169 | 0.095 | 0.093 | 0.123 | 0.095 | 0.093 | 0.123 |
| | ESE | 0.142 | 0.143 | 0.169 | 0.095 | 0.093 | 0.124 | 0.095 | 0.093 | 0.124 |
| | CPr | 0.949 | 0.837 | 0.954 | 0.954 | 0.908 | 0.961 | 0.954 | 0.904 | 0.958 |
| -1 | β_1 | 1.003 | 0.854 | 1.004 | 1.002 | 0.848 | 1.004 | 1.002 | 0.848 | 1.004 |
| | SSE | 0.075 | 0.070 | 0.091 | 0.063 | 0.059 | 0.080 | 0.053 | 0.059 | 0.050 |
| | ESE | 0.074 | 0.068 | 0.091 | 0.060 | 0.057 | 0.079 | 0.050 | 0.057 | 0.079 |
| | CPr | 0.942 | 0.432 | 0.955 | 0.942 | 0.245 | 0.944 | 0.942 | 0.245 | 0.946 |
| | β_2 | 1.003 | 0.854 | 1.003 | 1.003 | 0.844 | 1.004 | 1.003 | 0.844 | 1.004 |
| | SSE | 0.120 | 0.119 | 0.145 | 0.097 | 0.093 | 0.121 | 0.097 | 0.093 | 0.121 |
| | ESE | 0.122 | 0.119 | 0.147 | 0.095 | 0.093 | 0.122 | 0.095 | 0.093 | 0.123 |
| | CPr | 0.956 | 0.754 | 0.954 | 0.944 | 0.613 | 0.953 | 0.945 | 0.940 | 0.952 |
| | 9 | -1.010 | -0.708 | -1.011 | -1.009 | -0.690 | -1.012 | -1.009 | -0.690 | -1.012 |
| | SSE | 0.157 | 0.157 | 0.191 | 0.009 | 0.095 | 0.135 | 0.099 | 0.095 | 0.135 |
| | ESE | 0.161 | 0.155 | 0.195 | 0.097 | 0.093 | 0.135 | 0.097 | 0.002 | 0.135 |
| | CPr | 0.961 | 0.519 | 0.956 | 0.941 | 0.098 | 0.952 | 0.942 | 0.094 | 0.955 |

Table 4.4: Simulation results under Design 3 with $(\pi^+,\pi^-)=(0.95,0.90)$ and the true values of parameters $\beta=(1,1)$

| | | | GQL | | | OGQL | | | ML | |
|-----|---------------|--------|--------|--------|--------|--------|--------|--------|--------|--------|
| 2 | Quantity | (1) | (2) | (3) | (1) | (2) | (3) | | (2) | (3) |
| 1.5 | $SM(\beta_1)$ | 0.990 | 0.327 | 1.018 | 0.999 | 0.381 | 1.017 | 0.999 | 0.381 | 1.013 |
| | SSE | 0.059 | 0.053 | 0.171 | 0.135 | 0.049 | 0.166 | 0.066 | 0.049 | 0.161 |
| | ESE | 0.071 | 0.055 | 0.167 | 0.139 | 0.051 | 0.161 | 0.069 | 0.051 | 0.160 |
| | CPr | 0.949 | 0.000 | 0.950 | 0.958 | 0.000 | 0.953 | 0.958 | 0.000 | 0.958 |
| | $SM(\beta_2)$ | 0.990 | 0.293 | 1.012 | 0.995 | 0.494 | 1.010 | 0.995 | 0.494 | 1.005 |
| | SSE | 0.164 | 0.116 | 0.367 | 0.135 | 0.058 | 0.347 | 0.135 | 0.058 | 0.336 |
| | ESE | 0.165 | 0.121 | 0.367 | 0.139 | 0.091 | 0.347 | 0.139 | 0.092 | 0.347 |
| | CPr | 0.956 | 0.000 | 0.957 | 0.962 | 0.000 | 0.955 | 0.962 | 0.000 | 0.962 |
| | $SM(\gamma)$ | 1.520 | 0.686 | 1.533 | 1.512 | 0.351 | 1.536 | 1.512 | 0.361 | 1.530 |
| | SSE | 0.184 | 0.148 | 0.335 | 0.124 | 0.084 | 0.316 | 0.124 | 0.084 | 0.307 |
| | ESE | 0.180 | 0.159 | 0.357 | 0.122 | 0.089 | 0.328 | 0.122 | 0.092 | 0.326 |
| | CPr | 0.951 | 0.000 | 0.959 | 0.953 | 0.000 | 0.958 | 0.952 | 0.000 | 0.966 |
| 0 | $SM(\beta_i)$ | 1.006 | 0.451 | 1.024 | 1.003 | 0.434 | 1.021 | 1.003 | 0.434 | 1.019 |
| | SSE | 0.053 | 0.056 | 0.136 | 0.056 | 0.049 | 0.1306 | 0.056 | 0.049 | 0.130 |
| | ESE | 0.056 | 0.058 | 0.139 | 0.060 | 0.051 | 0.133 | 0.060 | 0.051 | 0.133 |
| | CPr | 0.952 | 0.000 | 0.962 | 0.964 | 0.000 | 0.963 | 0.964 | 0.000 | 0.962 |
| | $SM(\beta_2)$ | 1.006 | 0.427 | 1.013 | 1.001 | 0.381 | 1.009 | 1.001 | 0.381 | 1.001 |
| | SSE | 0.127 | 0.112 | 0.262 | 0.103 | 0.085 | 0.244 | 0.103 | 0.085 | 0.243 |
| | ESE | 0.128 | 0.113 | 0.263 | 0.105 | 0.087 | 0.245 | 0.105 | 0.087 | 0.246 |
| | CPr | 0.953 | 0.001 | 0.960 | 0.953 | 0.000 | 0.963 | 0.953 | 0.000 | 0.966 |
| | $SM(\gamma)$ | -0.009 | -0.117 | -0.016 | -0.002 | -0.036 | -0.010 | +0.002 | -0.036 | -0.007 |
| | SSE | 0.143 | 0.155 | 0.278 | 0.095 | 0.089 | 0.246 | 0.005 | 0.059 | 0.244 |
| | ESE | 0.142 | 0.154 | 0.275 | 0.095 | 0.086 | 0.240 | 0.095 | 0.056 | 0.241 |
| | CPr | 0.951 | 0.875 | 0.955 | 0.950 | 0.927 | 0.947 | 0.951 | 0.929 | 0.952 |
| -1 | $SM(\beta_1)$ | 0.998 | 0.402 | 1.000 | 0.997 | 0.399 | 1.000 | 0.997 | 0.399 | 0.997 |
| | SSE | 0.129 | 0.112 | 0.273 | 0.129 | 0.107 | 0.270 | 0.129 | 0.107 | 0.268 |
| | ESE | 0.123 | 0.107 | 0.272 | 0.122 | 0.106 | 0.269 | 0.122 | 0.106 | 0.269 |
| | CPr | 0.936 | 0.000 | 0.953 | 0.939 | 0.000 | 0.948 | 0.938 | 0.000 | 0.950 |
| | $SM(\beta_2)$ | 1.004 | 0.471 | 1.017 | 1.012 | 0.220 | 1.021 | 1.012 | 0.220 | 1.005 |
| | SSE | 0.272 | 0.251 | 0.558 | 0.216 | 0.171 | 0.514 | 0.216 | 0.171 | 0.505 |
| | ESE | 0.2687 | 0.244 | 0.557 | 0.213 | 0.170 | 0.514 | 0.213 | 0.170 | 0.515 |
| | CPr | 0.948 | 0.436 | 0.947 | 0.949 | 0.006 | 0.947 | 0.949 | 0.006 | 0.947 |
| | $SM(\gamma)$ | -0.997 | -0.579 | -1.005 | -1.006 | -0.237 | -1.011 | -1.006 | -0.237 | -0.996 |
| | SSE | 0.214 | 0.246 | 0.407 | 0.121 | 0.094 | 0.337 | 0.121 | 0.094 | 0.329 |
| | ESE | 0.210 | 0.248 | 0.403 | 0.120 | 0.095 | 0.342 | 0.120 | 0.095 | 0.344 |
| | CPr | 0.942 | 0.619 | 0.956 | 0.958 | 0.000 | 0.960 | 0.959 | 0.000 | 0.965 |

Table 4.5: Simulation results under Design 1 with $(\pi^+, \pi^-) = (0.75, 0.80)$ and the true values of parameters $\beta = (1, 1)$

| | | | GQL | | - | OGQL | | | ML | |
|----------|---------------|--------|--------|--------|--------|--------|--------|--------|--------|--------|
| γ | Quantity | | (2) | (3) | (1) | (2) | (3) | | (2) | (3) |
| 1.5 | $SM(\beta_1)$ | 1.005 | 0.804 | 1.008 | 0.996 | 0.794 | 1.004 | 0.996 | 0.795 | 1.005 |
| | SSE | 0.122 | 0.100 | 0.161 | 0.112 | 0.095 | 0.130 | 0.112 | 0.096 | 0.130 |
| | ESE | 0.127 | 0.108 | 0.160 | 0.116 | 0.104 | 0.132 | 0.117 | 0.104 | 0.131 |
| | CPr | 0.958 | 0.955 | 0.960 | 0.962 | 0478 | 0.954 | 0.962 | 0.472 | 0.952 |
| | $SM(\beta_2)$ | 1.003 | 1.1613 | 1.009 | 1.037 | 1.231 | 1.011 | 1.037 | 1.230 | 1.003 |
| | SSE | 0.439 | 0.357 | 0.571 | 0.227 | 0.119 | 0.342 | 0.227 | 0.200 | 0.331 |
| | ESE | 0.431 | 0.369 | 0.535 | 0.240 | 0.215 | 0.339 | 0.240 | 0.210 | 0.331 |
| | CPr | 0.944 | 0.914 | 0.948 | 0.958 | 0.826 | 0.959 | 0.958 | 0.826 | 0.962 |
| | $SM(\gamma)$ | 1.550 | 0.852 | 1.579 | 1.501 | 0.776 | 1.534 | 1.492 | 0.774 | 1.540 |
| | SSE | 0.492 | 0.354 | 0.661 | 0.181 | 0.149 | 0.434 | 0.180 | 0.149 | 0.432 |
| | ESE | 0.457 | 0.365 | 0.591 | 0.192 | 0.158 | 0.435 | 0.192 | 0.158 | 0.431 |
| | CPr | 0.942 | 0.546 | 0.946 | 0.956 | 0.002 | 0.959 | 0.946 | 0.002 | 0.953 |
| 0 | $SM(\beta_1)$ | 0.998 | 0.424 | 1.020 | 0.997 | 0.423 | 1.014 | 0.997 | 0.423 | 1.012 |
| | SSE | 0.094 | 0.080 | 0.197 | 0.093 | 0.080 | 0.197 | 0.053 | 0.080 | 0.195 |
| | ESE | 0.094 | 0.082 | 0.202 | 0.094 | 0.082 | 0.200 | 0.094 | 0.082 | 0.201 |
| | CPr | 0.944 | 0.000 | 0.955 | 0.953 | 0.000 | 0.954 | 0.953 | 0.000 | 0.956 |
| | $SM(\beta_2)$ | 0.988 | 0.343 | 0.971 | 1.006 | 0.371 | 1.017 | 1.006 | 0.371 | 0.995 |
| | SSE | 0.272 | 0.232 | 0.592 | 0.168 | 0.123 | 0.465 | 0.168 | 0.123 | 0.442 |
| | ESE | 0.253 | 0.225 | 0.565 | 0.167 | 0.127 | 0.456 | 0.167 | 0.127 | 0.458 |
| | CPr | 0.944 | 0.175 | 0.953 | 0.947 | 0.002 | 0.950 | 0.948 | 0.002 | 0.962 |
| | $SM(\gamma)$ | 0.020 | 0.039 | 0.047 | -0.001 | 0.000 | -0.007 | -0.001 | 0.000 | 0.014 |
| | SSE | 0.272 | 0.285 | 0.604 | 0.132 | 0.097 | 0.437 | 0.132 | 0.097 | 0.411 |
| | ESE | 0.259 | 0.273 | 0.562 | 0.132 | 0.099 | 0.422 | 0.130 | 0.009 | 0.424 |
| | CPr | 0.944 | 0.948 | 0.951 | 0.944 | 0.957 | 0.949 | 0.944 | 0.957 | 0.958 |
| -1 | $SM(\beta_1)$ | 1.004 | 0.442 | 1.007 | 1.003 | 0.425 | 1.005 | 1.003 | 0.425 | 1.002 |
| | SSE | 0.092 | 0.085 | 0.196 | 0.089 | 0.080 | 0.190 | 0.059 | 0.080 | 0.189 |
| | ESE | 0.093 | 0.083 | 0.197 | 0.090 | 0.081 | 0.193 | 0.090 | 0.081 | 0.194 |
| | CPr | 0.953 | 0.000 | 0.955 | 0.968 | 0.000 | 0.955 | 0.958 | 0.000 | 0.959 |
| | $SM(\beta_2)$ | 0.992 | 0.406 | 0.999 | 1.000 | 0.202 | 1.014 | 1.000 | 0.202 | 1.001 |
| | SSE | 0.203 | 0.198 | 0.402 | 0.138 | 0.119 | 0.341 | 0.138 | 0.119 | 0.335 |
| | ESE | 0.202 | 0.197 | 0.393 | 0.141 | 0.120 | 0.343 | 0.141 | 0.119 | 0.344 |
| | CPr | 0.949 | 0.160 | 0.955 | 0.952 | 0.000 | 0.952 | 0.954 | 0.000 | 0.951 |
| | $SM(\beta_2)$ | -0.990 | -0.573 | -0.994 | -1.001 | -0.262 | -1.012 | -1.001 | -0.262 | -0.997 |
| | SSE | 0.218 | 0.255 | 0.423 | 0.109 | 0.095 | 0.313 | 0.109 | 0.095 | 0.300 |
| | ESE | 0.214 | 0.251 | 0.406 | 0.111 | 0.094 | 0.319 | 0.111 | 0.094 | 0.320 |
| | CPr | 0.947 | 0.618 | 0.944 | 0.568 | 0.000 | 0.961 | 0.939 | 0.000 | 0.963 |

Table 4.6: Simulation results under Design 2 with $(\pi^+, \pi^-) = (0.75, 0.80)$ and the true values of parameters $\beta = (1, 1)$

| | | | GQL | | | OGQL | | | ML | |
|----------|----------------|--------|--------|--------|--------|--------|--------|--------|--------|--------|
| γ | Quantity | (1) | (2) | (3) | (1) | (2) | (3) | | (2) | (3) |
| 1.5 | $SM(\beta_i)$ | 0.999 | 0.327 | 1.018 | 0.999 | 0.381 | 1.017 | 0.999 | 0.381 | 1.013 |
| | SSE | 0.059 | 0.053 | 0.171 | 0.056 | 0.049 | 0.165 | 0.056 | 0.049 | 0.161 |
| | ESE | 0.071 | 0.055 | 0.167 | 0.069 | 0.051 | 0.160 | 0.059 | 0.051 | 0.160 |
| | CPr | 0.949 | 0.000 | 0.950 | 0.958 | 0.000 | 0.953 | 0.958 | 0.000 | 0.958 |
| | $SM(\beta_2)$ | 0.990 | 0.293 | 1.012 | 0.996 | 0.494 | 1.010 | 0.996 | 0.494 | 1.005 |
| | SSE | 0.164 | 0.116 | 0.367 | 0.135 | 0.088 | 0.347 | 0.135 | 0.088 | 0.336 |
| | ESE | 0.165 | 0.121 | 0.367 | 0.139 | 0.091 | 0.347 | 0.139 | 0.092 | 0.347 |
| | CPr | 0.956 | 0.000 | 0.957 | 0.962 | 0.000 | 0.955 | 0.962 | 0.000 | 0.952 |
| | $SM(\gamma)$ | 1.520 | 0.686 | 1.533 | 1.512 | 0.361 | 1.536 | 1.512 | 0.361 | 1.530 |
| | SSE | 0.184 | 0.148 | 0.335 | 0.124 | 0.084 | 0.316 | 0.124 | 0.084 | 0.307 |
| | ESE | 0.180 | 0.159 | 0.347 | 0.122 | 0.089 | 0.328 | 0.122 | 0.088 | 0.326 |
| | CPr | 0.951 | 0.000 | 0.959 | 0.953 | 0.000 | 0.958 | 0.952 | 0.000 | 0.956 |
| 0 | $SM(\beta_1)$ | 1.006 | 0.451 | 1.024 | 1.003 | 0.434 | 1.021 | 1.003 | 0.434 | 1.019 |
| | SSE | 0.053 | 0.056 | 0.136 | 0.056 | 0.049 | 0.131 | 0.056 | 0.049 | 0.130 |
| | ESE | 0.056 | 0.058 | 0.139 | 0.060 | 0.051 | 0.133 | 0.050 | 0.051 | 0.133 |
| | CPr | 0.952 | 0.000 | 0.962 | 0.964 | 0.000 | 0.963 | 0.954 | 0.000 | 0.962 |
| | $SM(\beta_2)$ | 1.006 | 0.427 | 1.013 | 1.001 | 0.381 | 1.009 | 1.001 | 0.381 | 1.006 |
| | SSE | 0.127 | 0.112 | 0.262 | 0.103 | 0.085 | 0.244 | 0.103 | 0.085 | 0.243 |
| | ESE | 0.128 | 0.113 | 0.263 | 0.106 | 0.087 | 0.245 | 0.106 | 0.087 | 0.246 |
| | CPr | 0.953 | 0.001 | 0.960 | 0.953 | 0.000 | 0.963 | 0.953 | 0.000 | 0.956 |
| | $SM(\gamma)$ | -0.009 | -0.117 | -0.016 | -0.002 | -0.036 | -0.010 | -0.002 | -0.036 | -0.007 |
| | SSE | 0.143 | 0.155 | 0.278 | 0.005 | 0.089 | 0.246 | 0.095 | 0.089 | 0.244 |
| | ESE | 0.142 | 0.154 | 0.275 | 0.005 | 0.085 | 0.240 | 0.095 | 0.085 | 0.241 |
| | CPr | 0.951 | 0.875 | 0.955 | 0.95 | 0.927 | 0.947 | 0.951 | 0.929 | 0.952 |
| -1 | A ₁ | 1.001 | 0.457 | 1.013 | 0.998 | 0.379 | 1.009 | 0.998 | 0.379 | 1.005 |
| | SSE | 0.074 | 0.065 | 0.161 | 0.059 | 0.052 | 0.148 | 0.059 | 0.052 | 0.146 |
| | ESE | 0.074 | 0.062 | 0.155 | 0.060 | 0.050 | 0.145 | 0.050 | 0.051 | 0.149 |
| | CPr | 0.948 | 0.000 | 0.939 | 0.961 | 0.000 | 0.955 | 0.961 | 0.000 | 0.954 |
| | Â ₂ | 1.001 | 0.487 | 1.016 | 0.999 | 0.312 | 1.013 | 0.999 | 0.312 | 1.008 |
| | SSE | 0.124 | 0.117 | 0.251 | 0.096 | 0.083 | 0.225 | 0.096 | 0.083 | 0.223 |
| | ESE | 0.122 | 0.112 | 0.245 | 0.095 | 0.083 | 0.222 | 0.095 | 0.083 | 0.223 |
| | CPr | 0.950 | 0.008 | 0.949 | 0.943 | 0.000 | 0.951 | 0.943 | 0.000 | 0.954 |
| | Ŷ | -1.002 | -0.706 | -1.025 | -0.999 | -0.357 | -1.020 | -0.999 | -0.357 | -1.011 |
| | SSE | 0.163 | 0.172 | 0.321 | 0.096 | 0.083 | 0.261 | 0.096 | 0.083 | 0.255 |
| | ESE | 0.161 | 0.170 | 0.318 | 0.097 | 0.085 | 0.264 | 0.097 | 0.086 | 0.264 |
| | CPr | 0.950 | 0.563 | 0.947 | 0.944 | 0.000 | 0.955 | 0.945 | 0.000 | 0.957 |

Table 4.7: Simulation results under Design 3 with $(\pi^+, \pi^-) = (0.75, 0.80)$ and the true values of parameters $\beta = (1, 1)$

4.2.3.3 Insight to robustness: a continued simulation study

Recall that in the previous subserving, we choole the preferences of the field matrixes, naive estimation, and enserving estimation of $\sigma = (\sigma, \gamma)$ where the GQL, OGQL, and ML approaches, and found that the currented estimation with the previous abstration best. In this advance, in some starting to the transmission of the previous angle assumed that the sensitivity τ^* and predicity τ^* on terms. This sumhyperstands that the same starting the starting of the start that estimates of these two parameters, τ^* and τ^* , may be derived based on a main independent with datas in independent in previous. Some archites suggest that estimates of these two parameters, τ^* and τ^* , may be derived based in a main independent with datas in subservation. Some as Theorem is the start main interval in this isolated to be interlayed in the advances of the current of estimation or intrinsive that eligibly lowed estimates of τ^* is any τ^* may are subserved.

For the regression promoter β and the dynamic dependence parameter γ , we down $\beta = (1, (\gamma, \alpha)$ on the volume for γ and β . After a due true works over $\alpha < \alpha$ and π^{-1} are recovered, we choose the same pairs of volume und in the previous subsection, which are $(r^{+}, \tau^{-}) = (0.55, 0.03)$ and $(r^{+}, \tau^{-}) = (0.55, 0.03)$. The bar specific, in the ord $(r^{+}, \tau^{-}) = (0.55, 0.03)$, (1004), (1004) and (1004) $(0.07, \tau^{-})$ are used when $\gamma = 1 \approx 0$, they are (1056, 0.031), (1004, 0.03) and (1004), (1004), (1004) and $(r^{+}, \tau^{-}) = (170, 0.00)$, there pairs of handon stations of (r^{+}, τ^{-}) are used for halo are shown here $\gamma = 1 \propto 0.04$ are (1004, 0.03). Similarly, for the true volume $(r^{+}, \tau^{-}) = (170, 0.00)$, there pairs of handon stations of (r^{+}, τ^{-}) are used for halo are shown here $\gamma = 1 \propto 0.04$ are (1004, 0.012). The source pairs (0.765, 0.815) and (0.735, 0.785) are used when $\gamma = 0.500$ simulations are conducted in the case of $(\pi^+, \pi^-) = (0.75, 0.89)$. The simulation results are reported in Table 4.8 for $(\pi^+, \pi^-) = (0.55, 0.90)$, and Table 4.9 for $(\pi^+, \pi^-) = (0.75, 0.80)$, where in both black, covariate density afters in Subsection 4.2.3 is used.

In one sum finur Table 4.8 and Table 6.9 dut the OGQL segments ratio may be perform simility to WM. Segments in most of ones. Therefore in the billowing part of this perpends, beying in mind that the OGQL segments by the OGQL segments, beying in mind that the OGQL segments by the the OGQL segments. It is not in the the OGQL segments by the ML segments at the same time. It is in due to the simulation results that all three spensors have segments in the other than the other of the other and confidence intervals for same parameters may one perform well when the line of π at π^{-1} of the same, the results of the other of π^{-1} of π^{-1} (3.0 due in the thermal segments in the theorem that π^{-1} (1.0 due), π^{-1} (3.0 due

Furthermore, the simulation much in Tables 4.8 and 4.9 show that the OOQL approach trends to performs better than the OQL approach as far as the simulated mathematic set of the sourcement. For example, when $(\pi^+,\pi^-) = (0.50, 0.00), \tau =$ $1 (Table 4.8), and the "wavking" values <math>(\pi^+,\pi^-) = (0.51, 0.00)$ are used, the computed SEVs of $\partial_{0.000}$, and $\partial_{0.0000}$, which are given to (0.60, 0.124, cm). Whereas under respectively. We have given by (0.000, 0.214, cm). Whereas under both attached, the ISE's of δ are quite close to the respective SSTs. However, the biases of the OOQL estimates of θ may be greater than the OQL estimates the excitat the "weakled" sensitivity and specified have much biases. For example, $\delta_{OQQ} = 0.844$ more biased from the true value 1 than $\delta_{OQL} = 0.883$. This is because the OOQL approach may use more error information than OQL approach dow when the much biased sensitivity and sensitive weak.

Another interacting finding is that when the working volume $(\pi^+, \tau^-) = ort (d) d)$, scattering the corresponding simulation member, respectively, the BSSE and CPAY, show to be letter as compound with the initiation where the winking volumes are digitally underestimated. For instance, for $(\pi^+, \tau^-) = (0.75, 0.80)_{-1} = 0$ (Table 42, then $R_{\rm D}(3.64)$ are used and working values of $(\tau^+, \tau^-) = (0.75, 0.80)_{-1} = 0$ (Table 42, then $R_{\rm D}(3.64)_{-1}$ are used in working values of $(\tau^+, \tau^-) = (0.72, 0.72)_{-1}$ are given by (12).03, 0.51, 0.320), when the underestimated value $(\tau^+, \tau^-) = (0.72, 0.72)_{-1}$ are used. The same finding folds for the OG2 k ML approximate.

In summary, based on the simulation random in Table 6.4 and 5.4, we recommodto our of diaglity second measuring via a simulation of the same statisticturity robust estimation random. Also, we suggest the use of the OOQL approach, since it produces top biases on the estimation of model parameters and multity RMTs the OOQ and the OOQL approach is the simulation of the same statistical term of the SMT statistical sectors are statistical to the SMT statistical sectors and multity RMTs. The SMT statistical term of the SMT statistical sectors and multity RMTs areas. In addition, then, OOQL approach is due is no supersub-theorem with the ML approach who comparisation is corrected.

| | | | | CQL | | | OCQL | | | ML | |
|---|------------------|----------|-----------|-----------|--------|-----------|-----------|--------|-----------|-----------|----------|
| 7 | (π^+, π^-) | Quantity | β_1 | β_2 | 7 | β_1 | β_2 | γ | β_1 | β_2 | γ |
| Т | (0.95, 0.90) | SM | -1.003 | 1.013 | 1.004 | -1.001 | 1.008 | 1.001 | -1.001 | 1.005 | 1.001 |
| | | SSE | 0.085 | 0.183 | 0.394 | 0.076 | 0.119 | 0.160 | 0.076 | 0.119 | 0.159 |
| | | ESE | 0.083 | 0.186 | 0.389 | 0.075 | 0.123 | 0.161 | 0.075 | 0.124 | 0.161 |
| | | CPr | 0.938 | 0.950 | 0.960 | 0.956 | 0.960 | 0.950 | 0.950 | 0.960 | 0.942 |
| | (0.96, 0.91) | SM | -0.968 | 0.985 | 0.948 | -0.966 | 0.983 | 0.941 | -0.966 | 0.982 | 0.942 |
| | | SSE | 0.081 | 0.175 | 0.370 | 0.073 | 0.114 | 0.149 | 0.073 | 0.114 | 0.149 |
| | | ESE | 0.079 | 0.178 | 0.364 | 0.071 | 0.118 | 0.150 | 0.071 | 0.119 | 0.150 |
| | | CPr | 0.936 | 0.952 | 0.948 | 0.938 | 0.952 | 0.930 | 0.937 | 0.948 | 0.930 |
| | (0.94, 0.89) | SM | -1.040 | 1.043 | 1.062 | -1.038 | 1.032 | 1.065 | -1.038 | 1.032 | 1.065 |
| | | SSE | 0.089 | 0.191 | 0.422 | 0.080 | 0.124 | 0.171 | 0.080 | 0.124 | 0.171 |
| | | ESE | 0.088 | 0.194 | 0.417 | 0.078 | 0.129 | 0.173 | 0.079 | 0.129 | 0.173 |
| | | CPr | 0.930 | 0.942 | 0.964 | 0.930 | 0.950 | 0.952 | 0.930 | 0.960 | 0.954 |
| | (0.97, 0.92) | SM | -0.934 | 0.960 | 0.895 | -0.933 | 0.959 | 0.886 | -0.933 | 0.959 | 0.886 |
| | | SSE | 0.077 | 0.168 | 0.348 | 0.070 | 0.109 | 0.140 | 0.070 | 0.109 | 0.139 |
| | | ESE | 0.075 | 0.171 | 0.342 | 0.068 | 0.114 | 0.141 | 0.068 | 0.114 | 0.141 |
| | | CPr | 0.848 | 0.954 | 0.934 | 0.804 | 0.946 | 0.860 | 0.808 | 0.948 | 0.864 |
| 0 | (0.95, 0.90) | SM | -1.008 | 1.007 | -0.007 | -1.005 | 1.005 | -0.007 | -1.005 | 1.005 | -0.007 |
| | | SSE | 0.070 | 0.180 | 0.327 | 0.070 | 0.122 | 0.156 | 0.070 | 0.122 | 0.156 |
| | | ESE | 0.072 | 0.182 | 0.328 | 0.071 | 0.123 | 0.151 | 0.071 | 0.124 | 0.152 |
| | | CPr | 0.950 | 0.944 | 0.950 | 0.950 | 0.944 | 0.950 | 0.950 | 0.948 | 0.950 |
| | (0.96, 0.91) | SM | -0.976 | 0.974 | -0.009 | -0.974 | 0.971 | -0.007 | -0.974 | 0.971 | -0.007 |
| | | SSE | 0.057 | 0.173 | 0.314 | 0.067 | 0.116 | 0.147 | 0.057 | 0.116 | 0.147 |
| | | ESE | 0.059 | 0.175 | 0.315 | 0.068 | 0.118 | 0.143 | 0.058 | 0.119 | 0.143 |
| | | CPr | 0.932 | 0.948 | 0.950 | 0.928 | 0.538 | 0.948 | 0.930 | 0.938 | 0.950 |
| | (0.94, 0.89) | SM | -1.042 | 1.043 | -0.005 | -1.039 | 1.041 | -0.007 | -1.039 | 1.041 | -0.007 |
| | | SSE | 0.074 | 0.188 | 0.341 | 0.073 | 0.128 | 0.165 | 0.073 | 0.1283 | 0.165 |
| | | ESE | 0.076 | 0.190 | 0.343 | 0.075 | 0.130 | 0.160 | 0.075 | 0.131 | 0.161 |
| | | CPr | 0.932 | 0.950 | 0.946 | 0.934 | 0.938 | 0.952 | 0.940 | 0.936 | 0.946 |
| | (0.965, 0.915) | SM | -0.961 | 0.959 | -0.010 | -0.959 | 0.955 | -0.007 | -0.959 | 0.955 | -0.007 |
| | | SSE | 0.066 | 0.170 | 0.307 | 0.055 | 0.114 | 0.143 | 0.055 | 0.114 | 0.143 |
| | | ESE | 0.068 | 0.172 | 0.308 | 0.056 | 0.115 | 0.139 | 0.057 | 0.116 | 0.139 |
| | | CPr | 0.918 | 0.944 | 0.950 | 0.902 | 0.922 | 0.948 | 0.902 | 0.918 | 0.950 |
| | (0.935, 0.885) | SM | -1.050 | 1.053 | -0.005 | -1.056 | 1.061 | -0.007 | -1.056 | 1.051 | -0.007 |
| | | SSE | 0.076 | 0.192 | 0.349 | 0.075 | 0.132 | 0.170 | 0.075 | 0.132 | 0.170 |
| | | ESE | 0.078 | 0.194 | 0.351 | 0.076 | 0.133 | 0.165 | 0.077 | 0.134 | 0.166 |
| | | CPr | 0.906 | 0.952 | 0.948 | 0.906 | 0.935 | 0.952 | 0.906 | 0.934 | 0.944 |
| | (0.97, 0.92) | SM | -0.947 | 0.943 | -0.011 | -0.945 | 0.939 | -0.007 | -0.945 | 0.939 | -0.007 |
| | | SSE | 0.065 | 0.166 | 0.301 | 0.064 | 0.111 | 0.139 | 0.064 | 0.111 | 0.139 |
| | | ESE | 0.065 | 0.169 | 0.302 | 0.055 | 0.113 | 0.135 | 0.065 | 0.113 | 0.135 |
| | | CPr | 0.872 | 0.932 | 0.950 | 0.868 | 0.908 | 0.950 | 0.868 | 0.906 | 0.950 |

Table 4.8: Robustness about estimated (π^+, π^-) based on 500 simulations under Design 3 with true values $(\pi^+, \pi^-) = (0.95, 0.90), \beta = (-1, 1), \gamma = 1, 0$

| | | | | GOL | | | OGOL | | | ML | |
|----------|------------------|----------|--------|-------|--------|-----------|-----------|-------|-----------|-----------|-------|
| γ | (π^+, π^-) | Quantity | B1 | 32 | 2 | β_1 | β_2 | 2 | β_1 | β_2 | γ |
| Ť | (0.75.0.80) | SM | | | 1.058 | -1.007 | | 0.996 | -1.002 | 1.035 | 0.993 |
| | | SSE | 0.137 | 0.309 | 0.672 | 0.121 | 0.216 | 0.352 | 0.119 | 0.213 | 0.358 |
| | | ESE | 0.138 | 0.300 | 0.677 | 0.124 | 0.218 | 0.356 | 0.124 | 0.219 | 0.366 |
| | | CPr | 0.964 | 0.950 | 0.956 | 0.952 | 0.952 | 0.950 | 0.960 | 0.952 | 0.966 |
| | (0.76.0.81) | SM | -0.963 | 0.954 | 0.997 | -0.955 | 0.978 | 0.943 | -0.952 | 0.983 | 0.944 |
| | | SSE | 0.135 | 0.280 | 0.642 | 0.116 | 0.203 | 0.338 | 0.115 | 0.202 | 0.322 |
| | | ESE | 0.129 | 0.279 | 0.618 | 0.116 | 0.204 | 0.332 | 0.116 | 0.205 | 0.332 |
| | | CPr | 0.922 | 0.954 | 0.956 | 0.924 | 0.948 | 0.938 | 0.928 | 0.956 | 0.932 |
| | (0.74.0.79) | SM | -1.055 | 1.088 | 1.093 | -1.061 | 1.070 | 1.065 | -1.052 | 1.083 | 1.058 |
| | | SSE | 0.145 | 0.330 | 0.752 | 0.129 | 0.232 | 0.399 | 0.125 | 0.227 | 0.387 |
| | | ESE | 0.148 | 0.322 | 0.749 | 0.133 | 0.233 | 0.406 | 0.133 | 0.235 | 0.404 |
| | | CPr | 0.954 | 0.950 | 0.958 | 0.942 | 0.948 | 0.964 | 0.956 | 0.956 | 0.969 |
| | (0.77, 0.82) | SM | -0.917 | 0.949 | 0.883 | -0.912 | 0.953 | 0.844 | -0.910 | 0.955 | 0.837 |
| | | SSE | 0.122 | 0.259 | 0.570 | 0.108 | 0.190 | 0.302 | 0.107 | 0.189 | 0.298 |
| | | ESE | 0.121 | 0.265 | 0.565 | 0.109 | 0.193 | 0.304 | 0.109 | 0.194 | 0.305 |
| | | CPr | 0.856 | 0.936 | 0.930 | 0.846 | 0.942 | 0.928 | 0.846 | 0.944 | 0.934 |
| 0 | (0.75, 0.80) | SM | -1.021 | 0.995 | -0.047 | -1.012 | 1.007 | 0.005 | -1.012 | 0.998 | 0.007 |
| | | SSE | 0.118 | 0.313 | 0.570 | 0.116 | 0.233 | 0.337 | 0.116 | 0.229 | 0.323 |
| | | ESE | 0.123 | 0.297 | 0.567 | 0.119 | 0.224 | 0.331 | 0.120 | 0.225 | 0.323 |
| | | CPr | 0.972 | 0.928 | 0.944 | 0.954 | 0.940 | 0.950 | 0.962 | 0.940 | 0.962 |
| | (0.76, 0.81) | SM | -0.972 | 0.946 | 0.041 | -0.964 | 0.955 | 0.002 | -0.964 | 0.951 | 0.007 |
| | | SSE | 0.110 | 0.298 | 0.542 | 0.108 | 0.215 | 0.311 | 0.108 | 0.215 | 0.303 |
| | | ESE | 0.115 | 0.294 | 0.538 | 0.111 | 0.209 | 0.306 | 0.111 | 0.210 | 0.307 |
| | | CPr | 0.956 | 0.939 | 0.948 | 0.944 | 0.944 | 0.952 | 0.944 | 0.944 | 0.968 |
| | (0.74, 0.79) | SM | -1.075 | 1.051 | 0.055 | -1.065 | 1.064 | 0.003 | -1.064 | 1.050 | 0.011 |
| | | SSE | 0.127 | 0.329 | 0.607 | 0.125 | 0.251 | 0.355 | 0.124 | 0.243 | 0.358 |
| | | ESE | 0.133 | 0.323 | 0.595 | 0.128 | 0.247 | 0.350 | 0.129 | 0.242 | 0.352 |
| | | CPr | 0.940 | 0.940 | 0.946 | 0.944 | 0.940 | 0.958 | 0.948 | 0.946 | 0.960 |
| | (0.765, 0.815) | SM | -0.950 | 0.923 | 0.038 | -0.942 | 0.933 | 0.003 | -0.942 | 0.929 | 0.010 |
| | | SSE | 0.107 | 0.286 | 0.515 | 0.105 | 0.211 | 0.300 | 0.105 | 0.209 | 0.294 |
| | | ESE | 0.112 | 0.267 | 0.497 | 0.108 | 0.202 | 0.295 | 0.108 | 0.203 | 0.296 |
| | | CPr | 0.930 | 0.912 | 0.948 | 0.922 | 0.924 | 0.958 | 0.925 | 0.920 | 0.961 |
| | (0.735, 0.785) | SM | -1.099 | 1.080 | 0.187 | -1.094 | 1.095 | 0.003 | -1.092 | 1.078 | 0.037 |
| | | SSE | 0.134 | 0.349 | 0.608 | 0.130 | 0.251 | 0.381 | 0.129 | 0.251 | 0.379 |
| | | ESE | 0.138 | 0.322 | 0.605 | 0.134 | 0.252 | 0.376 | 0.134 | 0.251 | 0.377 |
| | | CPr | 0.912 | 0.940 | 0.946 | 0.922 | 0.938 | 0.960 | 0.928 | 0.944 | 0.964 |
| | (0.77, 0.82) | SM | -0.928 | 0.901 | 0.035 | -0.921 | 0.910 | 0.001 | -0.922 | 0.907 | 0.008 |
| | | 225 | 0.104 | 0.268 | 0.509 | 0.101 | 0.204 | 0.289 | 0.101 | 0.203 | 0.280 |
| | | ESE | 0.108 | 0.266 | 0.493 | 0.104 | 0.195 | 0.284 | 0.105 | 0.197 | 0.280 |

Table 4.9: Robustness about estimated (π^+, π^-) based on 500 simulations under Design 3 with true values $(\pi^+, \pi^-) = (0.75, 0.80), \beta = (-1, 1), \gamma = 1, 0$

4.3 Application to Children Asthma Data

BIGS is a large population based inspirational actual during data data and the sequences of the sequences of the sequences of the data data, which lappas in 1974. As a pair of 1963, a study was conducted in Stendeuricia, Ohio in evaluation the effect of passive muscing on calculate and the sequences of the sequences of the sequences of the sequences and the sequences of the sequences about the requestion and discover methods and and the sequences at the sequences of the sequences of the sequences of the sequences at the sequences of the sequences of the sequences of the sequences at the sequences of the sequenc

Compared with dials constants of each diff, quotientsmetres are futurely enter and more recomminal to conduct. Understanded, herease of the complexities of wherease of averaging the complexity multiple state of the complexity of the impact of the termination of approximations. The different state information is for any structure of the state of the structure of the different distances and the structure of the structure of the comparison of the structure of the structure of the structure contrast of the structure of the structure of the structure of the community of the structure of the structure of the structure of the community of the structure of the structure of the structure of the community of the structure of the structure of the structure of the community with diagonic terms. The structure of the structure of the removable of the moders at the structure of the structure assumed to be free of classification errors.

In this study, samking hash is considered to be an important rish forces of distions atomic, Maxy atomic and the study of the study of the childran engineery hashin. HitCl represent an injufatout increase in the frequency organing and whereas it is defined in the study of the study of the parential analysis application, in the line study are small mailing. Implied in datheon (Frieder (2006), if is non-also pixeline of the distribution, respected that parential analysis (applicating) around on the followed or study. Distribution Photomics et al. (2006) that the prevalence of velocing in studies of interaction where a strength marking. In order to realisticat the effects of markers marking halfs on children starting, stores simple predintance and the left of a for the Str21 distribution version is considered, and the results shown in Table 13.

It can be seen from Table 4.20 that among three shiften intring with semiginstandings, SigNS of disc has had at least 1 attains attain in the part of parts, are compared to the rare 3.22.56 image of children with summarking models. We have the is a significant discrete between these two portunitages. The Harvan Calsupares not (ρ -schward-222)) and Haldhood ratio into it of the whether the constraints of the discrete between these two portunitages. The Harvan Calsupares not (ρ -schward-222) and Haldhood ratio into (ρ -sub-ord-223), howere, indiconstraints of the discrete between these two portunitages. The Harvan Calmon manifold of the discrete between the two portunitages of the discrete between the contentrory in the analysis. One possible means in the possible minimultification, operalisely between the dashins at lower in the discrete barries of the discrete barries of the discrete barries of the discrete barries of the discrete barries at the analysis, the discrete barries that the discrete barries that at discrete barries that at discrete barries at the schward barries of the discrete barries at the schward barries at the discrete barries at the schward barries at the schwa

| | | | Materna | smok | cing | | |
|----------|-------------------|-----|------------|------|------------|-------|------------|
| Status | Asthma attacks | 0 | Percentage | 1 | Percentage | Total | Percentage |
| Heathy | 0 | 237 | 67.7% | 118 | 63.1% | 355 | 66.1% |
| Infected | 1 | 65 | 18.6% | 32 | 17.1% | 97 | 18.1% |
| | 2 | 25 | 7.1% | 19 | 10.2% | -44 | 8.2% |
| | 3 | 12 | 3.4% | 11 | 5.9% | 33 | 6.2% |
| | 4 | 11 | 3.1% | 7 | 3.7% | 18 | 3.4% |
| | Subtotal | 113 | 32.3% | 69 | 36.9% | 182 | 33.9% |
| | Total | 350 | | 187 | | 537 | |

Table 4.10: Exploratory Analysis of Asthma Data of 537 Children from Steubenville,

risk group, which includes children never attacked by asthma and those attacked only once from age 7-10, the Parame Chi-square test (g-value=0.006) and lästlihood ratio test (g-value=0.070) indicate a non-ignorable association between passive smoking and children starbans.

Loss of attachs have been dones to evaluate the adverse effect of structur's studies of adverse starthan baseling on the disc dates. Als: exemption, Zerur and Gappika (2008) sin alyzed the BECS data by using the generation estimating equations (GER) expressed based on a random effect model. Fitzmannics and Lind (2008) Out-ophical Bioliford enforces based on the endorse" smallest attack to the disc based on the interaction between the trees. Due to the entog association between the current atthins man and the ensemption series attack (2008) expressed to the interaction between the trees. Due to the entog association between the current atthins man and the ensemption series attack (2014) and equations the line transtion model (21) appears to be a remainful choice to analyze the data, and it was the by Stratification and strengt (2007) to commission of effect of strengther and the step Stratification and strengt (2007) to commission and Farendl (2007) developed the generalized optical likelihood (QCL), equivalent to the strengt (2007) developed the QLA) approaches to tractite the effect of matchine mainleight and the previous attitum attace, the current actions actions of dollabous in a dynamic model act-up Strengther (2007). Strengther conf. First (2007) approximation model act-up Strengther (2007). Strengther conf. First (2007).

Becover, all at these analysis were carried out much the annumption that the dotserved data as the size of assumements errors and the data calculated by quere/simulates prachet encoded by accurate induced by questionnaires are posses to chandication were routed interplaced producted by the previous vectors due induced by quering measurement errors in the data back to kined estimation of the unknown measurement. Therefore, is this since it was also also that the outgoing measurement errors in the data back to kined estimation of the unknown measurement. Therefore, is this sincitude, we small out the analysis of the unknown corrected QQL, QQQL, and ML approaches taking the minicularities in into encoderegion to conditionizing mechanism.

Recall that the inherent artima status of the th child in the jth year, which was denoted by T_{ij} , may not be directly observable. Instead, his/her manifest status Y_{ij} can be easily obtained from the information provided by the parenti-reported IBCS questionnaires. Therefore, the relationship between Y_{ij} and T_{ij} can be described by the indicadification model given by

$$Y_{ii} = \pi^+ * T_{ii} + (1 - \pi^-) * (1 - T_{ij})$$
, for $i = 1, ..., 537$ and $j = 1, 2, 3, 4$, (4.33)

where π^+ is the sensitivity and π^- is the specificity of the H6CS questionnaires. We
further assume that the dynamic asthma status of a child can be characterized by the non-linear transition model:

$$\lambda_{i,jj=1}^{*} = P(T_{ij} = 1|T_{ij-1} = t_{i,j-1})$$

= $\frac{exp(\beta_1 + \beta_2 M S_{ij} + \gamma t_{i,j-1})}{1 + exp(\beta_1 + \beta_2 M S_{ij} + \gamma t_{i,j-1})}$, (4.34)

where MS_{ii} is the maternal smoking status, and $t_{i0} = 0$ are the baseline observations.

We some that the smalliply τ^{a} and specificity τ^{a} of the quarkmain work is that aday are constants over time and they are independent of adapters and covariants. To not look knowledge, the semithryg and specificity of this study are not jet well contained. However, Yang et al. (1990) conducted a survey to assume the effect of these revisionnest or a divident antibus. To increase have effect of the theorem and the theorem of the BOCS and they repeated a semitivity of 0.00 and a specificity of 0.05. Thus, we use that immunity, $(\tau^{a}, \tau^{a}) = 0.00, 0.03)$ as a dives submit of the true multicity and specificity or or not hy.

Read that in the simulation step conducted is fielded 1.2.4 but noise estimates of $\sigma = (\sigma, \tau)$, we concern the plasming single of $(\sigma, \tau) = -(0, 1.3)$ and that the done-radiant set error from. When one is that $\pi^2 < 1$. Similarly, in this subsection, we choose to compare that $\pi^2 = \tau = (0, 1.3)$ with the second second

It can be seen from Table 4.11 that, when taking misclassification into account, the estimates of the model parameters are very different from those obtained by ignoring misdiagnosis, such as the estimates provided in [Sutradhar and Parrell (2007)]. To be people, the energed OGG anti-anison of $A_{\rm pol}$ and γ are larger than the correct origing more enstances. Notice that in Table Lin-Space reads a supplied symmitdynamicrow which is update measurable, which may be due to the out clintermute frame the higher order responses. It can be would by the fact that $S_{\rm COG}$ and $S_{\rm LC}$ including more information from data produce a significant positive so-accident with the prior mathematication from data produce with models protein. Even, see, the COGL and Lin Querendon-dimensional higher efficiency over the CQ2, argumeds. We therefore profer to average the results provided by the COQ2, and Mir.

Abbaga the material anaking offset β_1 and the dynamic dispersion reflect γ_1 and the main instease of antiteristican, the olds result (O)(1) of arbitrans with respect to material making, and the olds ratio with respect to gravitous and analysis of the main forces or the olds ratio of the OS with respect to material making areads directly the solution with of authom attack of shifteen leving with making methods directly the solution with α of mathem at the old with respect to material the OS with respect to material making methods directly the solution with the material analoging, the correlated OOQLs estimates of σ^{0} is equal to 1200 which is generic than the mainer dimathem OOQLs with the OS with respect to the corrected estimation of β_1 is graviter, that the corresponding naive estimator of β_2 under all these estimators

Another quantity of interest in this study is c², the odds ratio of developing asthma with respect to whether or not the child had an attma attack in the previous year. As mantianed above the GQL estimate shows an unreasonable maprive effect of a previous astma attack, while the OGQL and ML estimates indicate strongly positive disc. Therefore, we prefer to use the numbur under the OOQL of MM, sequences. From Table 11, use can use that the corrected OOQL dominate of this older note of developing further asthmas strate is 10.802 (M) (SO CI in (19330) 10.27573) when $(\tau^+, \tau^-) = 10.80.803$, which the observation is only 2000 (1935 CI i-(1230), 550477) yiels micelocalification is ignored. Analysis tables dipondum errors into consideration reveals strategiz association between the correct atualman attava with the previous strate, which is in noncolinear with the mediad pretext. Therefore, the prepared at a prior attimum attack is of importance in disposing a duB's correct atualman attava.

Read that is the simulation study in Section 42.3, we combined a relative commission on the events obtained of $\theta = (T_{i}^{-1}/2)^{-1}$ this speet to different chicks of π^{*} and π^{-} . In the analysis of the aximum data, we have also examined the rebundance of the corrected outstants of β_{i} and γ_{i} support with the robundance of the simulation of the odds ranks, σ^{*} and σ^{*} , many. The parameters are pictual versus the samitrity π^{*} and querifying π^{*} in Figures 1.4.3, whereas the estimated of variants or plotted in Figures 4.4.5. As interesting finding is that is is always more moder where estimating the interproper, β_{i} , which durings and β_{i} estimation of the simula information of antions incidence. Different values of estimated semisitivity and specificity true during influence on the estimation of Λ in an interesting frame of the order model parameters.

It can be seen from Figure 4.4 (a), (b), (c) that when both the sensitivity and specificity increase, the estimated OR e^{b_1} decreases. This indicates the atternation effect of overestimated sensitivity and specificity on the estimation of model parameters. Furthermore, the decremaing rate of the OGQL and MI estimates of OR (e^{b_1} along the sensitivity is higher in the case of a low sensitivity than that in the case of a big specificly. Also the decuss of rational relative risk along predictly in mathlice from the shark size discriming of available with higher that the estimated specificity has much stronger influence on the estimation of OR (n^{3}) shares that of the astimation simulative. Furthermore, than Faguer 6.5 (a) and (b), in its opportution the continue of n^{3} and n^{4} CO (n^{3} CO (n^{3}

Based on Figures 1.1–3. we conjecture that the second we higher order correlation among the responses of the actual atthms data may carry important information about the interested parameters, specifiedly the dynamic dependence parameter γ . This may also replain the fullness of GQL approach in estimating γ , which involves only the first order responses. Therefore the OGQL and ML approaches are recommended to analyze the atthms data.

In manage, in this series, we analyze the children satemat size the SireNetwork Network (SireNetwork Network) and the large 1 dependence model (3.43), the corrector distribution and (4.13) and the large 1 dependence of the distribution and with the parameters. Use more relations the distribution methers with the parameters distribution of the distribution of the

attacks in predicting the current atimas atoms. By the comparison of the OQLapproach with the OQL and ML, expension. He weread order attaction of the sponsons include important information about the interested effects, especially the poser of the parsions attalan, attack is possible in the finite starts. So we recomtact OOQL and ML equipruds the startistical informer, Their barrents attack defined on the OOQL and ML equipruds the startistical informer, the to attrack dependence of the ML approach on model assumptions, the OOQL approach is therefore workshold beam on Bio flat acids our concentral.

Transhy, we give a kinds discussion about the semilative s² will and specificity viconduction the social, and accounds in the factor, we are available constrained periodically full factors in the factor sectors find in parallel, these two quantifies may change over these factors. For example, is and self reported after than [Specier (1996)], ackains (1996) reported by the fiftheries socializing and appendixely between the parential report of the effective in socializing of appendixely between the parential report group and eff-report group and self-report after than [Specier (1996)], ackains (1996) reported the difference in somithicity and specificity between the parential report group and eff-report group and periodic provides attack and the emsittivity, specificity simultaneously for these data set.

| | | GQL | | | OGQL | | | ML | | |
|------------------|------------------|-----------|-----------|---------|-----------|-----------|----------|-----------|-----------|---------|
| (π^+, π^-) | Quantity | β_1 | β_2 | γ | β_1 | β_2 | 2 | β_1 | β_2 | γ |
| (1,1) | Estimate | -1.7737 | 0.2843 | -0.4959 | -2.1886 | 0.2205 | 1.9554 | -2.1886 | 0.2205 | 1.9554 |
| | Ste [®] | 0.1185 | 0.1264 | 1.2068 | 0.0891 | 0.1323 | 0.1532 | 0.0893 | 0.1307 | 0.1444 |
| | $p - value^*$ | 0.0000 | 0.0245 | 0.6811 | 0.0000 | 0.0955 | 0.0000 | 0.0000 | 0.0916 | 0.0000 |
| | OR ^b | | 1.3288 | 0.6090 | | 1.2468 | 7.0669 | | 1.2468 | 7.0059 |
| | LB ^c | | 1.0372 | 0.0572 | | 0.9620 | 5.2340 | | 0.9650 | 5.3252 |
| | UBd | | 1.7024 | 6.4838 | | 1.6159 | 9.5417 | | 1.6108 | 9.3782 |
| (0.8,0.95) | Estimate | -1.9567 | 0.4071 | -0.5419 | -2.7858 | 0.3079 | 3.7769 | -2.7771 | 0.3206 | 3.7296 |
| | Ste | 0.1686 | 0.1902 | 1.9938 | 0.1628 | 0.1992 | 0.4265 | 0.1187 | 0.1622 | 0.2308 |
| | p-value | 0.0000 | 0.0323 | 0.7858 | 0.0000 | 0.1221 | 0.0000 | 0.0000 | 0.0481 | 0.0000 |
| | OR | | 1.5024 | 0.5816 | | 1.3505 | 43.6822 | | 1.3779 | 41.6622 |
| | LB | | 1.0349 | 0.0117 | | 0.9209 | 18.9339 | | 1.0027 | 26.5040 |
| | UB | | 2.1812 | 28.9577 | | 2.0103 | 100.7787 | | 1.8936 | 65.4896 |

Table 4.11: Analysis of Asthma Data of 537 Children from Steubenville, Ohio in 1970 Thios Michael and Consideration

*Standard error

^bOdds ratio of maternal smoking and a previous asthma attack given by $exp(\hat{\beta}_1)$ and $e^{i + 1}$ ^clower bound of 95% confidence interval of OR, given by $e^{i - 1}e^{i - 1}e^$





(c) ML estimate of A

Figure 4.1: Estimates of the Intercept β_1 in Model (4.34) for Asthma Data of 537 Children from Steubenville, Ohio in H6CS Taking Misdiagnosis into Consideration.





Figure 4.2: Estimates of the Effect of Mothes's Smoking Status β_2 in Model (4.34) for Asthma Data of 537 Children from Steubenville, Ohio in H6CS Taking Misdiagnesis into Consideration.







4.4 Joint Modeling the Misclassified Data with Missing Information Due to "Unsure" Responses

4.4.1 Model description

The channel michandracian model for entrapriced data are different between the categories watering. The all the shore-orienterized X. T. and Y. generally have an equal number of channe, for example the mindaminiation described by Table 2.2 in Chapter 2.2, But, in practice, there are some cases that the manifest watebox Y may in Table 2.2.3. Chapters 2. It is accludingly and equivalent term of the in Table 2.2.4. Chapters 2. It is accludingly provided parameters are predominantly in Table 2.2.4. Chapters 2. It is accludingly provided parameters are many in Table 2.2.4. Chapters 2. It is accludingly and equivalent term of the transfer of c1 data's table are culticed through some preforman general maters to table table the transfer of the transfer of the transfer of the "tamatic" of c1 data's based". For instance, in studies of thiddens anthana, a question must be est as follow:

Do you think that your child had asthma in the past 12 months?

(1) Yes; (2) No; (3) Unsure.

The possible mickenfizition on behaves by Table 4.12. This situation is much the unbalanced mickenfizition in Chapter 2. A chift's adminst attains may be roported by his or her possition as "addendig" (answer; yos), "healthy" (canver; yo), or "undeckdet" (answer; unsare), while his or her true statuss can only be one of the two statuses: infected ond buddy. The information about those children who are rorouted by responses. The status information about those children who are rorouted by responses. The status information about these children who are rotored by responses. The status information about these children who are rorouted by responses.

| | Asthma status (T) | | | | | | |
|---------------------|-------------------|-------------|--|--|--|--|--|
| Reported status (Y) | Infected (1) | Healthy (2) | | | | | |
| Yes (1) | π_{11} | π12 | | | | | |
| No (2) | π_{21} | π_{22} | | | | | |
| Unsure(3) | π_{11} | π_{32} | | | | | |

Table 4.12: Unbalanced misclassification of children asthma

are often transfer in as hird of mining values in provides. Generally, the quantity x_1 in Table 4.12, the producting that the protocol maternal producted table gives as "namer" answers, is different from x_0 which is the probability that the present of a hubble duil give an "namer" response. This implies that students a subject has a "mather" approach with realized to hardy-term team. Therefore, it is in mining at reasonic MMM but not on implicit productly at reasonic (MMM). In fact, even thus " 3^{110} " as " 3^{100} " response may needer charalization in errors. Addimingd the trans statuse of an alloptica an impossible to know, the productification errors. Addimingd the trans statuse two probabilities x_1 and x_2 patient of the analysis with "namer" impeases, can be emitted thready hours be addition and the status of t

In this section, we apply the unbalanced mixcloseffication model to describe this kind of data subject to both mixcloseffication and missing information. Statistical inference based on the GQL approach about the unbalancingly miscloseffield data is developed. We define the observed response Y_{ii} as follows:

$$Y_{ij} = \begin{cases} (1,0)^r, & \text{the ith child's parents answer "yes" at the jth time point;} \\ (0,1)^r, & \text{the answer is "mo";} \\ (0,0)^r, & \text{the answer is "unsure"}. \end{cases}$$

where i = 1, 2, ..., I and j = 1, 2, ..., J. The true response T_{ij} takes the value 1, if the true status is "infected", 0 otherwise. So Y_{ij} is a trinomial variable, while T_{ij} is a binary variable.

Let II denote the FMC-matrix which is defined by

$$\vec{\Pi} = \begin{pmatrix} \pi_{11} & \pi_{12} \\ \pi_{21} & \pi_{22} \\ \pi_{31} & \pi_{32} \end{pmatrix}.$$
(4.35)

Then the MC-matrix is given by

$$\Pi = \begin{pmatrix} \pi_{11} & \pi_{12} \\ \pi_{21} & \pi_{22} \end{pmatrix} = [\pi_1, \pi_2], \quad (4.36)$$

where $\pi_1 = (\pi_{11}, \pi_{22})'$, and $\pi_2 = (\pi_{12}, \pi_{22})'$. Based on the MC-matrix, the unbalanced misclassification model describing the relationship between Y_{ij} and T_{ij} , according to Section 4.1, can be written as

$$Y_{ii} = \Pi * \hat{T}_{ii} = \pi_1 * T_{ij} + \pi_2 * (1 - T_{ij}),$$
 (4.37)

where $\hat{T}_{ij} = (T_{ij}, 1 - T_{ij})'$.

It is easy to derive some useful moments of the observed response Y_{ij} . The expectation of Y_{ii} is given by

$$\mu_{ij} = E(Y_{ij}) = \Pi \begin{pmatrix} \eta_{ij} \\ 1 - \eta_{ij} \end{pmatrix} = \pi_2 + (\pi_1 - \pi_2)\eta_{ij},$$

where $\eta_{ij} = E(T_{ij})$ is the expectation of the true response T_{ij} which is a scalar. The variance-covariance matrix of Y_{ij} is given by

$$Var(Y_{ij}) = Var[E(Y_{ij}|T_{ij})] + E[Var(Y_{ij}|T_{ij})]$$

 $= \eta_{ij}V_{\tau_1} + (1 - \eta_{ij})V_{\tau_2}$
 $+ (\pi, - \pi_2)Var(T_2)(\pi, - \pi_2)'.$

As discussed in Section 4.1, $Y_{ij} \sim Multimomial(1, \mu_{ij})$, hence the covariance matrix of Y_{ii} can be written in an alternative form

$$\Sigma_{ijj} = Var(Y_{ij}) = V_{\mu_{ij}} = diag(\mu_{ij}) - \mu_{ij}\mu'_{ij}$$
(4.38)

For u < j, the covariance between Y_{ij} and Y_{iu} is given by

$$E_{ijw} = Cov(Y_{ij}, Y_m)$$

$$= \Pi Cov(\overline{T}_{ij}, \overline{T}_m)\Pi^{\prime}$$

$$= \Pi \left(\frac{ov(T_{ij}, T_m) - ov(T_{ij}, 1 - T_m)}{ov(1 - T_{ij}, 1_m) - ov(1 - T_{ij}, 1 - T_m)} \right)\Pi^{\prime}$$

$$= (\tau_i - \tau_i)Cov(T_i - T_{ij})(\tau_i - \tau_i)^{\prime}$$

It can be seen that Σ_{iin} is singular and of rank 1.

Based on the previous development, we can write the covariance matrix of Y_i in the form of

$$\Sigma_{i} = \begin{pmatrix} \Sigma_{i1} & \Sigma_{i2} & \cdots & \Sigma_{iJ} \\ \Sigma_{i2} & \Sigma_{i2} & \cdots & \Sigma_{iJ} \\ \vdots & \vdots & \ddots & \vdots \\ \Sigma_{iJ1} & \Sigma_{iJ2} & \cdots & \Sigma_{iJJ} \end{pmatrix}_{JJ \times JJ} , \quad (4.39)$$

144

4.4.2 Estimation of model effects

There are several methods to deal with the imperfect categorical data subject to both misclassification and missing values due to the "unsure" responses

Case I: Diobics glue "mouse" response and gaussian minimum dimensions. In this may, the observed maps made have been dimensional transmission of the strength in the dimension of the strength strength strength and the strength maps strength gaussian of the strength strength strength maps and the constant and the constants strength gaussian dimension of Gaussian Japan, and the constants strength gaussian dimension of Gaussian Japan and the constants strength gaussian dimension of Gaussian Japan maps and the constants strength gaussian dimension of the strength Ta.

$$\tilde{Y}_{ii} = \pi_w^+ * T_{ii} + (1 - \pi_w^-) * (1 - T_{ii}),$$
 (4.40)

and the "working" FMC matrix is

$$\hat{\Pi}_{w} = \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix}$$
,

therefore the "working" sensitivity and specificity $(\pi^+_{e_1},\pi^-_{w}) = (1, 1)$. Notice that, in this case, subjects may have unequally-spaced observations due to deleting the "unsure" values.

Case II: Deleting the "assure" values but taking misclassification into consideration. In this case, we use the same misclassification model as (4.40) due to deleting the third category "unsure" from observed response Y_i. Accordingly, the probabilities in a FMC matrix will be proportionally reassigned as follows

$$\widetilde{\Pi}_{qr} = \left(\begin{array}{cc} \frac{\pi_{11}}{\pi_{11} + \pi_{21}} & \frac{\pi_{12}}{\pi_{12} + \pi_{22}} \\ \frac{\pi_{22}}{\pi_{11} + \pi_{21}} & \frac{\pi_{22}}{\pi_{12} + \pi_{22}} \end{array} \right)$$

which implies that the "working" sensitivity and specificity $(\pi_w^+, \pi_w^-) = (\frac{m_w^{-1}}{m_w^{-1}+m_w^{-1}}, \frac{m_w^{-1}}{m_w^{-1}+m_w^{-1}})$. The subjects may also have unequally-spaced observations similar to those in *Case L*.

Case III: [Jurving these subject with all hard new mixing values but hilds printcharofication into consideration. In the case that we delote all the subjects with at loss one "maxuo" response from the study, the mixical subject maxuo of the study of the mixical subject model (4.40) can still be used because there are zo "maxuo" response in the data any more At the same time, the probabilities in a FMC matrix will be proportionally remaining an follow:

$$\widetilde{\Pi}_{gr} = \begin{pmatrix} \frac{\tau_{11}}{\pi_{11} + \pi_{21}} & \frac{\tau_{12}}{\pi_{12} + \pi_{22}} \\ \frac{\tau_{22}}{\pi_{11} + \pi_{21}} & \frac{\pi_{22}}{\pi_{12} + \pi_{22}} \end{pmatrix},$$

and the "working" sensitivity and specificity $(\pi_{\mu}^{+},\pi_{\mu}^{-}) = (\frac{\pi_{\mu}}{\pi_{\mu}+\pi_{\mu}}, \frac{\pi_{\mu}}{\pi_{\mu}+\pi_{\mu}})$. D3-ferent from *Case II*, the data in this case do not involve unequally-spaced observations.

Case IV: Ignoring the mixelassification but taking the "source" values into ensuieration. If we take "meson" responses into consideration, there will be three observed categories. In this situation, the response Y_{ij} is a tritomital variable of which the value (0, 0)' implies that we get an "source" assocret. Ignoring mixelassification mesons that the probabilities of Type I and type II encourse as assumed to be 0's. Therefore, the "working" FMC matrix is

$$\hat{\Pi}_{w} = \begin{pmatrix} \pi_{11} + \pi_{21} & 0 \\ 0 & \pi_{12} + \pi_{22} \\ \pi_{21} & \pi_{32} \end{pmatrix}.$$

In this case, all subjects have complete data at time j = 1, 2, ..., J. Then we use the unbalanced misclassification model (4.37) with the assumed FMC matrix $\hat{\Pi}_w$ which is given above.

Case V: Taking fork the mixing values and mixelassification into consideration. One may take both the "amone" values and classification errors into consideration to obtain more reliable statistical inference. In this situation, we use the model (4.37) with the true FMC matrix to develop statistical inference. The true FMC matrix is

$$\widehat{\Pi}_{w} = \widehat{\Pi} = \begin{pmatrix} \pi_{11} & \pi_{12} \\ \pi_{21} & \pi_{22} \\ \pi_{31} & \pi_{32} \end{pmatrix}$$
.

In this case, we have equally-spaced observations for all subjects at time $j=1,2,\ldots,J.$

Case VI: Ideal case that the data $\{t_{ij}\}$ are available. Suppose that we know the data t_{ij} of the true response, and use the data to conduct statistical inference. The estimation can be based on the model in Section 3.2. In an alternative way, we can use the model (≤ 20) with the FMC

$$\hat{\Pi}_{w} = \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix}$$

for the data t_{ij} , i = 1, 2, ..., I, and j = 1, 2, ..., J. In this case, the sensitivity π_w^+ and specificity π_w^- are (1, 1). It should be pointed out that, this case is seldom possible in practice, and it is only applicable in simulation studies.

In the following part, we discuss the estimation of model parameters based on the GQL approach in the six cases. For this purpose, we assume, like that in Section 4.2, that the true response T_0 follows the nonlinear transition model (4.9). Therefore, the moments of the true response T_{c0} on the wealty obtained from Section 3.1.2.

As 'mann' response is due trends as a missing value in practice as matriced in Section 4.1. Therefore, we simply due that limit of mining turber in cases I_c H and H_c we can see \tilde{h}_{cl} in the development of OQL estimation. The expectation h_{ll} and continues $Var(\tilde{h}_{cl})$ of \tilde{h}_{ll} are similar to those in (143) and their compatibies on follow the similar development in Section 4.1. In cases H' and V_c we use the eriginal response Y_{lc} . In fact the Milowing development can be generalized to some other cases of mining values with MAR mechanism which can be modeled by the valuebarred micloarderine.

4.4.2.1 Ignoring the "unsure" responses

Gravity, there are two metanisms to obtain the mining values more placetime of the star of the adoptent with a basist new "manne", which makes the analysis simple but leads to hand efficiency. The obtainet and in *Cost II*, is in only delete these "manne" regressions and use all valid observations ("yes" or "m"), which would in its low of the information that the first machination. However, the second second second considering in minimum star of COS assesses. The estimative second considering in minimum stars of COS assesses. The estimative second based on the available data can be written as

$$\sum_{i=1}^{p^{\mu}} \left(\frac{\partial \tilde{\mu}_{i}^{\mu}}{\partial \theta} \right)^{\prime} (\tilde{\Sigma}_{i}^{\mu})^{-1} (\tilde{g}_{i}^{\mu} - \tilde{\mu}_{i}^{\mu}) = 0. \quad (4.41)$$

In the showed GQL estimating quantum (14.1), gf dwates the showed values of adopts (1, and g7 and g2 sequentit the coveraging expectation and coverages matrix, respectively. In addition, P⁴ dwates the total number of subjects with at locat one with the gap (1) are randomly single (1) the gap (2) are randomly single (1) the formation of the gap (2) are randomly single (1) the formation of the gap (2) are randomly single (1) the choice of the gap (2) are randomly single (1) the formation of the gap (2) are randomly single (2) the choice of the gap (2) are randomly single (2) the choice of the gap (2) are randomly single (2) the choice of the gap (2) are randomly single (2) the choice of the gap (2) are randomly single (2) the choice of the gap (2) and the choice of the gap (2) and the gap (2) are randomly single (2) the gap (2) are randomly single (2) the gap (2) are randomly (2) and (2) the gap (2) are randomly (2) and (2) are randomly (2) are randomly (2) and (2) are randomly (2)

$$\sum_{i=1}^{l} \frac{\partial \hat{\mu}'_{i}}{\partial \theta} (\tilde{\Sigma}_{i}^{A})^{+} (\hat{y}_{i} - \tilde{\mu}_{i}) = 0, \quad (4.42)$$

where the adjusted countance matrix $\tilde{\Sigma}_{i}^{A}$ is built by introducing a missing indicator matrix (MM) Λ_{i} to the complete covariance matrix $\tilde{\Sigma}_{i}$, $(\tilde{\Sigma}_{i}^{A})^{A}$ denotes the Moore-Penrose inverse of $\tilde{\Sigma}_{i}^{A}$. In the following part, we will show how to construct the adjusted covariance matrix $\tilde{\Sigma}_{i}$.

For simplicity in notation, we here consider one subject with a series of repeated measurements, which can be easily generalized to longitudinal studies with multiple participants. To be specific, we suppose that the complete data are $\hat{y} =$ (g_1, g_2, \dots, g_k) , the observed data as $p^{-1} = (g_1, g_2, \dots, g_k)$, at the time point $g_1 \in g_1 \subset \dots \subset g_k$ where n + s = J. Similar to the previous paragraph, we let $k_1 \subset k_2 \subset \dots \subset k_m$ where n + s = J. Similar to the previous paragraph, we let E and D denotes the variance-constance matrix of complex Γ and the observed Γ^{-1} respectively. It is only use the D^{-1} is complex by denotes at the intersection of the (g_1, g_2, \dots, g_k) are been D^{-1} is compositive. The model of the (g_1, g_2, \dots, g_k) is denote the MIM as a $-Q_{k-1/2}$ of the the (g_1, g_1, \dots, g_k) are having $n \in \mathbb{N}$.

$$\lambda_w = \begin{cases} 1, & \text{for } u = v \text{ and } \hat{y}_u \text{ is a valid answer ("pes" or "uo");} \\ 0, & \text{for } u = v \text{ and } \hat{y}_u \text{ is an "unsure" answer;} \\ 0, & \text{for } u \neq v. \end{cases}$$

So the diagonal vector of Λ is the missing indicator vector. It is obvious that Λ is an identity matrix for the complete data \hat{y} , and is singular in presence of "unsure" values with $\Lambda^+ = \Lambda$. Then the adjusted covariance matrix is defined by

$$\tilde{\Sigma}^{A} = \Lambda \tilde{\Sigma} \Lambda'$$
, (4.43)

the elements at the $(k_1, k_2, ..., k_n)$ th rows and $(k_1, k_2, ..., k_n)$ th columns of $\tilde{\Sigma}^A$ are all zeros, and $\tilde{\Sigma}^A$ is singular.

Following the development of the adjusted covariance matrix $\overline{\Sigma}^{4}$, we can prove that the adjusted GQL estimating equations (4.42) based on $\overline{\Sigma}^{4}$ are equivalent to the observed GQL estimating equations (4.41).

It is easy to see that $\tilde{\Sigma}^{4}$ can be non-singularly transformed into $\left(\widetilde{\Sigma}^{*}_{2,ee} \quad 0_{exem}\right)$ by exchanging the non-zero rows and columns $(j_{1}, j_{2}, ..., j_{d})$ with the zero rows and columns $(k_{1}, k_{2}, ..., k_{d})$. The covariance matrix $\tilde{\Sigma}$ of the observed data g^{μ} is a nonsingular submatrix of $\tilde{\Sigma}^{A}$. This implies that

$$\widetilde{\Sigma}^{A} = I_{1,j_{1}}I_{2,j_{2}}\cdots I_{a,j_{a}}\begin{pmatrix} \widetilde{\Sigma}^{a} & \mathbf{0} \\ \mathbf{0} & \mathbf{0} \end{pmatrix} I_{a,j_{1}}\cdots I_{2,j_{7}}I_{1,j_{7}}.$$
 (4.44)

A matrix I_{uv} is defined by



where the elements 0(u) and 0(v) in the diagonal line mean that the uth and wth elements are zeros. It is issues that $L_{ab}^{-1} = L_{ab}$ and $L_{ab} = L_{ab} = L_{ab}^{-} = R_{ab}^{-}$. For any matrix $B, I_{ab}B$ represents exchanging the wth and wth rease of matrix B, while BL_{ab} implies exchanging ut that we development of B.

Keeping the following well-known conclusion in mind

$$\begin{pmatrix} B_{\mu\nu} & 0 \\ 0 & 0 \end{pmatrix}^+ = \begin{pmatrix} B_{\mu\nu}^{-1} & 0 \\ 0 & 0 \end{pmatrix}$$
.

it is easy to show that the Moore-Penrose inverse of Σ^A is

$$(\widetilde{\Sigma}^{4})^{+} = I_{1,j_{1}}I_{2,j_{2}} \cdots I_{nj_{n}}\begin{pmatrix} (\widetilde{\Sigma}^{n})^{-1} & 0 \\ 0 & 0 \end{pmatrix} I_{nj_{n}} \cdots I_{2,j_{2}}I_{1,j_{1}}.$$
 (4.45)

By comparing (4.41) with (4.40), one can seen that $(\overline{\Sigma}^{A})^{+}$ has the same zero rows and columns as $\overline{\Sigma}^{A}$.

We now show that the adjusted GQL estimating equations (4.42) is equivalent to the observed GQL estimating equations (4.41).

Denote

$$\frac{\partial \tilde{\mu}'}{\partial \theta} = (\frac{\partial \tilde{\mu}_1}{\partial \theta}, \frac{\partial \tilde{\mu}_2}{\partial \theta}, \dots, \frac{\partial \tilde{\mu}_J}{\partial \theta})$$

$$\begin{split} d_{T} - \tilde{\mu} &= (\tilde{\mu}_{-} - \tilde{\mu}_{0}, \tilde{\mu}_{-} - \tilde{\mu}_{0}, \dots, \tilde{\mu}_{0} - \tilde{\mu}_{1})^{T}. \text{ We have } \\ & \frac{d_{T}}{d} \tilde{\Omega}^{1} (\gamma - \tilde{\mu}) = \frac{d_{T}}{d\theta} L_{1,0} L_{2,0} \dots L_{2,0} \begin{pmatrix} (\tilde{D}^{-1} - 0) \\ 0 & 0 \end{pmatrix} L_{2,0} \dots L_{2,0} L_{2,0} L_{2,0} (L_{2,0} - \tilde{\mu}_{0}) \\ & \tilde{\mu}_{0} - \tilde{\mu}_{0} \\ & \tilde{\mu}_{0} - \tilde{\mu}_{0} \\ & \tilde{\mu}_{0} - \tilde{\mu}_{0} \\ & \tilde{$$

where $\frac{m_{1}}{2}L_{1,k}L_{1,k}\cdots L_{k,k} = \left\{\frac{m_{1,k}}{2}, \frac{m_{2}}{2}, \dots, \frac{m_{2}}{2}, \dots,$

$$\sum_{i=1}^{l} \frac{\partial \hat{\mu}'_{i}}{\partial \theta} (\Lambda_{i} \widetilde{\Sigma}_{i} \Lambda_{i})^{+} (\hat{g}_{i} - \hat{\mu}_{i}) = 0, \quad (4.46)$$

where Λ_i is the MIM of the subject i, for i = 1, 2, ..., I. By applying the adjusted GQL approach in simulation studies, we do not have to identify the \hat{p}_i^* , $\hat{\mu}_i^*$ and $\tilde{\Sigma}_i^*$ for each sample unit, and we only need to find Λ_i . In the practical applications, we can simply assign any finite values to the "unsure" responses.

In addition, if one simply delete a subject with at least one "unsure" answer, he/she can still use the adjusted GQL approach by assigning a zero MIM for this subject. The MIM for a subject *i* can be defined as

$$\Lambda_i = \begin{cases} 0_{J_{N,J_i}} & \text{for subject } i \text{ with incomplete data;} \\ I_{J_{N,J_i}} & \text{for subject } i \text{ with complete data,} \end{cases}$$
(4.47)

where $I_{J\times J}$ represents the $J \times J$ identity matrix. Let i_u , u = 1, 2, ..., F denote the individuals with complete observations during the studying period. It is apparent that the adjusted estimating equations are equivalent to

$$\sum_{n=1}^{P} \frac{\partial \hat{\mu}_{n}^{\prime}}{\partial \theta} \tilde{\Sigma}_{i_{n}}^{+}(\hat{y}_{i_{n}} - \hat{\mu}_{i_{n}}) = 0. \quad (4.48)$$

4.4.2.2 Taking missing values into account

Ablong, none participate any give "samar" answer, the values of the coverpanding conviction may still be useful. In addition, we may how good heading and the probability is the similared michaelism matrix fit in (CM). In this distinct, the mining values implicit with the corresponding convision and the michaelismetic periodition can able provide model information for statistical minimum. Therefore, one may want to take these "samar" answers into convidestants. Note that, as maximum before, there are three descence outprive "saw", "ye and "samar", "So we will use to regular balance the site of the state of the state. Note that, as maximum before, there are these descence outprive "saw".

In Case IV and Case V, we construct the GQL estimation based on the unbalanced misclassification model (4.37) and the corresponding FMC matrices described in Section 4.4.2. The estimating equations are given by

$$\sum_{i=1}^{l} \frac{\partial \mu'_i}{\partial \theta} \Sigma_i^+(y_i - \mu_i) = 0, \quad (4.49)$$

 μ_i and Σ_i can be computed based on the calculations in Section 4.4.1.

In fact, the misclassification model (4.37), the related FMC matrix in (4.35) and an adjusted version of the GQL estimating equations which are given by

$$\sum_{i=1}^{I} \frac{\partial p_i'}{\partial \theta} (\Lambda_i \Sigma_i \Lambda_i)^+ (y_i - \mu_i) = 0. \quad (4.50)$$

can accommodate the three cases where missing values are deleted in Section 4.4.2.1. In (4.50). A. is defined as

$$\Lambda_{c} = \begin{pmatrix} \Lambda_{cc} & & \\ & \ddots & \\ & & \Lambda_{cf} & \\ & & \ddots & \\ & & & \Lambda_{cf} \end{pmatrix}, \quad (4.51)$$

where

$$\Lambda_{ij} = \begin{cases} I_{2\times 2}, & \text{the answer of the subject } i \text{ at the time } j \text{ is "yes" or "no";} \\ 0_{2\times 2}, & \text{otherwise.} \end{cases}$$

in Case I and Case II, whereas

$$\Lambda_i = \begin{cases} J_{2J\times 2J} & \text{there are no "unsure" anwser for the subject } i; \\ \mathbf{0}_{2J\times 2J}, & \text{the ith subject has at least one "unsure".} \end{cases}$$

in Case III

The working FMC involved in (4.37) is, in Case I,

$$\widetilde{\Pi}^w = \begin{pmatrix} 1 & 0 \\ 0 & 1 \\ 0 & 0 \end{pmatrix}.$$

And it is

$$\widehat{\Pi}^{w} = \begin{pmatrix} \frac{\pi_{11}}{\pi_{11} + \pi_{12}} & \frac{\pi_{21}}{\pi_{21} + \pi_{22}} \\ \frac{\pi_{12}}{\pi_{11} + \pi_{12}} & \frac{\pi_{22}}{\pi_{21} + \pi_{22}} \\ 0 & 0 \end{pmatrix}$$

in both Case II and Case III. In equations (4.50), we use the data y_i and expectations μ_i and covariance matrices Σ_i instead of \hat{y}_i and its moments.

4.4.3 Simulation

In this section, we carry out simulation studies to check the performance of the GQL estimates in all the size cases described in Section 4.4.1. The ideal case means that we use the data with the rare response T, which follows the nonlinear transition model (4.9) to a estimate parameters in the model.

4.4.3.1 Design

We apply the same covariate design as Design 3 described in Section 4.2.3.1. It is rewritten here by

$$x_{0(1)} = 1, (j = 1, 2); x_{ij(1)} = 0, (j = 3, 4), i = 1, ..., 140,$$

 $x_{ij(1)} = 1, j = 1, ..., 4, i = 141, ..., 420,$
 $x_{0(1)} = 1, (j = 1, 2); x_{ij(1)} = -1, (j = 3, 4), i = 421, ..., 560.$
 $x_{0(1)} = -0.5, (j = 1, 2); x_{ij(2)} = 0.5, (j = 3, 4), i = 1, ..., 140;$

$$x_{ij(2)} = j/4, \ j = 1, ..., 4, \ i = 141, ..., 420,$$

 $x_{ij(2)} = 0.5(j-2), \ j = 1, ..., 4, \ i = 421, ..., 566$

As far as the FMC matrix design is concerned, we consider three different settings which are along by

(a).
$$\vec{\Pi} = \begin{pmatrix} 0.8 & 0.02 \\ 0.03 & 0.9 \\ 0.17 & 0.08 \end{pmatrix}$$

(b). $\vec{\Pi} = \begin{pmatrix} 0.7 & 0.1 \\ 0.1 & 0.75 \\ 0.2 & 0.15 \end{pmatrix}$
(c). $\vec{\Pi} = \begin{pmatrix} 0.8 & 0.08 \\ 0.17 & 0.9 \\ 0.03 & 0.02 \end{pmatrix}$

From setting (a), (b) to (c), the classification errors becomes severer and severe. On the other hand, among the three settings, setting (c) involves the levest values of $(0, 0)^{\prime}$ for y_{40} , which correspond to "mours" answers, while setting (b) generate the must values of $(0, 0)^{\prime}$ for y_{10} .

In each setting, the true binary data t_{ij} , i = 1, 2, ..., 560 and j = 1, 2, ..., 6 are generated following the model (4.3). The observed data y_{ij} are generated from the unbalanced mischasification model (4.33). The generation of y_{ij} follows the procedure described below.

Once we have t_{ij}, we can firstly generate a trinomial variable U from Trinomial(t_{ij}, π₁).

In R package, it should be $U < -rmaltinom(t[i, j], 1, \hat{\pi}_1)$, where $\hat{\pi}_1 = (\pi'_1, 1 -$

- Y_{π_1} .
- (2) Secondly, we generate another trinomial variable V from Trinomial(1 − t_{ij}, π₂). In R package, it should be V < -rmultinom(1 − t_{ij}, 1, π₂), where π₂ = (π'₂, 1 − 1'π₂). Then we let W = U + V.
- (3) Finally, we get y_{ij} = W[1 : 2], which means y_{ij} takes the vector consists of the first two element of W.

Notice that in R package, U, V and W are three dimensional vectors. Therefore, to accommodate the development in this thesis, our two dimensional response y_i takes the first two elements of W, that is $y_i^{(i)}_{(i)} < -W[1 : 2]$ in R package. When a y_i takes $(0, 0)^i$, it means that this subject gives an "manue" answer which can be further trunds at a mining value.

We conduct 500 simulations in each one described in Section 4.3.1. In Cose I, II and III, we apply the silposted GQL estimating equations (4.4.2), and in Cose T i and V_1 we apply the GQL estimating equations (50). For Gase V, we apply the GQL estimating equations (4.12) but use data t, with perfect sensitivity and specificity, that is $v_1 = 1$ and $v_0 = 1$, which is equivalent to the estimating equations (3.22) in Section 3.1.2.

4.4.3.2 Simulation results

The simulation results are presented in Table 4.13. It can be seen from the table that ignoring both microardination and missing values (*Gase I*) leads to nonignorable biases on estimates of model parameters $\theta = (\beta_{ij}, \beta_{ij}, \gamma_{ij})$ and the biases become bigger when there are bighter degree of microardination in the there see settings of FMC matrix from (h, (b) to (c)). For example, in Cure 1, the CQL estimates of c 1 in three settings are, respectively, (33.10, 3.066, 1.276), (0.155, 0.081, 1.679) and (0.452, 0.408, 0.019), These emission are are and more hand from three three volues ((-1, -1, -1)) as the degree of minimum constraints) and the constraints of the constraint probabilities of CQL confidence intervents becomes been and lower. When we only ignore minimum fidence in the constraints of the constraints of the constraint of the degree of minimum constraints. The constraints in the constraint of the better, which can be near by the emigration of much in Gauss T and Const P is the constraints of the constraints of the minimum constraints. The correspondence of the constraints of the constraints of the constraints in the constraints Cane I and P' with them in Case V, or encodered that ignoring indemotivity basis in miniparable biases on estimates models of monotor the constraints.

 in *Case III*. Moreover, it can be seen that, in *Case II* and *III*, estimatos in setting (b) tends to be the work and estimatos in setting (c) eshibit the best performance as far as the CPv's and ESE's are concerned. Therefore, it can be concluded that ignoring missing values results in hose of efficiency And deleting all observations of these subjects with incomplete data conserverer ions.

To exercit hans of the estimator ensured by michaelfordination and box of the densy due to mining shows, we use the exercised OGL estimuting equations (143) to axiante the model parameters by taking mixing values and michaelforiation into consideration. The number are repetided Case 1 ¹⁵ m Table 1.3.1. Is this case, the results when that the exercised OGL equipandian protocol encoding parameter mixing and exponential σ^2 = (10.2018, 13.014 eV eV (1934), 0.881, 0.883, 0.9863) in setting (1) with most measurement excues, and δ^2 = (1.001, 0.884, 0.813, 0.9863) in setting (2) with most measurement excues, and δ^2 = (1.001, 0.814, 0.813, 0.8633) of 0.813, 0.813

Based on the above discussion and the comparisons for ensample, Cus F and FJ and Cus V, Cus H and H wich Cus V. Cus F and F with Cus H and H, we can conclude that ignoring micelauditation back to non-ignorable biases on submation of model parameters, whereas anglecting insign values multiple in highly himses and insignation of parameters and indigital himses on estimated statasets and ensampted in a conclusion of parameters and indigital himses on estimated statasets and ensampted in a conclusion of parameters and indigital himses are continued in stataset for parameters and large transmission of the context of the indiget with investigation transmission of parameters and the context of the indiget with investigation transmission of the context of the indiget with investigation of the indiget of the indicet of the indiget of the indiget of the indicet of t

| | | (a) | | | (b) | | | (c) | | |
|------|----------|-----------|-----------|-------|-----------|-----------|-------|-----------|-----------|-------|
| Case | Quantity | β_1 | β_2 | Ŷ | β_1 | β_2 | 2 | β_1 | β_2 | 7 |
| I | SM | 0.810 | 0.805 | 1.274 | 0.554 | 0.491 | 1.079 | 0.452 | 0.438 | 0.821 |
| | SSE | 0.117 | 0.273 | 0.312 | 0.102 | 0.259 | 0.314 | 0.097 | 0.217 | 0.256 |
| | ESE | 0.117 | 0.271 | 0.305 | 0.108 | 0.249 | 0.303 | 0.007 | 0.217 | 0.273 |
| | CPr | 0.618 | 0.888 | 0.864 | 0.016 | 0.468 | 0.684 | 0.000 | 0.244 | 0.302 |
| п | SM | 0.956 | 0.990 | 1.425 | 0.965 | 0.983 | 1.578 | 1.032 | 1.026 | 1.528 |
| | SSE | 0.144 | 0.329 | 0.351 | 0.193 | 0.470 | 0.482 | 0.218 | 0.467 | 0.420 |
| | ESE | 0.141 | 0.320 | 0.344 | 0.201 | 0.450 | 0.460 | 0.213 | 0.459 | 0.442 |
| | CPr | 0.941 | 0.940 | 0.935 | 0.962 | 0.942 | 0.930 | 0.944 | 0.948 | 0.961 |
| ш | SM | 0.974 | 1.017 | 1.401 | 1.035 | 1.114 | 1.403 | 1.032 | 1.046 | 1.526 |
| | SSE | 0.199 | 0.414 | 0.450 | 0.321 | 0.680 | 0.798 | 0.219 | 0.505 | 0.493 |
| | ESE | 0.143 | 0.322 | 0.338 | 0.213 | 0.463 | 0.452 | 0.208 | 0.456 | 0.448 |
| | CPr | 0.852 | 0.884 | 0.832 | 0.856 | 0.882 | 0.828 | 0.955 | 0.936 | 0.928 |
| IV | SM | 0.860 | 0.809 | 1.381 | 0.581 | 0.488 | 1.147 | 0.456 | 0.437 | 0.833 |
| | SSE | 0.115 | 0.275 | 0.311 | 0.102 | 0.258 | 0.312 | 0.098 | 0.218 | 0.256 |
| | ESE | 0.116 | 0.271 | 0.306 | 0.107 | 0.250 | 0.304 | 0.097 | 0.217 | 0.273 |
| | CPr | 0.746 | 0.888 | 0.922 | 0.030 | 0.468 | 0.756 | 0.000 | 0.256 | 0.318 |
| v | 854 | 1.022 | 1.008 | 1.529 | 1.001 | 0.988 | 1.548 | 1.025 | 1.025 | 1.514 |
| | SSE | 0.143 | 0.326 | 0.348 | 0.190 | 0.462 | 0.475 | 0.216 | 0.460 | 0.425 |
| | ESE | 0.142 | 0.325 | 0.341 | 0.199 | 0.450 | 0.457 | 0.205 | 0.451 | 0.435 |
| | CPr | 0.952 | 0.944 | 0.952 | 0.958 | 0.950 | 0.942 | 0.945 | 0.961 | 0.956 |
| VI | SM | 1.019 | 1.011 | 1.524 | 1.001 | 0.991 | 1.503 | 1.015 | 1.011 | 1.509 |
| | SSE | 0.119 | 0.277 | 0.313 | 0.120 | 0.289 | 0.322 | 0.120 | 0.278 | 0.281 |
| | ESE | 0.120 | 0.278 | 0.303 | 0.120 | 0.277 | 0.305 | 0.120 | 0.277 | 0.301 |
| | CPr | 0.948 | 0.953 | 0.954 | 0.956 | 0.949 | 0.946 | 0.948 | 0.956 | 0.952 |

Table 4.13: Simulation results under GQL approach for imperfect data due to missing values and misclassification with the true value: $\theta = (1, 1, 1.5)$.

sification and very few missing values. Our corrected GQL can effectively estimate model parameters and the corresponding standard errors as well as the confidence intervals with a specific nominal level.

Chapter 5

Modeling Mis-measured Longitudinal Count Data

5.1 Overview

In Gauger 2, we developed its models (2.13) and (2.17) to develoe out entrust is suggested data. See the models can be and to characterize surrounded and understanding data's with impedient startility and specificity. Also, they can array and the the prefect startility or a specificity or both of them. Act toddy, is sum in parpendinois based longitudinal starks, the follow-up observations of court response are observational starting of the index of the starting based on the models of the prefection of the starting of the starting based on the starting of the starting based on the star

5.2 Miscounted Binomial Count Data with Dynamic Population

5.2.1 Models

The model dealing with miscounted binomial data is given by

$$Y = \pi^{+} * T + (1 - \pi^{-}) * (N - T),$$
 (5.1)

where N is population airs is as any. T is the true round of pattern infected by an epidemic discuss in this area, Y is the properties of count of discuss cases from a registration system, and s^* and s^* are, respectively, the sensitivity and the specificity of the registration system. As maximum before, as a long-trained study builty for sensity struct, the analysis pathetics in a new parently changes over time due to birth, doub, immigration and emigration. Therefore, it is meanable to assume that the population size N is random, and it is following a long-trainal process. In this decision, we same out the topopulation in time a specified system. Total decision we can be the the expedication in the specific system. Total

Let n_{ij} be the observation of the population size N_{ij} of the *i*th district in the *j*th year. The true response T_{ij} describing the count of disease cases (e.g. infection by asthma) in this area can be assumed to follow the binomial model which is given by

$$T_{ij}|_{N_0=n_{ij}} \sim b(n_{ij}, p_{ij}),$$
 (5.2)

where

$$p_{ij} = \frac{exp(x'_{ij}\beta)}{1 + exp(x'_{ij}\beta)}$$
(5.3)

is the dissent rate in the risk area during the jtdy year. The population size N_0 is from a signature model and lise expectation and variance are, respectively, δ_{ij} and δ_{jr}^2 . The minimum properties the solution respectively, δ_{ij} and δ_{irr}^2 the minimum properties of the solution rate of the series of the observed frame transmission of the relationship between the true response T_{ij} and the observed respectively rate of solution of promises.

$$Y_{ij} = \pi^+ * T_{ij} + (1 - \pi^-) * (N_{ij} - T_{ij}),$$
 (5.4)

In the binomial model (5.2-5.3), it is reasonable to assume that, for $1 \le j \ne u \le J$, and $i = |..., I, T_{ij}|(N_{ij}), n_{ij} = \frac{d}{T_{ij}}|N_{ij}$, and $T_{iij}(N_{ij}), N_{iij} = \frac{d}{T_{ij}}|N_{ij}$. The latter assumption also means that, given N_{ij} and N_{in} . T_{ij} will be independent of T_{ii} . In fact, V_{ij} and Y_{ij} are also independent or den other conditional on N_{ij} and N_{in} .

As discussed in Section 2.3, given the population size $N_{ij} = n_{ij}$, the observed response follows a binomial distribution $Y_{ij} \sim b(n_{ij}, q_{ij})$, where $q_{ij} = 1 - \pi^- + (\pi^- + \pi^+ - 1)\rho_{ij}$ is the reported disease rate of the ith area in the *j*th year based on the
registration system. This means that we can ado the heunobervalub T_{ij} and built the direct relationship between V_{ij} and N_{ij} . In fact, the dynamic patterns of both V_{ij} of T_{ij} are determined by the longitudinal process of N_{ij} . Therefore, cover we have defined a specific model for N_{ij} . It is easy to model V_{ij} and T_{ij} . As a result, it is easy to calculate the moments of V_{ij} . The expectation and variance of V_{ij} , which are disting to the expressions (2.13) and (2.14). (Constet 2, are given by for

$$E(Y_{ij}|N_{ij}) = N_{ij}q_{ij} \qquad (5.5)$$

$$\mu_{ii} = E(Y_{ii}) = \phi_{ii}q_{ii},$$
 (5.6)

and

$$Var(Y_{ij}) = \phi_{ij}q_{ij}(1 - q_{ij}) + q_{ij}^2 Var(N_{ij}).$$
 (5.7)

The expectation of pairwise product of Y_{ii} and Y_{in} is formulated by

$$E(Y_{ij}Y_{in}) = E[E(Y_{ij}Y_{in}|T_{in}, T_{ij}, N_{in}, N_{ij})]$$

 $= E[E(Y_{ij}Y_{in}|N_{ij}, N_{in}]$
 $= E[g_{ij}g_{in}N_{ij}N_{in}]$
 $= q_{ij}q_{in}E(N_{ij}N_{in}).$

Hence, the covariance between Y_{ii} and Y_{in} is

$$Cor(Y_{ij}, Y_{in}) = E(Y_{ij}Y_{in}) - E(Y_{ij})E(Y_{in})$$

 $= q_{ij}q_{in}[E(N_{ij}N_{in}) - \mu_{ij}\mu_{in}]$
 $= q_{ij}q_{in}Cor(N_{ij}, N_{in}).$ (5.8)

These moments are very useful in the development of the GEE and GQL approaches for the estimation of model parameters. To address the issue of estimation, it is assumed that the population sizes N_{ij} of district *i* during the studying period follows the linear transition model (LT) described in section 3.2.1.2 of Chapter 3. The LT model is given by

$$N_{ij}|N_{i,j-1} = n_{i,j-1} \sim Poisson(\phi_{i,j;j-1} = \xi_{ij} + \gamma n_{i,j-1}),$$
 (5.9)

where

$$\xi_{ij} = exp(z'_{ij}\alpha).$$
 (5.10)

 z_{ij} represents some covariates related to the dynamics of the population size N_{ij} . The baseline observations are n_{i0} 's. From Subsection 3.2.3, it is easy to get the expectation and variance of N_{iii} .

$$\phi_{ij} = E(N_{ij}) = \gamma \phi_{ij-1} + \xi_{ij} = \sum_{\alpha=n+1}^{j} \xi_{ii} \gamma^{j-\alpha} + \gamma^{j-\alpha} \phi_{i\alpha}$$
 (5.11)

with $\phi_{i0} = n_{i0}$, and

$$\zeta_{ij}^2 = Var(N_{ij}) = \phi_{ij} + \gamma^2 \zeta_{ij-1}^2 = \sum_{n=n+1}^{j} \phi_{in} \gamma^{2(j-n)} + \gamma^{2(j-n)} \zeta_{in}^2$$
 (5.12)

for u < j, $\varsigma_{11}^{2} = \phi_{11}$ because $N_{i1} \sim Poisson(\phi_{i1})$. For convenience, we assume that $\varsigma_{20}^{2} = 0$. The covariance between N_{i2} and N_{in} is

$$Cov(N_{ij}, N_{in}) = \gamma^2 \varsigma_{uu}^2$$
, for $u < j$. (5.13)

The correlation coefficient between N_{ii} and N_{in} is given by

$$Corr(N_{in}, N_{ij}) = min\{1, \gamma^{j-n}\frac{\varsigma_{in}}{\varsigma_{ij}}\}$$

5.2.2 Estimation of the model parameters

We apply the GEE and GQL methods to estimate the model parameters. In the GEE approach, we choose the independent correlation structure as the working correlation.

Suppose that the semilitive γ^{a} and specificity γ^{a} are income or their estimators are be obtained from pior experience or validation stuffue. The parameters of interest are or $\langle \sigma_{i}',\sigma_{i}',\gamma',\gamma'$ where a rescence of the effects of the factors x_{ij} which is with the disasse rate p_{ij} in (3.3), α describes the effects of covariates z_{ij} which is related to q_{ij} that D_{ij} fixed (5.3) defined on the population size N_{ij} , and γ is the dynamic dependence parameter in the LT model.

To estimate the model parameters using the GEE approach, we solve the following estimating equations

$$\sum_{i=1}^{l} \frac{\partial \mu'_{i}}{\partial \theta} W_{i}^{-1}(y_{i} - \mu_{i}) = 0, \quad (5.14)$$

where $y_i = (y_{i1}, ..., y_{ik})^*$ is the observation vector of response Y_i . In the equation (5.14), μ_i is the expectation of Y_i . W_i is the "working" covariance matrix and it can be written in the form of W_i = $diag(\phi_{11}^{i_1}, \phi_{12}^{i_1}, ..., \phi_{12}^{i_1}), \sigma_{21}^{i_2} = V(Y_{ij}), \partial \mu_i / \partial \theta$ is the first order derivative of μ_i with respect to θ .

The GEz approach is considered to be an effective procedure to actimate the model parameters in the situation that the true covariance matrix of Y_{ii} which is source that Y_{ii} is which in the situation of the true covariance matrix X_{ii} is some cases, the GEZ approach employing working covariance matrix W_{ii} will lead to how of effective, compared with the GQL approach that explains the true covariance matrix Y_{ii} . The GQL instantiance matrix are size in low

$$\sum_{i=1}^{J} \frac{\partial \mu'_i}{\partial \theta} \Sigma_i^{-1}(y_i - \mu_i) = 0, \quad (5.15)$$

and all the elements of Σ_i can be calculated based on expressions (5.6-5.8) and (5.10-5.13). The expressions (5.6) and (5.11) lead to the following relationship between the expectations of Y_{ij} and Y_{ij-1}

$$\mu_{ij} = \gamma \mu_{ij-1} + q_0 \xi_{ij}. \quad (5.16)$$

Therefor, under the joint model (5.4) and (5.9), the elements of $\partial \mu_i / \partial \theta$ are given by

$$\frac{\partial b_{ij}}{\partial d} = (\pi^- \pi^+ - 1) \frac{\partial b_{ij}}{\partial d} \phi_{ij}$$

$$= (\pi^- \pi^+ - 1) \phi_{ij}(1 - p_i) x_{ij}, \quad (5.17)$$

$$\frac{\partial p_{ij}}{\partial a} = \theta_0 \frac{\partial \phi_{ij}}{\partial a}$$

$$= \theta_0 \left(\frac{\partial \phi_{ij-1}}{\partial a} + \theta_0 \right) x_{ij}$$

$$= \left(\gamma \frac{\partial \phi_{ij-1}}{\partial a} + \theta_0 \phi_{ij} \right) \delta_{ij} \quad (5.18)$$

for j = 1, ..., J, and

$$\frac{\partial q_{ij}}{\partial \gamma} = q_0 \frac{\partial \phi_0}{\partial \gamma}$$

 $= q_0 \left(\gamma \frac{\partial \phi_{ij-1}}{\partial \gamma} + \phi_{ij-1}\right)$
 $= \gamma \frac{\partial \mu_{ij-1}}{\partial \gamma} + \mu_{ij-1}$ (5.19)

for j = 1, ..., J.

Once we have the estimates of ϑ under the GEE and GQL approaches, say $\hat{\theta}_{GEE}$ and $\hat{\theta}_{QQL}$, their corresponding covariance matrix can be consistently estimated by

$$\hat{V}(\hat{\theta}_{GEE}) = \left[\sum_{i=1}^{J} \frac{\partial \mu'_{i}}{\partial \theta} W_{i}^{-1} \frac{\partial \mu_{i}}{\partial \theta}\right]_{\theta = \theta_{GEE}},$$
 (5.20)

and

$$\hat{V}(\hat{\theta}_{GQL}) = \left[\sum_{l=1}^{J} \frac{\partial \mu'_{l}}{\partial \theta} \Sigma_{l}^{-1} \frac{\partial \mu_{l}}{\partial \theta}\right]|_{\theta = \hat{\theta}_{GQL}},$$
 (5.21)

respectively.

One may notice that even we do not know the specific dynamic model of (N_{ij}) , the GEE and GQI approaches are still feasible as long as the first and the second moments of $N_i = (N_{i1}, N_{i1}, ..., N_{in})$ are known. More practical applications of those approaches can be expected due to this fieldility.

5.2.3 Simulation studies

We have two objectives of the simulation studies in this subsection. The first care is to examine the attermations of the naive estimates of model parameters due to ignoring the measurement errors. The second one is to check the performance of the corrected GEE and GQL approaches in correcting the atternation by taking account errors into consideration when using the measurement error model (5.4).

The pursuances of interest include |1 in the bisonial model (5.2-3), and (c/z), in the II model (3.0). The summitty and the specificity (r^+, r^-) in the own are set and (5.4) are assumed to be known in minimization. The scontrate well be briefed at show in the simulation design in Stechnis 2.3.3.1. From the host of N, T, and Vis the barmpiel disfunction the dark gameration reproduced gravity fields the N of N and Npursuances will show be animated by solving the GEE and GQL estimation gravitons (3.4) and (3.3), respectively. Fassing the GEE and GQL estimates are summarized in Disb 6.4 and Table 2.9 cm b00 summittee reso.

5.2.3.1 Covariate design

We consider I = 100 independent regions each with J = 4 repeated observations for the count responses N_s T_s and Y. As far as the time dependent covariates are concerned, we consider the following design. The covariates x_{ij} involved in the true disease rate model (5.3) are given by

$$x_{\text{off}1} = 1$$
 for $j = 1, 2, 3, 4$ and $i = 1, 2, \dots, 100$.

$$x_{ij(2)} = sin(\frac{\pi}{4}j)$$
, for $j = 1, 2, 3, 4$ and $i = 1, 2, ..., 100$,

which implies that

$$p_{ij} = \frac{exp(\beta_1 + \beta_2 sin(\frac{\pi}{2})))}{1 + exp(\beta_1 + \beta_2 sin(\frac{\pi}{2})))}.$$
 (5.22)

The covariates z_{ij} related to the population size in the LT model (5.9) are given by

$$n_{0(1)} \sim \log(P(10 + j)), j = 1, 2, 3, 4, i = 1, 2, ..., 10;$$

 $n_{i(1)} \sim \log(P(20 + j), j = 1, 2, 3, 4, i = 51, 52, ..., 100,$
 $n_{i(1)} = (1, 1, 0, 1), i = 1, 2, ..., 10i;$
 $n_{i(1)} = (1, 0, 1, 1), i = 51, 52, ..., 100,$
 $n_{0(1)} \sim N((j - 1)/4, 0, 5^2), j = 1, 2, 3, 4, i = 1, 2, ..., 100$

5.2.3.2 Data generation

We choose two different acting for the backine observations of the population size n_0 in the LT model (5.9), that is, $\phi_0 = 200$ and 10. The first setting produces in the propulation ine n_0 , whereas the latter yields small values of n_1 . By assigning the true values of parameters as $\beta = (-250, 0.09)$, $\alpha = (1.00, -1.00, 1.00)$, $\gamma = 0.85$, and $(v^+, \tau^-) = (0.75, 0.90)$, the dual or n_0 , t_{10} , and y_0 can be generated from the procedure described back.

- The baseline observations of population sizes are generated from the Poisson model, say N_B ≈ P(φ₀), i = 1, 2, . . . , 100.
- 2. n_{ij} are generated from the linear transition model (5.9): $N_{ij}|N_{i,j-1} = n_{i,j-1} \sim P(\gamma n_{i,j-1} + \xi_{ij})$, where $\xi_{ij} = exp(z'_{ij}\alpha)$.
- t_{ii} is sampled from the binomial model b(n_{ii}, p_{ii}), where p_{ii} is given by (5.22).
- µ_{ij} is generated from the measurement error model (5.4). We first sample U_{ij} from b(t_{ij}, π⁺) and V_{ij} from b(n_{ij} − t_{ij}, 1 − π⁻), then we calculate µ_{ij} by adding U_{ij} and V_{ij} up, that is µ_{ij} = U_{ij} + V_{ij}.

5.2.3.3 Simulation results

In this subsection, we comider two settings of baseline observations of the pepulation size n_{q_0} . Simulation results under the first setting with $n_{q_1} \stackrel{H}{\rightarrow} Poiscon(\phi_q = 200)$ are given in Table 5.1, and the results under the second setting with $n_{q_1} \stackrel{H}{\rightarrow} Poiscon(\phi_q = 10)$ are growned in Table 5.2.

From the two tables, it is one lowers that ignoring sensurement retrees in the control table label to algorization bases of the entropy of parameter p. Inclusions in the model (3.2.3.1) which thready defines the halternst response 7. The based submission absorpently result is good performance of courage probabilities (GYe's) of CTs with conducionse best 90%. For example, in Table 3.1, the simulation toms (MM) of the naive GEE estimate of β_1 in 2.322 which is for a same from in time wake $-2\beta_{-2}$ and the CYP-course is encouring performance with the saminal local 90%. Similar phenomena also happens in Table 3.2 with small peptitution time, i.e. δ_{+} = 0.3. Therefore, we construct the correspondence or empotent with the saminal local 90%. significant attenuation on the estimates of effects of covariates in the binomial model (5.2-5.3) of the true count response T_{ij} . As a result, mideading evaluation of the true disease rate p_{ij} is obtained.

It is any provide the balance of CEI and CQL studies of the parameters on set which are from the Taule CQL studies (10) should be the propulsion sizes $N_0 h_{\rm e}$ are very dens to their true values. For instance, in Table 5.1, the native CEE instance of a shoggare = (1,0017)-1.0008,0.00097, and the maines CEE similar of a 0.0484. They are equide also to their true walues as = (1, -1, 1) and $\gamma = 0.85$. The maines CQL method has similar models. The their the presentance of the main summars of a in the similar models. The their the presentation the the based models of attaching and pipets n_{ijk} were conduct 500 simulations by studies of a small $n_{ijk} = 1.0$ of which the results are reported to Table 5.2. In the simulations of a and γ_{ijk} and the results are reported in Table 5.2. In the simulations of a mall γ_{ijk} multicular are provided in Table 5.2. In the simulations of a mall γ_{ijk} multicular are provided as well as instances or consolute that large indexidiation in the structure of the energoment data g_{ijk} in the true result of (1)(1). This is because that while well the energoment data g_{ijk} in the true result are and in antimation, the true and gauge q_{ijk} is the simulations g_{ijk} in the true result h_{ijk} are used in an attaching. In the all number of people q_{ijk} is the true result data f_{ijk} are used in a structure, the gauge q_{ijk} data q_{ijk} is the simulation h_{ijk} the simulation h_{ijk} the simulation h_{ijk} the simulation h_{ijk} is the simulation h_{ijk} the sim

population size (n_{ij})

=true count of disease cases (t_{ij}) + true number of healthy $pcople(n_{ij} - t_{ij})$

=reported count of disease cases (t_{ij}) + reported number of healthy people $(n_{ij} - y_{ij})$.

As far as the corrected GEE and GQL approaches are concerned, Table 5.1 and Table 5.2 show that they can effectively estimate all the model parameters. The attenuation on the naive estimates of β in the function p_{ij} can be well adjusted under the corrected DEE and CQL approaches. As an example, in Table 3.1, Rogger = (-2.4000, 4.800) and $R_{\rm Sigma} = (-2.4000, 4.800)$ and $R_{\rm Sigma} = (-2.4000, 4.800)$ and $R_{\rm Sigma} = (-2.400, 4.800)$ and

As auxiliard in Section 3.5.2, the GEE approach hormoring the "working" labpendence exceedation structure tends to yield how efficient antimator compared with the GGE approach comparing the true correlation structures. Many measurements are thin conclusion for error from data. Most of our simulation means in Table 1.2 should be structured and simulation structures Table 2.2 should be structured as a simpler tilts exceeding a structure of $G_{\rm C}$ and G_{\rm

As for an the efficiencies of the corrected antiantics are concreted, the corrected EU ergestrade shows how the soft efficiency. The non-base much limit must cause of Table 51 and Table 52, the CPVs under the corrected GGL approach, are during in: Table 52, the CPVs and x the corrected GGZ approach, bremaphi, in Table 52, the CPVs and x the corrected GGZ approach, breted by are (1564, 1668) which the corrected GGZ approach, break approach, and they are (1564, 1668) which the corrected GGZ approach is not been which, income cause, the same arransism under the GGZ, approach also trade to have highly efficiency than the naive GEE estimates, especially when magnetizes and x are corrected.

| Table 5.1: Simulation results of GEE and | GQL approaches ba | and on the count error |
|---|-----------------------------|-----------------------------------|
| model (6.4), the binomial model (6.2-6.3 |) and the LT model | (6.9) with $(\beta_1, \beta_2) =$ |
| $(-2.50, 0.50), \alpha = (1.00, -1.00, 1.00), \gamma =$ | 0.85 and $(\pi^+, \pi^-) =$ | (0.75, 0.90) under the |
| setting $\phi_0 = 200$. | | |

| | | GEE | | GQL | | | |
|----------------|---------|---------|-----------|---------|---------|-----------|--|
| Quantity | Ideal | Naive | Corrected | Ideal | Naive | Corrected | |
| $SM(\beta_1)$ | -2.4905 | -1.7521 | -2.4937 | -2.4995 | -1.7522 | -2.4940 | |
| SSE | 0.0674 | 0.0544 | 0.1275 | 0.0672 | 0.0542 | 0.1272 | |
| ESE | 0.0688 | 0.0588 | 0.1354 | 0.0670 | 0.0562 | 0.1311 | |
| CPr | 0.960 | 0.000 | 0.956 | 0.952 | 0.000 | 0.948 | |
| $SM(\beta_2)$ | 0.5022 | 0.2219 | 0.4933 | 0.5022 | 0.2219 | 0.4933 | |
| SSE | 0.1241 | 0.1070 | 0.2240 | 0.1227 | 0.1055 | 0.2213 | |
| ESE | 0.1347 | 0.1188 | 0.2473 | 0.1285 | 0.1102 | 0.2287 | |
| CPr | 0.960 | 0.344 | 0.970 | 0.960 | 0.278 | 0.995 | |
| $SM(\alpha_1)$ | 1.0012 | 1.0017 | 1.0018 | 1.0011 | 1.0017 | 1.0017 | |
| SSE | 0.0403 | 0.0339 | 0.0339 | 0.0392 | 0.0326 | 0.0326 | |
| ESE | 0.0394 | 0.0325 | 0.0325 | 0.0408 | 0.0338 | 0.0338 | |
| CPr | 0.936 | 0.930 | 0.930 | 0.950 | 0.952 | 0.950 | |
| $SM(\alpha_2)$ | -1.0084 | -1.0076 | -1.0073 | -1.0085 | -1.0081 | -1.0076 | |
| SSE | 0.1508 | 0.1243 | 0.1245 | 0.1474 | 0.1187 | 0.1189 | |
| ESE | 0.1492 | 0.1238 | 0.1239 | 0.1516 | 0.1238 | 0.1257 | |
| CPr | 0.944 | 0.945 | 0.947 | 0.952 | 0.953 | 0.952 | |
| $SM(\alpha_3)$ | 0.9949 | 0.9930 | 0.9932 | 0.9951 | 0.9933 | 0.9935 | |
| SSE | 0.1105 | 0.0913 | 0.0913 | 0.1086 | 0.0887 | 0.0888 | |
| ESE | 0.1037 | 0.0862 | 0.0862 | 0.1118 | 0.0948 | 0.0949 | |
| CPr | 0.920 | 0.932 | 0.932 | 0.948 | 0.964 | 0.965 | |
| $SM(\gamma)$ | 0.8487 | 0.8484 | 0.8488 | 0.8488 | 0.8485 | 0.8490 | |
| SSE | 0.0301 | 0.0252 | 0.0251 | 0.0300 | 0.0247 | 0.0245 | |
| ESE | 0.0311 | 0.0260 | 0.0258 | 0.0305 | 0.0254 | 0.0252 | |
| CPr | 0.962 | 0.956 | 0.952 | 0.956 | 0.948 | 0.948 | |

| Table 5.2: Sir | nulation resul | ts of GEE and | GQL appro | aches based o | in the count error |
|-------------------|---------------------|--------------------------|-------------|--------------------------|-------------------------------|
| model (5.4), | the binomial | model (5.2-5.3 |) and the L | T model (5.9 |) with $(\beta_1, \beta_2) =$ |
| (-2.50, 0.50), | $\alpha = (1.00, -$ | $.00, 1.00$, $\gamma =$ | 0.85 and (π | $^{+}, \pi^{-}) = (0.7)$ | 5,0.90) under the |
| setting $d_0 = 1$ | 10. | | | | |

| | GEE | | | GQL | | | |
|----------------|---------|---------|-----------|---------|---------|-----------|--|
| Quantity | Ideal | Naive | Corrected | Ideal | Naive | Corrected | |
| $SM(\beta_i)$ | -2.5048 | -1.7776 | -2.5651 | -2.5042 | -1.7771 | -2.5647 | |
| SSE | 0.1834 | 0.1493 | 0.3721 | 0.1830 | 0.1495 | 0.3722 | |
| ESE | 0.1886 | 0.1601 | 0.3938 | 0.1878 | 0.1556 | 0.3848 | |
| CPr | 0.950 | 0.002 | 0.974 | 0.948 | 0.002 | 0.968 | |
| $SM(\beta_2)$ | 0.4940 | 0.2434 | 0.5393 | 0.4932 | 0.2428 | 0.5386 | |
| SSE | 0.2396 | 0.1954 | 0.4270 | 0.2369 | 0.1948 | 0.4252 | |
| ESE | 0.2483 | 0.2155 | 0.4789 | 0.2427 | 0.2079 | 0.4591 | |
| CPr | 0.970 | 0.790 | 0.974 | 0.964 | 0.790 | 0.956 | |
| $SM(\alpha_1)$ | 1.0026 | 0.9997 | 1.0006 | 1.0027 | 0.9998 | 1.0008 | |
| SSE | 0.0411 | 0.0332 | 0.0329 | 0.0405 | 0.0324 | 0.0321 | |
| ESE | 0.0383 | 0.0312 | 0.0308 | 0.0396 | 0.0326 | 0.0321 | |
| CPr | 0.930 | 0.930 | 0.930 | 0.942 | 0.944 | 0.942 | |
| $SM(\alpha_1)$ | -1.0052 | -0.9942 | -0.9947 | -1.0047 | -0.9958 | -0.9953 | |
| SSE | 0.0922 | 0.0784 | 0.0782 | 0.0014 | 0.0773 | 0.0772 | |
| ESE | 0.0940 | 0.0769 | 0.0767 | 0.0949 | 0.0776 | 0.0775 | |
| CPr | 0.952 | 0.950 | 0.952 | 0.956 | 0.952 | 0.954 | |
| $SM(\alpha_3)$ | 1.0007 | 0.9992 | 0.9993 | 1.0006 | 0.9992 | 0.0992 | |
| SSE | 0.0554 | 0.0490 | 0.0490 | 0.0552 | 0.0487 | 0.0487 | |
| ESE | 0.0507 | 0.0417 | 0.0417 | 0.0533 | 0.0447 | 0.0447 | |
| CPr | 0.924 | 0.906 | 0.906 | 0.938 | 0.924 | 0.924 | |
| $SM(\gamma)$ | 0.8493 | 0.8491 | 0.8498 | 0.8493 | 0.8496 | 0.0850 | |
| SSE | 0.0605 | 0.0512 | 0.0513 | 0.0600 | 0.0507 | 0.0508 | |
| ESE | 0.0604 | 0.0495 | 0.0497 | 0.0599 | 0.0489 | 0.0490 | |
| CPr | 0.954 | 0.940 | 0.938 | 0.954 | 0.944 | 0.942 | |

In conclusion, ignoring measurement errors in court response results in remarkable atternation on the estimates of the effects of covariates in the true disease rate p_{ij} (5.3), which is used to define the model of the true response T_{ij} . However, count errors do not influence the statistical inference on the effects of covariates associated with the population N_{ij} in N_{ij} in the T model (5.9).

Based on the count error model (Ca), the binarial model (C3-33) for two recovers powers T_{cost} of the Lim 2 model (3.2) for production size, q_{cost} encovered GEE and GQL approaches ran consistently originate all model parameters. In addition, the independence GEE appends basis to have of efficiency in estimation due to the observations of the GQL approach lower the truers. The state constances matrix is similable, the GQL appends lower the truer consistence effectives on improve the performance of the bindraws.

5.3 Miscounted Longitudinal Data with Little Information about Population Size

5.3.1 The model

In some situation, it is impossible to have the population size of an area and its first order messent, for example, the population size in the years for from the create population sizes. Suppose that there are two examples in yours 1000 and 3000, such are 1001, 1002, are area to model based on the information of the second years. But the population sizes in 1901, 1903, 1909 may be disficult to model. The information of even any zero in 1901, 1903, 1909 may be disficult to model. not be highly due to which and complexition singuistics, birth and stotds. Although, this draw, it is reasonable to summit that the population size in one of these poses follows a Paisson distribution, however, we do not know anything mos. This means that we do not have any homeloop house the expectation of the distribution, and show how no back about the correction structure thereas population aims in different years. In this situation, model (5.4) is not while for modeling minemated dismost ensuanging the population. And the corrected additive cours even model (2.17) developed in Clouder 2 depends to be an appropriate advectories.

We assume that T_{ij} is the true count of subjects who are infected by a kind of rpidemic disease in the rinh area during the jub year, i = 1, ..., I and j = 1, ..., I. And T_{ij} is the corresponding total of reported disease cases from a surveiliance system. The corrected additive count error model describing the relationship between Y_{ij} and T_{ij} is given by

$$Y_{ii} = \pi^{+} * T_{ii} + e_{iii}$$
 (5.23)

where r^{+} is the simulation of the accordinates optimum. The form $s^{+} s^{+} z_{ij}$ presented to that almost of polarized where a correctly represent by the according corpus regime. The true size of the indexed proposition Z_{ij} is assumed to follow a specificary experiments and it may be associated with some risk factors s_{ij} for example, the environmental experiments. s_{ij} absauch the number of hashing people who are intervertedly specield as dense ences, and it is assumed to be independent of Z_{ij} . However, there is a specific effect $s_{ij} = s^{+} - T_{ij}$. However, grows a specific direct s_{ij} are growed by the size intervertedly and that the growtener expection of $s_{ij} = s^{+} - T_{ij}$. However, growteners are specific direct $s_{ij} = s^{+} - T_{ij}$. However, growteners, $s_{ij} = s^{+} - T_{ij}$. miscounted subjects from the healthy population, for example, the health care level of an area. In some cases, x_{ii} and z_{ij} may share some common covariates.

Similar to the calculations in Section 2.4.2, we can write the expectation of Y_{ij} in the following expression

$$E(Y_{ij}) = \mu_{ij} = \pi^{+}\eta_{ij} + \psi_{ij},$$
 (5.24)

where η_{ij} is the expectation of the true count of disease cases T_{ij} . The variance of Y_{ij} is given by

$$Var(Y_{ij}) = \mu_{ij} + (\pi^+)^2 [Var(T_{ij}) - \eta_{ij}].$$
 (5.25)

The expectation of the pairwise product of Y_{ij} and Y_{is} can be expressed by

$$\begin{split} E(Y_{ij}Y_{is}) &= E[(\pi^{+}*T_{ij}+c_{ij})(\pi^{+}*T_{iu}+c_{iu})] \\ &= (\pi^{+})^{3}E(T_{ij}T_{iu}) + \pi^{+}(\eta_{ij}\psi_{iu}+\psi_{ij}\eta_{iu}) + E(c_{ij}c_{iu}). \end{split}$$

Therefore, the covariance between Y_{ii} and Y_{in} is given by

$$Cov(Y_{ij}, Y_{ik}) = (\pi^+)^2 Cov(T_{ij}, T_{ik}) + Cov(e_{ij}, e_{ik}).$$
 (5.26)

As far as the model of T_{ij} is concerned, we assume that the true count of disease cases T_{ij} follows the LT model described in Section 3.2.2 of Chapter 3, which is given by

$$T_{ij}|_{T_{i,j-1}=b_{i,j-1}} \sim Poisson(\eta_{ij}^{c} = \xi_{ij} + \gamma t_{i,j-1}),$$
 (5.27)

with

$$\xi_{ij} = exp(x'_{ij}\alpha).$$
 (5.28)

The baseline observations are t_{ab} 's. From Section 3.2.2, it is easy to get the expectation and variance of T_{ab} . They are given by

$$\eta_{ij} = E(T_{ij}) = \gamma \eta_{ij-1} + \xi_{ij} = \sum_{u=u+1}^{j} \xi_u \gamma^{i-v} + \gamma^{i-u} \eta_u$$
 (5.29)

with $\eta_0 = t_{i0}$, and

$$\zeta_{ij}^2 = Var(T_{ij}) = \eta_{ij} + \gamma^2 \zeta_{ij-1}^2 = \sum_{v=u+1}^{2} \eta_u \gamma^{2(j-v)} + \gamma^{2(j-u)} \zeta_{iu}^2$$
 (5.30)

for u < j. Let $\zeta_0^2 = 0$, and therefore $\zeta_0^2 = \eta_0$. The covariance between T_{ij} and T_{iw} is

$$Cov(T_{ij}, T_{iu}) = \gamma^2 \varsigma_{iu}^2$$
, for $u < j$, (5.31)

and the correlation coefficient between T_{ii} and T_{in} is given by

$$Corr(T_{ss}, T_{ij}) = min\{1, \gamma^{j-u}\frac{\zeta_{is}}{\zeta_{ij}}\}.$$

5.3.2 Estimation of the model parameters

In this molection, we pay the GEE and GQL methods to continue the model parameters. The interaction guarantees $m = \theta = (\tau_1, \tau_2) = 0$ is the set that the semitricity r^* is hownown its notications can be obtained from pair knowledges or while the interaction of the interaction guarantees are $\theta = (\theta^*, \tau_1, e^*, \pi^*)$ in the same that the simulation r^* is shownown. Jacqueres the fields of the factors x_1 which are anmonitoric within the true scenario of dimension ensures in dimension of the physical system is the dynamic parameters in model (237-252). In observables, the dynamic and x_2 which is shown of the dynamic scenario expected by the serverillance x_2 which is shown ($\theta = 0$ scenarios) expression.

In the GEE approach, we choose the independent correlation structure as the working correlation. The estimating equations are given by

$$\sum_{i=1}^{l} \frac{\partial \mu'_i}{\partial \theta} W_i^{-1}(y_i - \mu_i) = 0, \quad (5.32)$$

where $y_i = (y_{i1}, \dots, y_{ij})^r$ is the observation vector of Y_i , and μ_i is the expectation of Y_i . Under the independence correlation structure, $W_i = \text{diag}(\sigma_{11}^2, \sigma_{21}^2, \dots, \sigma_{2d}^2)$, where $\sigma_{2i}^2 = V \arg(Y_{i1})$. $\partial \mu_i / \partial \theta$ is the first order derivative of μ_i with respect to θ .

To improve the efficiency of the estimation, we use the GQL approach by exploiting the true covariance matrix of Y_i which is denoted by Σ_v . The GQL estimating equations are given by

$$\sum_{i=1}^{l} \frac{\partial \mu'_{i}}{\partial \theta} \Sigma_{i}^{-1}(y_{i} - \mu_{i}) = 0, \quad (5.33)$$

where Σ_i can be calculated based on formulas (5.24-5.26) and (5.29-5.31)).

Under the model assumptions, we can calculate the first order derivative of μ_{ij} with respect to θ . Here, if the sensitivity π^+ is known or can be estimated from other studies, the parameters of interest are $\theta = (\beta', \alpha', \gamma')$. If π^+ is unknown, then can we be interested in estimating $\theta = (\beta', \alpha', \gamma, \pi^+)$. The first order derivatives are

$$\frac{\partial \mu_{ij}}{\partial \beta} = \pi^{+} \frac{\partial \eta_{ij}}{\partial \beta} = \pi^{+} [\gamma \frac{\partial \eta_{ij-1}}{\partial \beta} + \xi_{ij} x_{ij}]$$

 $= \gamma \frac{\partial \mu_{ij-1}}{\partial \beta} + \pi^{+} \xi_{ij} x_{ij},$ (5.34)

$$\frac{\mu_{ij}}{\gamma} = \pi^{+} \frac{\partial \eta_{ij}}{\partial \gamma} = \pi^{+} [\gamma \frac{\partial \eta_{ij-1}}{\partial \gamma} + \eta_{ij-1}]$$

= $\gamma \frac{\partial \mu_{ij-1}}{\partial \pi} + \pi^{+} \eta_{ij-1},$ (5.35)

$$\frac{\partial \mu_{ij}}{\partial \alpha} = \psi_{ij} z_{ij},$$
 (5.36)

$$\partial \mu_{ij} = \eta_{ij},$$
 (5.37)

for j = 1, ..., J and i = 1, ..., I. Notice that the zero baseline observations of T_{ij} , that is $t_{ij} = 0$, lead to $\partial \eta_{3i}/\partial \gamma = 0$, and therefore, $\partial \mu_{ii}/\partial \gamma = 0$.

Once we have the estimates from the GEE and GQL approaches, that is $\hat{\theta}_{GEE}$ and $\hat{\theta}_{OOL}$, the corresponding consistent estimators of their covariance matrices are given

$$\tilde{V}(\tilde{\theta}_{GEE}) = \left[\sum_{i=1}^{l} \frac{\partial \mu'_i}{\partial \theta} W_i^{-1} \frac{\partial \mu_i}{\partial \theta}\right]|_{\theta = \tilde{\theta}_{GEE}},$$
 (5.38)

and

$$\hat{V}(\hat{\theta}_{OQL}) = \left[\sum_{i=1}^{l} \frac{\partial \mu'_i}{\partial \theta} \Sigma_i^{-1} \frac{\partial \mu_i}{\partial \theta}\right]|_{\theta = \hat{\theta}_{OQL}},$$
 (5.39)

respectively.

5.3.3 Numerical study

Similar to the similarity and the Schwerin S.2.2, we conduct a field range minimize the predomans of the new bar of corrected entiration bare of the GEE and GG, expendent. To be specific, we apply the GEE estimates bare of the GEE and GG, expendent. To be specific, we apply the GEE softmatice present (3.23), and the GQL estimating approximation (3.33) is the readom growerised dista to minimise the model parameters R^{-1} . In the invitation, we consider the core of R. The first one is the core N^{+} , we estimate $R = (f, \gamma, of \gamma, whereas is the around$ $core that <math>\pi^{+1}$ is unknown, we estimate $R = (f, \gamma, of \gamma, whereas is the software$ of the the data scenario and scenario transition.

5.3.3.1 Covariate design

We consider I = 60 independent districts each with J = 4 repeated count responses T and Y. As for an the choice of the true values of the parameters are concrued, we consider $\beta = (0.6, -1.0, 1.0)$, $\alpha = (0.30, -0.50)$, $\gamma = 0.8$ and 0.3, and $\pi^{+} = 0.7$ and 0.85. The covariate values for the simulation are given in the following reserves.

The first covariates is assumed to be a variable related to the population size.

Suppose that we randomly divide the 60 districts into 6 groups each consists of 10 districts. Each group is assigned a set of growth rates of population (pgr) which are given by

$$pgr[1,] = (1, 1.01, 1.01^2, 1.01^2),$$

$$pgr[2,] = (1, 0.99, 0.99^2, 0.99^2),$$

$$pgr[3,] = (1, 1.01, 0.90, 0.90^2),$$

$$pgr[4,] = (1, 0.99, 1.01, 1.01^2),$$

$$pgr[5,] = (1, 1.01, 0.99, 1.01),$$

$$pgr[6,] = (1, 0.99, 1.01, 0.99).$$

The time-varying population sizes are generated following the procedure described below:

 $pop((g-1)10+k, j] \sim Poisson(1000(g+k)pgr[g, j]), g = 1, ..., 6, k = 1, ..., 10, j = 1, 2, 3, 4.$

Then we randomly order the rows of matrix pop by the code

$$population < -pop(sample(60, 60))$$

in statistical package R. Then the first covariate $x_{ii(1)}$ is defined as

$$x_{ij1} \sim log(population[i, j]), \text{ for } j = 1, 2, 3, 4, i = 1, 2, ..., 60.$$

Other covariates of x11 are defined as

$$x_{i(2)} = (1, 1, 0, 0), i = 1, 2, ..., 30;$$

 $x_{i(2)} = (0, 0, 1, 1), i = 31, 32, ..., 60.$
 $x_{i(3)} \sim N((j - 1)/4, 0.5^2), j = 1, 2, 3, 4, i = 1, 2, ...,$

The following covariates z_{ij} are defined on e_{ij} . We assume that z_{ij} shares the first element of x_{ii} . So, the design for z_{ii} is given as

$$z_{0(1)} = x_{01}, j = 1, 2, 3, 4, i = 1, 2, ..., 60,$$

 $z_{0(2)} \sim N(1, 0, 5^2), j = 1, 2, i = 1, 2, ..., 30,$
 $z_{0(2)} \sim N(1, 0, 65^2), j = 3, 4, i = 1, 2, ..., 30;$
 $z_{0(2)} \sim N(1, 0, 55^2), j = 1, 2, i = 31, 32, ..., 60,$
 $z_{000} \sim N(1, 0, 7^2), j = 3, 4, i = 31, 32, ..., 60,$

5.3.3.2 Data generation

We set the baseline observations for the population sizes $t_{40} = 0$ in the LT model (5.27). The data T_{ij} , and Y_{ij} can be generated following the procedure described below:

- The true count t_{ij} are generated based on the linear transition model (5.27) with t_{ib} = 0: T_{ij}|_{T_{ij}=1:nl_{ij=1}} follows Poisson(γt_{ij=1} + exp(x'_{ij}α)).
- 2. y_{ii} is generated from the corrected additive error model (5.23).
 - (1). First, we generate $U_{ij} \sim b(t_{ij}, \pi^+)$.
 - (2). For simplicity, we generated independent additive errors e_{ij} given i, that is, e_{ij} ~ P(exp(zⁱ_cα)).
 - (3). The observed count data y_{ij} = U_{ij} + e_{ij}.

5.3.3.3 Simulation results

In this subsection, we consider two sets of values of narameters $\theta = (\beta', \gamma, \alpha', \pi^+)'$ where $\beta = (\beta, \beta, \beta, \gamma)$ are effects of covariates π_{-} associated with the true count of disease cases in the LT model (5.27), γ is the dynamic dependence parameter in the LT model, $\alpha = (\alpha_1, \alpha_2)'$ represent the effects of explanatory variable z_{ii} in the count error model (5.23), and π^+ is the sensitivity in model (5.23). One set of values of θ is $\beta = (0.6, -1.0, 1.0)$, $\gamma = 0.8$, $\alpha = (0.3, -0.5)$ and $\pi^+ = 0.7$, and the other set is $\beta = (0.6, -1.0, 1.0)$, $\gamma = 0.3$, $\alpha = (0.3, -0.5)$ and $\pi^+ = 0.85$. For each set of values of parameters, we consider four types of estimates: ideal, naive, corrected1 and corrected2 estimates under both GEE and GOL approaches. Among these four types of estimates, the ideal estimates are obtained by using the data of true response T., the naive estimates are based on the error-contaminated data n., ignoring the measurement errors: the corrected1 estimation implies that all parameters including the sensitivity π^+ are estimated by employing the observed data y_{ij} and taking errors into consideration under the assumption that π^+ is unknown, and the last one, i.e. the corrected2 estimates, means that all parameters except π^+ are estimated based on m_i taking count errors into account for known π^+ . Notice that, in the ideal and naive frameworks, the parameters need to estimate only consist of β and γ . The simulation results are repeated in Table 5.3 and Table 5.4.

From Table 5.3 and Table 5.4, it is easy to see that both naive GEE and OQL estimates of model parameters have significant biases due to ignoring count errors in the observed data. For example, for the true values of $\beta_2 = -1.0$ in Table 5.3, we estimate it as $\beta_{\rm perse} = -0.393$ is and $\beta_{\rm popul} = -9.9424$ with bias more than 5%. Especially, the coverage probabilities of 90% CTs for the naive estimates of β and γ are considerably biased from the nominal level 0.95. For instance, the CPs's for $\langle \beta_{SGEC}, \gamma_{SGER} \rangle$ are (0.000, 0.003, 0.012, 0.014), and the CPs's for $\langle \beta_{SGE}, \gamma_{SO2} \rangle$, are (0.000, 0.035, 0.003, 0.022), while the specified nominal level is 0.95.

To improve the unsatisfactory naive estimates, we apply the corrected GEE and GOL approaches to estimate the model parameters and construct confidence intervals. Two types of estimates, the corrected1 estimates for unknown sensitivity π^+ and the corrected2 estimates for known π^+ are computed, and the simulation results are given in Table 5.3 and Table 5.4. It can be seen that both corrected1 and corrected2 estimates under the GEE and GQL aneroaches produce tiny biases which can be neglected. For example, for the parameters $(\beta', \gamma)'$ with true values (0.6 -1.0.1.0.0.8) in Table 5.3, the first corrected GEE estimates are (Årouge, Scours) = (0.5994, -0.9995, 0.9997, 0.8001), and the second corrected GEE estimates are (Aurean, Scotter) = (0.6000-0.9997, 0.9998, 0.8002). Similarly, the corrected1 GQL estimates are (\$\delta_{CODE1}, \delta_{CODE1}) = (0.5994, -0.9993, 0.9998, 0.7999), and the corrected2 GQL estimates are (\$700011, \$200011) = (0.6000, -0.9994, 0.9999, 0.8000) Under the corrected approaches, we also obtain excellent estimates of parameters $\alpha = (\alpha_1, \alpha_2)^{\prime}$ which are defined on e_{α} , the number of false disease cases miscounted from the healthy nonnlation. For instance, in Table 5.3, for the true $\alpha = (0.3, -0.5)$, we have decome = (0.2980, -0.5517)', decome = (0.2982, -0.5685)' under the GEE approach, and decours = (0.2981, -0.55577', decours = (0.2983, -0.5714)' under the GOL approach. Beside the parameters mentioned above, we also obtained approximately unbiased estimates of the sensitivity π^+ . For example, with the true value $\pi^{+} = 0.7$ in Table 5.3, we get $\dot{\pi}_{maxes} = 0.7113$ and $\dot{\pi}_{maxes} = 0.7115$, which are very

186

does to fee true value 12. From these examples, we can also see that the second corsol attained assuming a lowers τ^{-1} disc process better than the fact corrected estimates in the case of unlarges τ^{-1} . It should be pointed on that the estimates about the error ended parameters α and τ^{-1} of gravity scientific latence. They are been of the scalar the energy of ansammeter mere in the calculat and factors to space the schuldry of the data subscript procedures such as surveillance parameters and science science.

Similar to the discussion in the minimum study is Section 3.23.1, the bidd and the two types of encoded animation using the GOL approach tend to have higher efficiency than these estimators used to GOL approach set in attaining model parameters. This is because the corrected GQL approach are instaining model independence constainer quarkanic [133], while the CDE method new as welving independence constainer quarkanic [133], while the CDE method new as in Table 3.2 and GDE approach are invested GQL approach are due to the summinlosed 1.65 Å, the GCP was due the corrected GEE approach are due to the summinlosed 3.65 Å, the GCP was due the corrected GEE approach are due to the summinbed DES than the GP's under the corrected GEE approach. For example, the GCP's of J are (1056, 6366, 1030) moder the corrected GEE approach, and (1054, 6368, the GP's) of J are (10540, GGS, 6300, 8300) moder the corrected GEE approach, and (1054, 8300, 2550), moder the corrected GEE approach, and (1054, 6360, 1054, 8300, 2550), moder the corrected GEE approach, and (1054, 8300, 2550, 1030) moder the corrected GEE approach, approach, and (1054, 8300, 2550, 1030) moder the corrected GEE approach, and (1054, 8300, 2550, 1030) moder the corrected GEE approach, and (1054, 8300, 2550), moder the corrected GEE approach, approach, approach approach

In summary, ignoring measurement errors in count data leads to non-ignorable biases of the submarks of effects of countries associated with the true response T_{ij} . Our corrected GEE and GQL approaches, based on count error model (5.23) and the LT model (5.27) for true response T_{ij} , can estimate model parameters almost univaled). In addition, the independence GEE approach hash to los of efficiency

| | GEE | | | | GQL | | | |
|----------------|---------|---------|----------|----------|---------|---------|----------|----------|
| Quantity | Ideal | Naive | Correct1 | Correct2 | Ideal | Naive | Correct1 | Correct2 |
| $SM(\beta_1)$ | 0.6000 | 0.5681 | 0.5994 | 0.6000 | 0.6000 | 0.5680 | 0.5994 | 0.6000 |
| SSE | 0.0010 | 0.0010 | 0.0152 | 0.0018 | 0.0010 | 0.0010 | 0.0151 | 0.0017 |
| ESE | 0.0009 | 0.0010 | 0.0128 | 0.0016 | 0.0009 | 0.0011 | 0.0142 | 0.0016 |
| CPr | 0.924 | 0.000 | 0.892 | 0.918 | 0.943 | 0.000 | 0.939 | 0.941 |
| SM (β_2) | -0.9995 | -0.9378 | -0.9995 | -0.9997 | -0.9994 | -0.9412 | -0.9993 | -0.9994 |
| SSE | 0.0146 | 0.0151 | 0.0233 | 0.0234 | 0.0134 | 0.0143 | 0.0213 | 0.0213 |
| ESE | 0.0132 | 0.0145 | 0.0206 | 0.0213 | 0.0131 | 0.0148 | 0.0202 | 0.0201 |
| CPr | 0.924 | 0.020 | 0.910 | 0.914 | 0.946 | 0.018 | 0.935 | 0.940 |
| $SM(\beta_3)$ | 1.0002 | 0.9624 | 0.9997 | 0.9998 | 1.0003 | 0.9620 | 0.9998 | 0.9999 |
| SSE | 0.0084 | 0.0093 | 0.0127 | 0.0123 | 0.0076 | 0.0090 | 0.0119 | 0.0114 |
| ESE | 0.0075 | 0.0081 | 0.0109 | 0.0106 | 0.0072 | 0.0084 | 0.0110 | 0.0109 |
| CPr | 0.921 | 0.012 | 0.911 | 0.915 | 0.941 | 0.010 | 0.934 | 0.938 |
| $SM(\gamma)$ | 0.8003 | 0.7682 | 0.8001 | 0.8002 | 0.8002 | 0.7721 | 0.7999 | 0.8000 |
| SSE | 0.0070 | 0.0077 | 0.0102 | 0.0099 | 0.0062 | 0.0074 | 0.0092 | 0.0090 |
| ESE | 0.0067 | 0.0076 | 0.0093 | 0.0091 | 0.0061 | 0.0071 | 0.0058 | 0.0088 |
| CPT | 0.940 | 0.014 | 0.924 | 0.934 | 0.948 | 0.024 | 0.940 | 0.946 |
| $SM(\alpha_1)$ | | | 0.2980 | 0.2982 | | | 0.2981 | 0.2983 |
| SSE | | | 0.0206 | 0.0198 | | | 0.0196 | 0.0191 |
| ESE | | | 0.0191 | 0.0187 | | | 0.0180 | 0.0178 |
| CPr | | | 0.944 | 0.946 | | | 0.954 | 0.951 |
| $SM(\alpha_2)$ | | | -0.5517 | -0.5685 | | | -0.5557 | -0.5714 |
| SSE | | | 0.2692 | 0.2999 | | | 0.2422 | 0.2378 |
| ESE | | | 0.2284 | 0.2214 | | | 0.2075 | 0.2064 |
| CPr | | | 0.940 | 0.942 | | | 0.954 | 0.952 |
| $SM(\pi^+)$ | | | 0.7113 | | | | 0.7115 | |
| SSE | | | 0.1013 | | | | 0.1009 | |
| ESE | | | 0.0726 | | | | 0.0936 | |
| CPr | | | 0.842 | | | | 0.936 | |

Table 5.3: Simulation results of GEE and GQL approaches based on the count error model (5.23) and the LT model (5.27) with $\beta = (0.6, -1.0, 1.0), \gamma = 0.8,$ $\alpha = (0.3, -0.5)$ and $\pi^+ = 0.7$.

188

Table 5.4: Simulation results of GEE and GQL approaches based on the count error model (5.23) and the LT model (5.27) with $\beta = (0.6, -1.0, 1.0), \gamma = 0.3,$ $\alpha = (0.3, -0.5)$ and $\pi^+ = 0.85$).

| | GEE | | | | GQL | | | |
|----------------|---------|---------|----------|----------|---------|---------|----------|----------|
| Quantity | Ideal | Naive | Correct1 | Correct2 | Ideal | Naive | Correct1 | Correct2 |
| $SM(\beta_1)$ | 0.6000 | 0.5893 | 0.6006 | 0.6001 | 0.6000 | 0.5891 | 0.6005 | 0.6001 |
| SSE | 0.0009 | 0.0009 | 0.0116 | 0.0017 | 0.0008 | 0.0009 | 0.0115 | 0.0017 |
| ESE | 0.0008 | 0.0009 | 0.0104 | 0.0017 | 0.0009 | 0.0009 | 0.0118 | 0.0017 |
| CPr | 0.930 | 0.000 | 0.926 | 0.940 | 0.952 | 0.000 | 0.964 | 0.948 |
| $SM(\beta_2)$ | -0.9999 | -0.9418 | -0.9991 | -0.9990 | -0.9998 | -0.9440 | -0.9991 | -0.9990 |
| SSE | 0.0112 | 0.0115 | 0.0163 | 0.0160 | 0.0111 | 0.0115 | 0.0160 | 0.0158 |
| ESE | 0.0106 | 0.0110 | 0.0167 | 0.0164 | 0.0109 | 0.0114 | 0.0168 | 0.0166 |
| CPr | 0.945 | 0.002 | 0.956 | 0.954 | 0.948 | 0.002 | 0.954 | 0.952 |
| $SM(\beta_3)$ | 0.9997 | 0.9537 | 0.9994 | 0.9993 | 0.9995 | 0.9533 | 0.9992 | 0.9991 |
| SSE | 0.0073 | 0.00787 | 0.0135 | 0.0124 | 0.0071 | 0.0078 | 0.0134 | 0.0124 |
| ESE | 0.0072 | 0.0076 | 0.0126 | 0.0120 | 0.0072 | 0.0080 | 0.0132 | 0.0125 |
| CPr | 0.946 | 0.000 | 0.930 | 0.940 | 0.952 | 0.000 | 0.942 | 0.953 |
| $SM(\gamma)$ | 0.2998 | 0.2870 | 0.2996 | 0.2995 | 0.2998 | 0.2895 | 0.2996 | 0.2996 |
| SES | 0.0060 | 0.0065 | 0.0069 | 0.0069 | 0.0055 | 0.0062 | 0.0056 | 0.0065 |
| ESE | 0.0058 | 0.0063 | 0.0069 | 0.0069 | 0.0057 | 0.0062 | 0.0057 | 0.0067 |
| CPr | 0.953 | 0.450 | 0.946 | 0.946 | 0.952 | 0.594 | 0.952 | 0.952 |
| $SM(\alpha_1)$ | | | 0.2980 | 0.2979 | | | 0.2980 | 0.2980 |
| SSE | | | 0.0202 | 0.0195 | | | 0.0201 | 0.0192 |
| ESE | | | 0.0194 | 0.0186 | | | 0.0196 | 0.0189 |
| CPr | | | 0.956 | 0.946 | | | 0.952 | 0.949 |
| $SM(\alpha_2)$ | | | -0.5508 | -0.5485 | | | -0.5492 | -0.5469 |
| SSE | | | 0.2325 | 0.2250 | | | 0.2218 | 0.2135 |
| ESE | | | 0.2185 | 0.2119 | | | 0.2142 | 0.2077 |
| CPr | | | 0.959 | 0.950 | | | 0.950 | 0.954 |
| $SM(\pi^+)$ | | | 0.8510 | | | | 0.8516 | |
| SSE | | | 0.0937 | | | | 0.0929 | |
| ESE | | | 0.0838 | | | | 0.0945 | |
| CPr | | | 0.928 | | | | 0.960 | |

in estimation due to application of a false covariance structure. Therefore, if the true covariance matrix is available, we prefer to use the OQL approach in estimating model effects. An maximal provisably, the estimation of the error related effects α and π^{\pm} under the corrected approaches is of significant interest in evaluating the accuracy of the data collected through a series of ascellar procedures.

Chapter 6

Discussion and Future Studies

6.1 Some Remarks

Due to the videly cateling measurement enters in data from probability of targets to for opin iterative to standing the strengtheness of the strengtheness of the information of the strengtheness of the strengtheness of the strength information of the strengtheness of the strengtheness of the strengtheness of commonlate both commonstrated and undersamewheat data. There are for studied ability in strengtheness from strengtheness of the strengtheness that and strengtheness of the strengtheness that strengtheness of the strengtheness that strengtheness of the strengtheness that and strengtheness of the stren which dougle describes the relationship between the ident wateries that in the describes of the second sec

Most of the validing illustrature about momentum energy problems we belowing to the noise narrow discretion for different types of proposes. In recover types, the analysis of targetarbillal proposes relative to momentum energy the noise metriculus and near activations. The theories minds freezes on the analysis of anisometry and the strength of the solution in the process of the analysis of the site of the strength of the strength process of the strength of the strength and the strength of the strength strength strength of the strength of the distribution of the strength strength strength of the strength of the strength strength of the strength strength of the strength of the strength strength strength strength strength strength strength strength decision for dynamic models for highly strength streng (14), the other is the endinory linear transition model (14). There two models on the ord in model forming penderint, matchine emutis, or prevalence courses in produce bandwise tradies. The simulation startine based on the the OQL approach produced bulgly difficult estimates of model parameters which the OQL approach produced bulgly difficult estimates of a model parameters which the OQL approach appears have the the NAM1) model, and the ML approach through the induce parameters which are one parameters which the OQL approach appears in the AD2 approach appears in the appears based are inducedly and the OPM approach course in the induced parameters which are not performed appeared based are inducedly and the OPM appears.

We modeled the misclassified longitudinal categorical responses in Chapter 4. For this nurnose, we combined the longitudinal models in Chapter 3 with the explicit misclassification models in Chapter 2 to clearly describe the relationship between the latent and the observed longitudinal responses. The analysis of misclassified dynamic estimates of parameters of interest, hence result in poor performance of confidence intervals. To obtain more reliable statistical inference, we developed the corrected GQL, OGQL and ML approaches taking into account the measurement errors which are described by EMC model (4.8). All the three approaches produced approximatchy unbiased estimates of model narameters and antisfactory confidence intervals. Especially, the OGQL approach which includes the second order responses into the estimating procedure exhibited almost identical estimates to those from the ML appreach. The EM algorithm was applied in the ML approach due to the unobservable latent response. We reached the same conclusion as that in the case where data are mation matrix which is demonstrated to be consistent by the simulations. In addition, we also considered an interesting situation under which the data involve unbalanced

michanization. This approach can be used to doal with a special type of minicaindication which is MAR caused by "assess" responses. It showed that ignoring the minicy useds note to "muses" responses have to show of ediffering that III produces approximately unbiased estimates of model parameters. On the other hand, ignoring classification enreus definitely symbic in biased estimates and paor confidence intervals of model parameters.

As far as the error-contaminated aggregated data are concerned, we discussed two kinds of mis-measured count responses in Chapter 5. In the first case, the dynamic negulation sizes of an area are assumed to follow the LT model. The error-prone count response is described by the binomial count error model. Simulation study demonstrated that ignoring measurement errors in count responses leads to biased estimates, for both the GEE and GOL approaches, of the effects of covariates associated with true disease rate. However, measurement errors do not affect the estimates of parameters in the model defined for population sizes. This is because the data of population size keep the same no matter we use the true counts or the error-prone counts of disease cases. Our analysis also showed that the corrected GEE and GQL the population size was only assumed to follow a Poisson distribution of which we do not have any further knowledge. The true count response was assumed to follow the LT model, and the observed response was characterized by the corrected additive error model. Analysis showed that ignoring measurement errors leads to biased estimates of all parameters in the model of true response. It is interesting to see that, under the corrected GEE and GQL approaches, besides the unbiased estimates in the

including the parameters in the model of additive errors and the sensitivity, which describe the type I and type II errors of a surveillance program or a registration system. This analysis implies that the effective estimates of these error-related parameters can be used to access the severity of the measurement errors and to evaluate the quality of the data collecter mescders.

6.2 Future studies

Our research on mis-measured longitudinal discrete data presents numerous additional research opportunities.

1. We previously summed that all the minimum probabilities in the BMS models and the multisensitic outer our models are shown with the transmustation of more produces a second second term indication of more produces the second sec

Section 5.2. The second way may involve many extra parameters in estimation, which could lead to loss of efficiency as showed in the simulation study in Section 5.2.6.3.

- 2. Another intermeting problem related to these mitochnologiest probabilities is white they may not be constant over the small subjects. The MC antatic is the EMC models (14), the sensitivity and specificity is the limital const must model (5.4), and the sensitivity in the rometoid addition error model (253) may be associated with some time dependence transition. Or they may vary over different subpositions. In the rometoid addition error model (253) the defense and populations. For example, in the shiftenest atlass attribute the addition error of the defense of the spectra of the sense of the spectra of the spectra of the sense of the sense of the spectra of the sense o
- 3. In the convected additive manuscourt error model, the sampring that the loppulation size follows a Follow flow for the state of the state of

where p is the proportion of the people older than 50 among the population in this zero. Under this assumption, the size of the infected group T and the size of the healty group T^* are not any larger independent Poisson variables. Hence assumptions about the corrected additive model should be adjusted. Therefore, this problem deserves further investigations.

- 4. Brocht im in Chapter 4, we diesend the michaelfed angewich date with a specific type of mining information due to imand² response widek on the arcumediatel by the unblasted misinformation. The monitor has the H is world be an interesting topic in sploinningic studies to analyze data antifering from both measurement erress and alming value. In this correct, the methos mining at rankem terress and mining value. In this correct, the methos mining at rankem (AGA) modelinis. The endition of the sployment [Yi (2006); Wang et al. (2006); Lin (2006); Non-brit, Preacht and Folken (2006); Elsevere, all of them discussion are function; at mediate modeling at the manument. Trades in the simulation, Stability the intereduction, the just interesting of anomalysis. This is the interesting at modeling of an ansumement errors and non-ignorable mining uses on the responses is of great titlenest.
- 5. Furthermore, it can be reasonably assumed that measurement errors in booth and population data are not likely to be spatially independent. Therefore, takening the spatial effects into consideration when modeling measurement errors or missing values is very promising in gaining extra efficiency in the statistical inference.

6. In practice, there are two important sources of measurement errors on count data,

the first type is the miscissification, the other type is the marken error ransed by unknown and superdictable changes. In this thesis, we focused on the measurement errors one could due due to indicatification. At far such random errors are concerned, they tend to yield the same expectations of the observed variable Y and the true variable T. Is in known that the measurement error model for continuous data, table to fine ma a

$$Y = T + e_i$$
 (6.1)

where is the random error which follows a distribution with an usua. For some the field of the sources are in the source of the field of the equation of the source of the source of the source of the source similar frame to (13) m which the random error follows a distribution with are many, majors mode and sequencing possibly maniform for the source space of the source space of the source of the source of the source of the source space of the source of the sou

$$P(x) = ce^{-\theta x^2}$$
, for some $\theta > 0$, (6.2)

on integer support $(-\infty, +\infty)$, where c is such that the total probability mass is one; that is, $c^{-1} = \sum_{n=0}^{\infty} exp(-\theta x^2)$. The other is a modified version of Roy's (2003) discrete normal distribution by latticing a normal distribution, therefore, we call it the "latticed normal distribution". A latticed normal (Lazornal) variate, L_{n}^{*} can be viewed at the discrete consentration of the normal variate X following $N(\mu, \sigma^2)$. The corresponding probability mass function of LX can be written as

$$P(x) = \Phi(\frac{x + 0.5 - \mu}{\sigma}) - \Phi(\frac{x - 0.5 - \mu}{\sigma}),$$
 (6.3)

where $Q_{i,j}$ represents the cumulative distribution function of the standard corand markow available Z. It is easy in see that both Daugupta's (1993) directle standard distribution and one latitude seemed distribution have similar properties as the standard normal distribution such as, symmetry, sero mean and the unique mode at 0 and so on. Studies about states properties and the specific applications of the two models are very resonance.

Besides the problems mentioned above, the more relevant topics are to be investigated.

Bibliography

- Amemiya, T. (1985) Advanced Econometrics. Harvard University Press, Cambridge, Massachusetts.
- [2] Besag, J. (1974) Spatial interaction and the statistical analysis of lattice systems. Journal of the Royal of Statistical Society: Series B, 36, 192-236.
- [5] Biewas, A., Datta, S., Fine, J.P. and Segal, M. R. (2008) Statistical Advances in the Biomodical Sciences: Clinical Trials Epidemiology, Survival Analysis, and Bioinformatics. John Wiley & Sons Inc., Hoboken, New Jersey.
- [4] Blundell, R., Griffith, R., and Windmeijer, F. (2002) Individual effects and dynamics in count data models. *Journal of Econometrics*. 108, 113-31.
- [5] Brandi, A.G., Young, D.M. and Stamey, J.D. (2009) Bayesian inference for comparing two Poisson rates using data subject to underreporting and validation data. *Statistical Methodology*, 7, 98-108.
- [6] Bratcher, T.L. and Stamey, J.D. (2002) Estimation of Poisson rates with misclassified counts. *Biometrical Journal*, 44, 946-56.

- [7] Buzas, J.S. and Tosteson, T.D. and Stefanski, L.A. (2003) Measurement error. Institute of Statistics Minuco Series, paper No. 2544.
- [8] Cameron, A.C. and Trivedi, P.K. (1998) Regression Analysis of Count Data. Cambridge University Press, Cambridge.
- [9] Carroll, R.J., Ruppert, D., Stefanski, L.A., and Crainiceanu, M. (2006) Measurment Error in Non Linear Models: A Modern Perspective. Chapman & Hall, New York.
- [10] Chen, T.T. (1989) A review of methods for misclassified categorical data in epidemiology. Statistics in Medicine, 9, 1095-106; discussion 1107-8.
- [11] Colby, T.V., Tasehar, H.D., Tawis, W.D., Bergstrahl, E.J. and Jett, J.R. (2002) Pathologic review of the Maya Lung Project suscers jourceted]. Is there a case for misdiagnosis or overlingnesis of lung carcinoma in the screened group? *Cancer*, 95, 2081-65.
- [12] Cooper, G.S., Yuan, Z., Stange, K.C., Dennis, L.K., Amini, S.B., and Rimm, A.A. (1999) The sensitivity of Medicare claims data for case ascertainment of six common cancers. *Medical Care*, **37**, 436-44.
- [13] Cui, Y. and Lund, R. (2009) A new look at time series of counts. *Biometrika*, 96, 781-92.
- [14] Dasgupta, R. (1993) Cauchy equation on discrete domain and some characterization theorems. Theory of Probability and Its Applications, 38, 520-24.
- [15] Diggle, P.J., Heagerty P., Liang, K.-Y. and Zeger, S.L. (2002) Analysis of Longitatinal Data. Oxford University Press, Oxford.
- [16] Dundas, I. and Mckenzie, S. (2006) Spirometry in the diagnosis of asthma in children. Current Opinion in Pulmonary Medicine, 12, 28-33.
- [17] Farrell, P.J. and Sutradhar, B.C. (2006) A non-linear conditional probability model for generating correlated binary data. Statistics and probability Letters, 76, 353-61.
- [18] Ferris, B.G., Ware, J.M., Berkey, C.S. Dockery, D.W., Spiro, III A. and Speizer, F.E. (1985) Effects of passive smoking on health of children. *Environmental Health Perspectives*, 62, 289-95.
- [19] Fisher E.S., Whaley F.S., Krushat W.M., Malenka, D.J., Fleming, C., Børon, J.A. and Haia, D.C. (1992) The accuracy of Medicate's hospital chims data: programs has been made, but problems remain. *American Journal of Public Health*, 82, 261-68.
- [20] Fitzmaurice, G.M. and Laird, N.M. (1993) A likelihood-based method for analysing longitudinal binary responses. *Biometrika*, 80, 141-51.
- [21] Friebele, E. (1996) The attack of asthma. Environmental Health Perspectives, 104, 22–25.
- [22] Fuhlbrigge, A.L., Kitch, B.T., Pahliel, A.D., Kuntz, K.M., Neumann, P.J., Dockery, D.W. and Weiss, S.T. (2001) FEV1 is associated with risk of asthma attacks in a pediatric population. *Journal of Allergy Clinical Immunology*, **107**, 61-67.

[23] Fuller, W. (1987) Measurement Error Models. Wiley, New York.

- [24] Furlow, B. (2007) Accuracy of US cancer surveillance under threat. Laword Oncology, 8, 762-63.
- [25] Giercksky, K.E. (1997) Misdiagnosis of cancer due to multiple glove powder granulomas. European Journal of Surgery, Supplement, 579, 11-14.
- [26] Gilliand, F., Li, Y.F., Peters J. (2001) Effect of maternal smoking during pregnancy and environmental tobacco smoke on asthma and wheering in children. *American Journal of Requiratory and Critical Care Medicine*, **163**, 429-36.
- [27] Gustafson, P. (2007) Measurement error modeling with an approximate instrumental variable. Journal of the Royal Statistical Society, Series B, 69, 797-815.
- [28] Gustafson P. (2003) Measurement Error and Misclassification in Statistics and i Immets and Bayesian Adjustments. Chapman & Hall/CRC, Boca Raton.
- [29] Hardin, J. and Hilbe, J. (2003) Generalized Estimating Equations. Chapman and Hall/CRC, London.
- [30] Hossain, S. and Gustafson, P. (2009) Bayesian adjustment for covariate measurement errors: a flexible parametric approach. *Statistics in Medicine*, 28, 1580-600.
- [31] Jenkins, M.A., Cheles, J.R., Carlin, J.B., Robertson, C.F., Hopper, J.L., Dalton, M.F., Holst, D.P., Choi, K. and Glas, G.G. (1996) Validation of quotionnaire and brouchial hyperresponsiveness against respiratory physician assessment in the diagnosis of arthma. Interastional Journal of Spinetioniology 25, 600–16.

- [32] Kanter, M. (1975) Autoregression for discrete processes mod 2. Journal of Applied Probability, 12, 371-75.
- [33] Kim, J.H. (2009) Estimating classification error rate: Repeated cross-validation, repeated hold-out and bootstrap. *Computational Statistics and Data Analysis*, 53(11), 3735-45
- [34] Kipnis, V., Carroll, R.J., Freedman, L.S. and Li, L. (1999) A new dictary measurement error modem and its application to the estimation of relatic risk: Application to four validation studies. *American Journal of Epidemiology*, 150, 612-51.
- [35] Kipnis, V., Midthune, D., Freedman, L.S., Bingham, S., Day, N.E., Riboli, E. and Carroll, R.J. (2003) Bias in dietary-report instruments and its implications for mutritional epidemiology. *Public Health Nutrition*, 5, 915-23.
- [36] Korn, E.L. and Whittemore, A.S. (1979) Methods for analyzing panel studies of acute health effects of air pollution. *Biometrics*, 35, 795-802.
- [37] Küchenhoff, H., Mwalili, S. M. and Lesaffrer, E. (2006) A general method for dealing with misclassification in regression: the misclassification SIMEX. *Biometrics*, 62, 85-96.
- [38] Liang, K.L. and Zeger, S.L. (1986) Longitudinal data analysis using generalized linear models. *Biometrika*, 73, 13-22.
- [39] Liu, W. (2006) The theory and Methods for measurement errors and missing data problems in semiparametric nonlinear mixed-effect models. *Doctorial Thesis*, The University of British Columbia, Vaccover, British Columbia, Canada.

- [40] Louis, T. A. (1982) Finding the Observed Information Matrix when Using the EM Algorithm. *Journal of the Royal Statistical Society: Series B*, 44, 226-33.
- [41] Mallick, B.K., and Gelfand, A.E. (1996) Semiparametric error-in-variables models: A Bayesian approach. Journal of Statistical Planning and Inference, 52, 307-21.
- [42] Mallick T.S. (2009) Conditional weighted generalized wussi-likelihood inferences in incomplete longitudinal models for binary and count data. *Dectorial Thesis*, Memorial University of Newfoundland, St. John's, Newfoundland, Canada.
- [43] Manski, C.F. (1987) Semiparametric analysis of random effects linear models from binary panel data. *Econometrics*, 55, 357-62.
- [44] Marshall, R.J. (1990) Validation study methods of estimating proportions and odds ratios with misclassified data. *Journal of Clinical Epidemiology*, 43, 941–47.
- [45] McGlothlin A., Stamey, J.D. and Seaman, J.W. (2008) Binary regression with misclassified response and covariate subject to measurement error: a Bayesian approach. Biometrical Journal, 50, 123-34.
- [46] McKenzie, E. (1985) Some simple models for discrete variate time series. Water Resources Bulletin, 21, 645-50.
- [47] McKenzie, E. (1986) Antoregressive-moving average processes with negativebinomial and geometric marginal distributions. Advances in Applied Probability 18, 679-705.
- [48] McKenzie, E. (1988) Some ARMA models for dependent sequences of Poisson counts. Advances in Applied Probability, 20, 822-35.

- [49] McKenzie, E. (2003) Discrete variate time series. In Handbook of Statistics, Rao, C.R. and Shanbhag, D., Eds., Elsevier Science, Amsterdam, 573-606.
- [50] Motoo, Y., Watanabe, H. and Sawabu, N. (1996) Sensitivity and specificity of tumor markers in cancer diagnosis. *Nepton Rivelso*, 54, 1587-91.
- [51] Neuhaus, J.M. (1999) Bias and efficiency loss due to misclassified responses in binary regression. *Biometrika*, 86, 843-55.
- [52] Neuhaus, J.M. (2002) Analysis of clustered and longitudinal binary data subject to response misclassification. *Biometrics*, 58, 675-83.
- [53] Newey, W.K. and McFadden, D. (1993) Estimation in large samples. In Handbook of Economics, McFadden, D. and Engler, R. eds., North Holland, Amsterdam.
- [54] Nicoletti, C., Pernechi, F. and Foliano, F. (2009) Estimating income poverty in the presence of missing data and measurement error. *German Socio-Economic Panel Study*, paper No. 252.
- [50] Pattenden, S., Antova, T., Neuberger, M., Nikforov, B., De Saris, M., Grize, L., Beinrich, J., Henhu, F., Jamese, N., Lottmann-Gilsson, H., Privalova, L., Rodmal, P., Splicholovs, A., Ziotkowska, R. and Flecher T. (2006) Parental smoking and children's rospicatory health: independent effects of prenatal and postnatal exposume. *Tokaco Control*, 124(), 294-301.
- [56] Qaqish, B.F. (2003) A family of multi-neiste binary distributions for simulating correlated binary variables with specified marginal means and correlations. *Biometriks*, 90, 455-63.

- [57] Rosychuk, R.J. (1999) Accounting for misclassification in binary longitudinal data. Doctorial Thesis, Waterloo University, Waterloo, Ontario, Canada.
- [58] Rosychuk, R.J., Thompson, M.E. (2001) A semi-Markov model for binary longitudinal responses subject to misclassification. *The Canadian Journal of Statistics*, 29, 305–404.
- [59] Rosychuka, R.J., and Islam, S. (2009) Parameter estimation in a model for misclassified Markov data - a Bayesian approach. Computational Statistics and Data Analysis, 53, 3805-16.
- [60] Roy, D. (2003) The discrete normal distribution. Communications in Statistics: Theory and Methods, 32(10), 1871–83.
- [61] Roy, S., Banerjee, T. and Maiti, T. (2005) Measurement error model for misclassified binary responses. Statistics in Modicine, 24, 269-83.
- [62] Roy, S., Banerjee, T. (2009) Analysis of misclassified correlated binary data using a multivariate probit model when covariates are subject to measurement error. Biometrical Journal, 51, 429-32.
- [63] Rabe-Hesketh, S., Pickles, A. and Skroudal, A. (2003) Correcting for covariate measurement error in logistic regression using nonparametric maximum likelihood estimation. Statistical Modelling, 3, 215-32.
- [64] Schafer, D.W. (1987) Covariate measurement error in generalized linear models. *Biometrika*, 74, 385-91.

- [65] Speizer, F.E. (1990) Asthma and persistent in Harvard Six Cities Study. Chest, 98, 1918-958.
- [66] Spingelman, D., Romer, D.L. and Logan, R. (2000) Estimation and inference for logistic regression with constitut misclassification and measurement error in main study/validation study dosign. *Journal of American Statistical Association*, 95(449), 33–64.
- [67] Stamey, D., Seaman, J.W. and Young, D.M. (2005) Bayesian analysis of complementary Poisson rate parameters with data subject to misclassification. *Journal of Statistical Planning and Inference*, **134**, 36–48.
- [68] Stefanski, L.A. and Cook, J. (1995) Simulation extrapolation: The measurement error iackhnife. Journal of the American Statistics Association, 90, 1247-56.
- [69] Stefanski, L.A. and Carroll, R.J. (1985) Covariate measurement error in logistic regression. Annual of Statististics, 13, 1335-51.
- [70] Stefanski, L.A. (1987) The effect of measurement error in parameter estimation. *Biometrika*, 72, 385-89.
- [71] Steutel, F.W. and Harn K-van (1979) Discrete analogues of self-decomposability and stability. Annual of Probability, 7, 893-99.
- [72] Steutel, F.W., Vervaat, W. and Wolfe, S.J. (1983) Integer valued branching processes with immigration. Advances in Applied Probability, 15, 713-25.
- [73] Satradhar, B.C. (2003) An overview on regression models for discrete longitudinal responses. *Statistical Science*, 18, 377-93.

- [74] Sutradhar, B.C. (2008) Inferences in familial Poisson mixed models for survey data. Systèps, 70, 18-33.
- [75] Sutradhar, B.C. and Farrell, P.J. (2007) On optimal lag 1 dependence estimation for dynamic binary models with application to Asthma data. Sankhya, 69, 448-67.
- [76] Sutradhar, B.C. and Das, K. (1999) On the efficiency of regression estimators in generalized linear models for longitudinal data. *Biometrika*, 86, 459-65.
- [77] Sutradhar, B.C., Jowaheer, V. and Szeddon, G. (2008) On a unified generalized Quasi-likelihood approach for familial-longitudinal non-stationary count data. *Soundimmics Journal of Statistics*, 35, 597–612.
- [78] Tong, H. (1990) Nonlinear Time Series: A Dynamical System Approach. Oxford University Press, Oxford.
- [79] Wang, C.Y. (2008) Nonparametric maximum likelihood estimation for Cox regression with subject-specific Measurement Error. Soundinarian Journal of Statistics, 35, 613–28.
- [80] Wang, C.Y., Huang, Y., Chao, E.C., and Jeffcoat, M.K. (2008) Expected estimating equations for missing data, measurement error, and misclassification, with application to longitudinal nonignorably missing data. *Biometrics*, 64, 85-95.
- [81] Wang, P.S., Walker, A.M., Tsuang M.T., Orav, E.J., Levin, R. and Avorn, J. (2001) Finding incident breast cancer cases through US claims data and a state cancer registry. *Cancer Cases Control*, **12**, 257-65

- [82] Wang, P.S., Walker, A., Tsuang, M., Orav, E.J., Levin, R. and Avorn, J. (2009) Strategies for improving comorbidity measures based on Medicare and Medicaid claims data. *Journal of Clinical Epidemiology*, 53, 571-78.
- [83] Ware, J.H., Dockery, D.W., Spiro, A. III, Speiner, F.E. and Ferris B.G. (1984) Passive smoking, gas cooking and respiratory health of children living in six cities. *American Reviews of Respiratory Disease*, **129**, 306-74.
- [84] Wedderburn, R.W. (1974) Quasi-likelihood functions, generalized linear models, and the Gauss-Newton method. *Biometrika*, 61, 439-47.
- [85] Whittemere, A.S. and Gong, G. (1991) Poisson regression with misclassified counts: application to cervical cancer. *Journal of Royal Statistics Society, Series* C, 40, 81-93.
- [86] Wong, W.H. (1986) Theory of partial likelihood. Annals of Statistics, 14, 88-123.
- [87] Yang, C.-Y., Tien, Y.-C., Hnich, H.-J., Kao, W.-Y. and Lin, M.-C. (1998) Indoor environmental risk factors and child asthma: a case-control study in a subtropical area. *Polistric Pulnonology*, 26, 120-24.
- [88] Yerushahny, J. (1947) Statistical problems in assessing methods of medical diagnosis with special reference to X-ray technique. Public Health Respective, 62, 1432–69.
- [89] Yi, G.Y. (2008) A simulation-based marginal method for longitudinal data with dron-out and mismeasured covariates. *Biostatistics*, 9, 501-12.

[90] Youden, W.J. (1950) Index for rating diagnostic tests. Cancer, 3, 32-35.

- [91] Zeger, S.L., Liang, K.-Y. and Albert, P.S. (1988) Models for longitudinal data: A generalized estimating equations approach. *Biometrics*, 44, 1049-60.
- [92] Zeger SL and Liang K.-Y. (1986) The analysis of discrete and continuous longitudinal data. *Biometrics*, 42, 121-30.
- [93] Zeger, S.L., Liang, K.-Y. and Self, S.G. (1985) The analysis of binary longitudinal data with time-independent covariates. *Biometrika*, 72, 31-8.
- [94] Zeger, S.L. and Qaqish, B. (1988) Markov regression models for time series: a quasi-likelihood approach. *Biometrics*, 44, 1019-31.
- [95] Zhao, L.P. and Prentice, R.L. (1990) Correlated binary regression using a quadratic exponential model. *Biometrika*, 77, 642–68.
- [96] Zucker D.M. and Spiegelman D. (2008) Corrected score estimation in the proportional hazards model with misclassified discrete covariates. *Statistics in Modelicine*, 27(11), 1911-33.





